## FORM 10-Q SECURITIES AND EXCHANGE COMMISSION Washington, D. C. 20549

QUARTERLY REPORT UNDER SECTION 13 OR 15(D) OF THE SECURITIES EXCHANGE ACT OF 1934 For the quarterly period ended June 30, 2005 OR TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(D) OF THE SECURITIES EXCHANGE ACT OF 1934 For the transition period from \_ Commission file number **0-10546** LAWSON PRODUCTS, INC. (Exact name of registrant as specified in its charter) Delaware 36-2229304 (State or other jurisdiction of (I.R.S. Employer Identification No.) incorporation or organization) 1666 East Touhy Avenue, Des Plaines, Illinois 60018 (Address of principal executive offices) (Zip Code) Registrant's telephone no., including area code: (847) 827-9666 Not applicable Former name, former address and former fiscal year, if changed since last report. Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes X No

Indicate by check mark whether the registrant is an accelerated filer (as defined in Rule 12b-2 of the Exchange Act). Yes  $\underline{X}$  No

Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date.

9,024,111 Shares, \$1 par value, as of July 18, 2005.

(Mark one)

# PART I – FINANCIAL INFORMATION

# Item 1. Financial Statements

# LAWSON PRODUCTS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED BALANCE SHEETS

(Amounts in thousands, except share data)	June 30, 2005	December 31, 2004
	(UNAUDITED)	
ASSETS Current Assets:		
Carrent Assets:  Cash and cash equivalents	\$14,662	\$28,872
Accounts receivable, less allowance for doubtful accounts	53,636	52,129
Inventories	71,102	65,687
Miscellaneous receivables and prepaid expenses	10,545	9,560
Deferred income taxes	1,534	1,729
Total Current Assets	151,479	157,977
Property, plant and equipment, less		
allowances for depreciation and amortization	40,465	42,452
Deferred income taxes	16,024	14,779
Goodwill, less accumulated amortization	28,649	28,649
Other assets	17,503	16,693
Total Assets	\$254,120	\$260,550
LIABILITIES AND STOCKHOLDERS FOLLTW		
LIABILITIES AND STOCKHOLDERS' EQUITY Current Liabilities:		
Accounts payable	\$11,310	\$8,746
Accrued expenses and other liabilities	25,934	32,628
Income taxes	1,194	
Current portion of long term debt	801	1,573
Total Current Liabilities	39,239	42,947
A 18195 1 5 1 1	22 525	24 520
Accrued liability under security bonus plans Other	22,727 16,593	21,528 15,743
Ouiei		_
	39,320	37,271
Stockholders' Equity:		
Preferred Stock, \$1 par value:		
Authorized - 500,000 shares		
Issued and outstanding - None		
Common Stock, \$1 par value:		
Authorized - 35,000,000 shares		
Issued and outstanding-(2005-9,043,903 shares; 2004-9,280,935 shares)	9,044	9,281
Capital in excess of par value	3,703	3,467
Retained earnings	163,259	167,187
Accumulated other comprehensive income	(445)	397
Total Stockholders' Equity	175,561	180,332
Total Liabilities and Stockholders' Equity	\$254,120	\$260,550

See notes to condensed consolidated financial statements.

# LAWSON PRODUCTS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF INCOME

# (UNAUDITED)

(Amounts in thousands, except per share data)

	Three Mon	For the Three Months Ended June 30,		ne s Ended 30,
	2005	2004	2005	2004
Net sales Cost of goods sold	\$113,555 43,797	\$104,443 38,796	\$220,637 85,315	\$205,101 74,057
Gross profit	69,758	65,647	135,322	131,044
Selling, general and administrative expenses	60,964	57,455	118,446	112,790
Operating income	8,794	8,192	16,876	18,254
Investment and other income Interest expense	873 (25)	582 (48)	1,403 (51)	1,110 (111)
Income before income taxes	9,642	8,726	18,228	19,253
Provision for income taxes	4,034	3,409	7,664	7,410
Net income	\$5,608	\$5,317	\$10,564	\$11,843
Net income per share of common stock: Basic	\$0.62	\$0.56	\$1.15	\$1.25
Diluted	\$0.61	\$0.56	\$1.15	\$1.25
Cash dividends declared per share of common stock	\$0.20	\$0.18	\$0.40	\$0.36
Weighted average shares outstanding: Basic	9,107	9,446	9,156	9,466
Diluted	9,126	9,475	9,177	9,493

See notes to condensed consolidated financial statements.

# LAWSON PRODUCTS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS

# (UNAUDITED)

For the

(Amounts in thousands)

	Six	Months Ended June 30,
	2005	2004
Operating activities: Net income Adjustments to reconcile net income to net cash provided by operating activities:	\$10,564	\$11,843
Depreciation and amortization Changes in operating assets and liabilities Other	3,374 (13,467) 1,790	3,415 (3,210) 1,164
Net Cash Provided by Operating Activities	2,261	13,212
Investing activities: Additions to property, plant and equipment Other	(1,206)	(2,139) 100
Net Cash Used in Investing Activities	(1,206)	(2,039)
Financing activities: Purchases of treasury stock Payments on long term debt Dividends paid Other	(11,196) (772) (3,640) 343	(2,964) (718) (3,413) 604
Net Cash Used in Financing Activities	(15,265)	(6,491)
Increase (Decrease) in Cash and Cash Equivalents	(14,210)	4,682
Cash and Cash Equivalents at Beginning of Period	28,872	23,555
Cash and Cash Equivalents at End of Period	\$14,662	\$28,237

#### NOTES TO CONDENSED CONSOLIDATED UNAUDITED FINANCIAL STATEMENTS

A) As contemplated by the Securities and Exchange Commission, the accompanying consolidated financial statements and footnotes have been condensed and therefore, do not contain all disclosures required by generally accepted accounting principles. Reference should be made to Lawson Products, Inc.'s (the "Company") Annual Report on Form 10-K for the year ended December 31, 2004. The Condensed Consolidated Balance Sheet as of June 30, 2005, the Condensed Consolidated Statements of Income for the three and six month periods ended June 30, 2005 and 2004 and the Condensed Consolidated Statements of Cash Flows for the six month periods ended June 30, 2005 and 2004 are unaudited. In the opinion of the Company, all adjustments (consisting only of normal recurring accruals) have been made, which are necessary to present fairly the results of operations for the interim periods. Operating results for the three and six month periods ended June 30, 2005 are not necessarily indicative of the results that may be expected for the year ending December 31, 2005.

Certain amounts have been reclassified in the 2004 financial statements to conform with the 2005 presentation.

#### B) Comprehensive Income

Comprehensive income (in thousands) was \$4,877 and \$5,133 for the second quarters of 2005 and 2004, respectively. Comprehensive income includes negative foreign currency translation adjustments of \$731 and \$184 for the three-month periods ended June 30, 2005 and 2004, respectively.

For the six-month periods ended June 30, 2005 and 2004, comprehensive income (in thousands) was \$9,722 and \$11,741, respectively. Comprehensive income includes negative foreign currency translation adjustments of \$842 and \$102 for the six months ended June 30, 2005 and 2004, respectively.

Accumulated comprehensive income consists only of foreign currency translation adjustments, net of related income tax.

#### C) Earnings per Share

The calculation of dilutive weighted average shares outstanding for the three and six months ended June 30, 2005 and 2004 are as follows (in thousands):

Basic weighted average shares outstanding Dilutive impact of options outstanding Dilutive weighted average shares outstanding

Three months ended June 30		
2005	2004	
9,107	9,446	
19	29	
9,126	9,475	
Six mo	onths ended June 30	
2005	2004	

Basic weighted average shares outstanding	9,156	9,466
Dilutive impact of options outstanding	21	27
Dilutive weighted average shares outstanding	9,177	9,493

## D) Revolving Line of Credit

On June 15, 2005, the revolving line of credit was amended to increase the maximum borrowing capacity to \$75 million from \$50 million, extended the maturity date to March 27, 2009 and amended certain financial covenants. The revolving line of credit carries an interest rate of prime minus 150 basis points floating or LIBOR plus 75 basis points, at the Company's option. Interest is payable quarterly on prime rate borrowings and at the earlier of quarterly or maturity with respect to the LIBOR contracts. The line of credit contains certain financial covenants regarding interest coverage, minimum stockholders' equity and working capital, all of which the Company was in compliance with at June 30, 2005. The Company had no borrowings under the line at June 30, 2005 and December 31, 2004.

## E) Other Charges

The table below shows an analysis of the Company's reserves for severance and related payments for the first six months of 2005 and 2004:

In thousands	2005	2004
Balance at beginning of year Cash paid	\$ 1,042 (490)	\$ 2,476 (716)
Balance at June 30	\$ 552	\$ 1,760

## F) Intangible Assets

Intangible assets subject to amortization, included within other assets, were as follows (in thousands):

Trademarks and tradenames
Customer lists

	June 30, 2005	
Gross Balance	Accumulated Amortization	Net Carrying Amount
\$ 1,747 <u>953</u> <u>\$ 2,700</u>	\$ 960 416 \$ 1,376	\$ 787 537 \$ 1,324
	<b>December 31, 2004</b>	
Gross Balance	Accumulated Amortization	Net Carrying Amount

Trademarks and tradenames	\$ 1,747	\$ 935	\$ 812
Customer lists	<u>953</u>	400	<u>553</u>
	\$ 2.700	\$ 1.335	\$ 1,365

Trademarks and tradenames are being amortized over a remaining life of 15.75 years. Customer lists are being amortized over a remaining life of 16.50 years. Amortization expense for intangible assets is expected to be \$83,000 for 2005 and for each of the next four years.

## G) Accounting for Stock-Based Compensation

The Company adopted FASB Statement No. 148, "Accounting for Stock Based Compensation – Transition and Disclosure." This Statement requires additional disclosure within interim financial statements. The following tables show the effect on net income and earnings per share as required by FASB Statement No. 123, "Accounting for Stock-Based Compensation."

	Three Months Ended June 30	
In thousands	2005	2004
Net income-as reported	\$ 5,608	\$ 5,317
Deduct: Total stock based employee compensation expense determined under fair value method, net of tax	-	(1)
Net income-pro forma	5,608	5,316
Basic earnings per share-as reported	.62	.56
Diluted earnings per share-as reported	.61	.56
Basic earnings per share-pro forma	.62	.56
Diluted earnings per share-pro forma	.61	.56
	_	Ionths Ended
		June 30
In thousands	2005	2004
Net income-as reported	\$10,564	\$11,843
Deduct: Total stock based employee compensation expense determined under fair value method, net of tax	-	(4)
Net income-pro forma	10,564	11,839
Basic and diluted earnings per share -as reported	1.15	1.25
Basic and diluted earnings per share -pro forma	1.15	1.25

## Stock Performance Rights

The Company's incentive stock plan provides for the issuance of Stock Performance Rights (SPRs). These SPRs vest over a three to five year period and entitle the recipient to receive a cash payment equal to the excess of the market value of the Company's common stock over the SPR price when the SPRs are surrendered. The Company records an accrued liability based on the number of outstanding vested SPRs and the market value of the Company's common stock.

The compensation expense accrual decreased \$1,157,000 in the first six months of 2005 compared to a \$443,000 increase in the first six months of 2004. The decrease in the compensation expense accrual for the first six months of 2005 is primarily due to the decline in the Company's stock price and fewer vested SPRs outstanding. The increase for the first six months of 2004 resulted from the significant increase in the Company's stock price. The Company issued 30,000 SPRs in the second quarter of 2005.

On December 16, 2004, the Financial Accounting Standards Board (FASB) issued FASB Statement No. 123(R), "Share-Based Payment," which is a revision of FASB Statement No. 123, "Accounting for Stock-Based Compensation." Statement No. 123(R) requires all share-based payments to employees, including grants of employee stock options to be recognized in the financial statements based on their fair values. On April 14, 2005 the Securities and Exchange Commission announced the adoption of a new rule that amended the effective date to be for fiscal years beginning after June 15, 2005. As a result, the Company expects to apply the modified prospective application beginning January 1, 2006. The Company expects that the impact of complying with Statement No. 123(R) will not be significant.

#### H) Segment Reporting

The Company has four reportable segments: Maintenance, Repair and Replacement distribution in the U.S. (MRO-US), International Maintenance, Repair and Replacement distribution in Canada (MRO-CAN), Original Equipment Manufacturer distribution and manufacturing in the U.S. (OEM-US), International Original Equipment Manufacturer distribution in the United Kingdom and Mexico (OEM-INTL).

The operations of the Company's MRO distribution segments distribute a wide range of MRO parts to repair and maintenance organizations primarily through the Company's force of independent field sales agents, as well as inside sales personnel.

The operations of the Company's OEM segments manufacture and distribute component parts to OEM manufacturers through a network of independent manufacturers representatives as well as internal sales personnel.

The Company's reportable segments are distinguished by the nature of products, types of customers, manner of servicing customers, and geographical location.

The Company evaluates performance and allocates resources to reportable segments primarily based on operating income.

Financial information for the Company's reportable segments consisted of the following:

In thousands	June 30		
		2005	2004
Net sales			
MRO-US	\$ 84	4,340	\$ 77,983
MRO-CAN	6	6,670	5,543
OEM-US	19,	,187	16,326

Three Months Ended

OEM-INTL	3,358	4,591
Consolidated total	\$ 113,555	\$ 104,443
Operating income(loss)		
MRO-US	\$ 7,078	\$ 6,885
MRO-CAN	678	739
OEM-US	1,535	623
OEM-INTL	(497)	(55)
Consolidated total	\$ 8,794	\$ 8,192

The reconciliation of segment profit to consolidated income before income taxes consisted of the following:

# Three Months Ended June 30

		June 30		
In thousands	2005	2004		
Total operating income from reportable segments	\$ 8,794	\$ 8,192		
Investment and other income Interest expense	873 (25)	582 (48)		
Income before income taxes	\$ 9,642	\$ 8,726		
		Six Months Ended June 30		
In thousands	2005	2004		
Net sales				
MRO-US	\$ 164,571	\$ 154,379		
MRO-CAN	12,987	10,595		
OEM-US	36,564	31,066		
OEM-INTL	6,515	9,061		
Consolidated total	\$ 220,637	\$ 205,101		
Operating income(loss)				
MRO-US	\$ 13,814	\$ 16,097		
MRO-CAN	1,418	1,139		
OEM-US	2,638	1,101		
OEM-INTL	(994)	(83)		
Consolidated total	\$ 16,876	\$ 18,254		

The reconciliation of segment profit to consolidated income before income taxes consisted of the following:

# Six Months Ended June 30

In thousands	2005	2004
Total operating income from reportable segments Investment and other income Interest expense	\$ 16,876 1,403 (51)	\$ 18,254 1,110 (111)

Income before income taxes	\$	18,228	\$	19,253
----------------------------	----	--------	----	--------

Asset information related to the Company's reportable segments consisted of the following:

In thousands	June 30, 2005	December 31, 2004
Total assets		
MRO-US	\$ 164,100	\$ 174,777
MRO-CAN	16,572	18,519
OEM-US	45,272	40,275
OEM-INTL	10,618	10,471
Total for reportable segments	236,562	244,042
Corporate	17,558	16,508
Consolidated total	\$ 254,120	\$ 260,550

At June 30, 2005 and December 31, 2004, the carrying value of goodwill within each reportable segment was as follows (in thousands):

MRO-US	\$ 22,104
MRO-CAN	4,294
OEM-US	2,251
OEM-INTL	
Consolidated total	\$ 28,64 <u>9</u>

## I) Short Term Debt

On July 1, 2003 the Company adopted FIN 46 which resulted in the consolidation of the Company's investment in a limited partnership, which owns an office building in Chicago, Illinois. In conjunction with the consolidation of its investment, the Company has recorded the long-term debt of the variable interest entity, which represents a mortgage payable relative to the building, of approximately \$0.8 million at June 30, 2005. The interest rate of the mortgage payable is 7.315%, with a maturity date of December 31, 2005. The building and land have a net carrying value of approximately \$4.1 million, which are included in property, plant and equipment. The remaining assets, none of which are significant, are recorded in other assets.

#### Report of Independent Registered Public Accounting Firm

Board of Directors and Shareholders Lawson Products, Inc.

We have reviewed the condensed consolidated balance sheet of Lawson Products, Inc. and subsidiaries as of June 30, 2005 and the related condensed consolidated statements of income for the three-month and six-month periods ended June 30, 2005 and 2004, and the condensed consolidated statements of cash flows for the six-month periods ended June 30, 2005 and 2004. These financial statements are the responsibility of the Company's management.

We conducted our reviews in accordance with standards of the Public Company Accounting Oversight Board (United States). A review of interim financial information consists principally of applying analytical procedures and making inquiries of persons responsible for financial and accounting matters. It is substantially less in scope than an audit conducted in accordance with auditing standards of the Public Company Accounting Oversight Board, the objective of which is the expression of an opinion regarding the financial statements taken as a whole. Accordingly, we do not express such an opinion.

Based on our reviews, we are not aware of any material modifications that should be made to the condensed consolidated interim financial statements referred to above for them to be in conformity with U.S. generally accepted accounting principles.

We have previously audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States), the consolidated balance sheet of Lawson Products, Inc. and subsidiaries as of December 31, 2004, and the related consolidated statements of income, changes in stockholders' equity and cash flows for the year then ended, not presented herein, and in our report dated March 10, 2005, we expressed an unqualified opinion on those consolidated financial statements. In our opinion, the information set forth in the accompanying condensed consolidated balance sheet as of December 31, 2004, is fairly stated in all material respects, in relation to the consolidated balance sheet from which it has been derived.

ERNST & YOUNG LLP

July 18, 2005

"Safe Harbor" Statement under the Securities Litigation Reform Act of 1995: This Quarterly Report on Form 10-Q contains certain forward-looking statements within the meaning of the Private Securities Litigation Reform Act of 1995 that involve risks and uncertainties. The terms "may," "should," "could," "anticipate," "believe," "continues", "estimate," "expect," "intend," "objective," "plan," "potential," "project" and similar expressions are intended to identify forward-looking statements. These statements are not guarantees of future performance and involve risks, uncertainties and assumptions that are difficult to predict. These statements are based on management's current expectations, intentions or beliefs and are subject to a number of factors, assumptions and uncertainties that could cause actual results to differ materially from those described in the forward-looking statements. Factors that could cause or contribute to such differences include, but are not limited to, the following risk factors:

- Lawson's business strategy requires the Company to carry a significant amount of inventory in order to meet rapid processing of customer orders.
- Disruptions of Lawson's information systems could adversely affect the Company.
- There is an ongoing risk that orders may be cancelled or rescheduled due to fluctuations in Lawson's customers' business needs or purchasing budgets.
- Increases in commodity prices are likely to lead to increased costs of goods sold. We may not be able to pass on these increased costs to our customers.
- A limited number of the Company's stockholders can exert significant influence over the Company.
- The Company operates in highly competitive markets. Some of the Company's competitors have a greater variety of merchandise, financial resources, services, personnel resources and competitive pricing than Lawson.
- Those related to general economic conditions and market conditions in the markets and industries the Company serves.
- The risks of war, terrorism, and similar hostilities may adversely affect operating results.

The Company undertakes no obligation to update any such factor or to publicly announce the results of any revisions to any forward-looking statements contained herein whether as a result of new information, future events or otherwise.

#### Item 2 Management's Discussion and Analysis of Financial Condition and Results of Operations

## Quarter ended June 30, 2005 compared to Quarter ended June 30, 2004

The following table presents a summary of the Company's financial performance for the second quarter of 2005 and 2004:

(D. II	200=	% of	2004	% of
(Dollars in thousands)		Net Sales	2004	Net Sales
Net sales	\$113,555	100.0	\$104,443	100.0
Cost of goods sold	43,797	38.6	38,796	37.1
Gross profit	69,758	61.4	65,647	62.9
Operating expenses	60,964	53.7	57,455	55.0
Operating income	8,794	7.7	8,192	7.8
Other	848	0.7	534	0.5
Income before taxes	9,642	8.5	8,726	8.4
Income tax expense	4,034	3.6	3,409	3.3
Net income	\$5,608	4.9	\$5,317	5.1

## REVENUES AND GROSS PROFIT

Net sales for the three-month period ended June 30, 2005 increased 8.7% to \$113.6 million, from \$104.4 million in the same period of 2004.

The following table presents the Company's net sales results for its MRO and OEM businesses for the second quarter of 2005 and 2004:

(Dollars in millions)	2005	2004
MRO	\$91.0	\$83.5
OEM	22.6	20.9
	\$113.6	\$104.4

Combined Maintenance, Repair and Replacement distribution (MRO-US and MRO-CAN) net sales increased \$7.5 million in the second quarter of 2005, to \$91.0 million from \$83.5 million in the prior year period. Sales increased in both the U.S. and Canada for the quarter, principally attributable to the addition of new customers and improved penetration of existing customers, including the continued growth of the non-stock product category. Net sales increased \$2.4 million, to \$7.2 million from \$4.8 million for the non-stock product category. The favorable impact of foreign exchange fluctuations accounted for approximately one-half of the \$1.1 million net sales gain realized in Canada.

Combined Original Equipment Manufacturer (OEM-US and OEM-INTL) net sales increased \$1.7 million in the second quarter of 2005, to \$22.6 million from \$20.9 million. Sales were higher by \$2.9 million in the U.S., but \$1.2 million lower internationally for the three-month period. The sales gain in the U.S. was primarily attributable to the addition of new customers.

The decline in sales internationally is principally due to lower sales volume from a major customer in Europe.

Gross profit margins for the quarters ended June 30, 2005 and 2004 were 61.4% and 62.9%, respectively. The second quarter of 2005 reflected lower gross profit margins in the MRO segment but a slight improvement for the OEM segment. MRO segment gross profit margins decreased from 72.0% in the second quarter 2004 to 70.0% in the comparable quarter of 2005. The 200 basis point decline is principally attributable to higher product costs, including surcharges from steel vendors. OEM segment gross profit margins improved to 26.7% in the second quarter 2005, 40 basis points higher than the prior period's 26.3%. This increase was principally due to price increases and surcharges implemented in the U.S. in early 2005, partially offset by price competition and higher product costs internationally.

#### OPERATING EXPENSES AND OPERATING INCOME

Operating expenses were \$61.0 million and \$57.5 million for the quarters ended June 30, 2005 and 2004, respectively. \$2.7 million of the \$3.5 million dollar increase in quarterly operating expenses for 2005 over the prior year quarter is primarily due to higher sales agent compensation resulting from the increase in sales and higher compensation due to the addition of employees in our marketing and technology departments. As a result of the sales increases and the Company's ability to leverage its operating cost infrastructure over a larger revenue base, operating expenses as a percent of net sales declined from 55.0% in the 2004 period to 53.7% of net sales in the 2005 period.

Operating income for the three-month period ended June 30, 2005 increased to \$8.8 million, from \$8.2 million in the comparable period of 2004. This \$0.6 million increase in operating income is principally attributable to higher net sales, which more than offset lower gross profit margins and operating expense increases.

#### INVESTMENT AND OTHER INCOME

The following table presents investment and other income for the quarters ended June 30, 2005 and 2004:

(Dollars in millions)	2005	2004
Rental income	\$0.5	\$0.5
Realized foreign exchange gains	0.4	
Interest and other		0.1
	\$0.9	\$0.6

The realized foreign exchange gains in the second quarter of 2005 was due to payments of intercompany payable balances by the Company's Canadian subsidiary.

#### INTEREST EXPENSE

Interest expense is primarily attributable to interest paid relative to a mortgage payable for the building owned by the limited partnership referred to in Note I.

## PROVISION FOR INCOME TAXES

The effective tax rates for the three months ended June 30, 2005 and 2004 were 41.8% and 39.1%, respectively. Approximately 2.0 percentage points of the 2.7 percentage point increase in the effective tax rate over these periods is due to higher operating losses from the foreign subsidiaries in the OEM segment in 2005, for which no tax benefit was provided, and to a lesser extent, lower exempt income in 2005 related to executive life insurance also contributed slightly to the increased effective tax in the second quarter 2005 compared to the prior year period.

#### NET INCOME

Net income for the second quarter of 2005 increased 5.5%, to \$5.6 million (\$.61 per diluted share), compared to \$5.3 million (\$.56 per diluted share) in the similar period of 2004. The \$0.3 million increase in net income is the result of the \$0.6 million increase in operating income and higher investment and other income, partially offset by an increase in the effective tax rate discussed above.

## Six Months ended June 30, 2005 compared to Six Months ended June 30, 2004

The following table presents a summary of the Company's financial performance for the first six months of 2005 and 2004:

(Dollars in thousands)	2005	% of Net Sales	2004	% of Net Sales
(=				
Net sales	\$220,637	100.0	\$205,101	100.0
Cost of goods sold	85,315	38.7	74,057	36.1
Gross profit	135,322	61.3	131,044	63.9
Operating expenses	118,446	53.7	112,790	55.0
Operating income	16,876	7.6	18,254	8.9
Other	1,352	0.6	999	0.5
Income before taxes	18,228	8.3	19,253	9.4
Income tax expense	7,664	3.5	7,410	3.6
Net income	\$10,564	4.8	\$11,843	5.8

#### REVENUES AND GROSS PROFIT

Net sales for the six-month period ended June 30, 2005 increased 7.6% to \$220.6 million, from \$205.1 million in the same period of 2004.

The following table presents the Company's net sales results for its MRO and OEM businesses for the first six months of 2005 and 2004:

(Dollars in millions)	2005	2004
MRO	\$177.5	\$165.0
OEM	43.1	40.1
	\$220.6	\$205.1

Combined Maintenance, Repair and Replacement distribution (MRO-US and MRO-CAN) net sales increased \$12.5 million in the first six months of 2005, to \$177.5 million from \$165.0 million, reflecting sales increases in both the U.S. and Canada. Sales gains were principally attributable to the addition of new customers and improved penetration of existing customers, including the continued growth of the non-stock product category, accounting for \$5.0 million of the sales increase for the period. The favorable impact of foreign exchange fluctuations in Canada accounted for approximately one-third of the \$2.4 million net sales increase.

Combined Original Equipment Manufacturer (OEM-US and OEM-INTL) net sales increased \$3.0 million in the first six months of 2005, to \$43.1 million from \$40.1 million, comprised of a \$5.5 million net sales increase in the U.S. offset by a \$2.5 million sales decrease internationally. The sales gain in the U.S. was primarily attributable to the addition of new customers. The decline in sales internationally is due to lower sales volume from a major customer in Europe.

Gross profit margins for the six-month periods ended June 30, 2005 and 2004 were 61.3% and 63.9%, respectively. The year-to-date June 2005 period reflected lower gross profit margins in both the MRO and OEM segments. MRO segment gross profit margins declined from 72.8% to 69.8% in these periods. The 300 basis point decrease in year over year MRO segment gross profit margins is principally due to higher product costs, including surcharges from steel vendors, and a sales mix shift, with more than one-third of the sales increase from sales of lower margin non-stock product category items. OEM segment gross profit margins declined 160 basis points, from 27.2% to 26.6%. This decline was principally due to price competition and higher product costs internationally, more than offsetting improved U.S. gross profit margins resulting from price increases and surcharges passed on to customers in the second quarter.

#### OPERATING EXPENSES AND OPERATING INCOME

Operating expenses were \$118.4 million and \$112.8 million for the six months ended June 30, 2005 and 2004, respectively. The \$5.6 million, or 5.0% increase, in year-to-date operating expenses for 2005 over the same period in 2004 was principally related to higher sales agent and employee compensation costs. The \$2.1 million increase in sales agent compensation was attributable to the increase in sales. A \$2.1 million employee compensation increase was driven by higher compensation costs associated with the Long-Term Capital Accumulation Plan, implemented in the third quarter of 2004, and marketing and technology personnel additions. As a result of the sales increases and the Company's ability to leverage its operating cost infrastructure over a larger revenue base, operating expenses as a percent of net sales declined from 55.0% in the 2004 period to 53.7% of net sales in the 2005 period.

Operating income for the six-month period ended June 30, 2005 was \$16.9 million, compared to \$18.3 million in the prior year-to-date period. The \$1.4 million decrease in operating income over these periods is principally attributable to lower gross profit margins and operating expense increases, which more than offset the gains in net sales.

#### **INVESTMENT AND OTHER INCOME**

The following table presents investment and other income for the six months ended June 30, 2005 and 2004:

(Dollars in millions)	2005	2004
Rental income	\$0.9	\$0.9
Realized foreign exchange gains	0.4	
Interest and other	0.1	0.2
	\$1.4	\$1.1

The realized foreign exchange gains for the six months ended June 30, 2005 was due to payments of intercompany payable balances by the Company's Canadian subsidiary.

#### INTEREST EXPENSE

Interest expense is primarily attributable to interest paid relative to a mortgage payable for the building owned by the limited partnership referred to in Note I.

#### PROVISION FOR INCOME TAXES

The effective tax rates for the six months ended June 30, 2005 and 2004 were 42.0% and 38.5%, respectively. Approximately 2.1 percentage points of the 3.5 percentage point increase in the effective tax rate over these periods is due to higher operating losses from the foreign subsidiaries in the OEM segment in 2005, for which no tax benefit was provided, and to a lesser extent, lower exempt income in 2005 related to executive life insurance also contributed slightly to the increased effective tax in the 2005 year-to-date period compared to the prior year period.

#### NET INCOME

Net income for the first six months of 2005 was \$10.6 million (\$1.15 per diluted share), compared to \$11.8 million (\$1.25 per diluted share) in the similar period of 2004. The \$1.2 million decline in net income is the result of the \$1.4 million decrease in operating income and a higher effective tax rate discussed above.

# LIQUIDITY AND CAPITAL RESOURCES

Net cash provided by operating activities for the six months ended June 30, 2005 and 2004 was \$2.3 million and \$13.2 million, respectively. In 2005, net cash provided by operating activities was impacted by lower net income, higher inventories and lower accrued expenses. Net cash used in financing activities increased by approximately \$8.8 million related to higher treasury stock purchases for the first six months of 2005. Working capital at June 30, 2005 was \$112.2 million as compared to \$115.0 million at December 31, 2004. At June 30, 2005 the current ratio was 3.9 to 1 as compared to 3.7 to 1 at December 31, 2004.

Additions to property, plant and equipment were \$1.2 million and \$2.1 million, respectively, for the six months ended June 30, 2005 and 2004. In addition, approximately \$3.2 million was expended for software development in the first six months of 2005 and recorded as prepaid

expenses, a portion of which may be reclassified to property, plant and equipment in the second half of 2005. Capital expenditures in 2005 were principally related to the purchase of machinery and equipment. In 2004, capital expenditures were principally related to the purchase of equipment and the development of software. The Company expects to spend approximately \$6.0 million for capital expenditures in 2005.

In the second quarter of 2005, the Company announced a cash dividend increase of \$.02 to \$.20 per share on common shares. This was an 11.1% increase over the previous \$.18 per share on common shares paid each quarter in 2004.

During the first six months of 2005, the Company purchased 248,614 shares of its own common stock at a cost of approximately \$11,196,000. Of those shares, 37,163 shares were purchased pursuant to the 2000 Board authorization for up to 500,000 shares and 211,451 shares were purchased pursuant to the October 2004 authorization. In the first six months of 2004, the Company purchased 91,434 shares of its own common stock for approximately \$2,964,000, pursuant to the 2000 Board authorization noted above. All shares purchased as of June 30, 2005 have been retired. Funds to purchase these shares were provided by cash and cash equivalents and net cash provided by operating activities. There is no expiration date relative to the 2004 share repurchase authorization.

Net cash provided by operating activities, current cash and cash equivalents and the \$75,000,000 unsecured line of credit are expected to be sufficient to finance the Company's future growth, cash dividends and capital expenditures for the next 12 months.

#### Item 3. Quantitative and Qualitative Disclosures About Market Risk

There have been no material changes in market risk at June 30, 2005 from that reported in the Company's Annual Report on Form 10-K for the year ended December 31, 2004.

#### **Item 4. Controls and Procedures**

The Company's chief executive officer and chief financial officer have concluded, based on their evaluation as of the end of the period covered by this report, that the Company's disclosure controls and procedures (as defined in the Securities Exchange Act of 1934 Rules 13a-15(e) and 15d-15(e)) are effective to ensure that information required to be disclosed in the reports that the Company files or submits under the Securities Exchange Act of 1934 is recorded, processed, summarized and reported within the time periods specified in the Securities and Exchange Commission's rules and forms, and that such information is accumulated and communicated to our management, including our chief executive officer and chief financial officer, as appropriate, to allow timely decisions regarding financial disclosures. There was no change in the Company's internal control over financial reporting that occurred during the quarter ended June 30, 2005 that has materially affected or is reasonably likely to materially affect, the Company's internal control over financial reporting.

## PART II - OTHER INFORMATION

Items 1, 3, and 5 are inapplicable and have been omitted from this report.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

Period	(a) Total Number of Shares Purchased	(b) Average Price Paid per Share	(c) Total Number of Shares Purchased as Part of Publicly Announced Plans or Programs	(d) Maximum Number of Shares that May Yet Be Purchased Under the Plans or Programs
01/01/05 - 01/31/05	86,357	48.72	86,357	450,806
02/01/05 - 02/28/05	9,515	48.54	9,515	441,291
03/01/05 - 03/31/05	22,100	45.65	22,100	419,191
04/01/05 - 04/30/05	37,657	44.93	37,657	381,534
05/01/05 - 05/31/05	53,139	40.80	53,139	328,395
06/01/05 - 06/30/05	39,846	41.60	39,846	288,549
Total	248,614	45.03	248,614	288,549

On May 16, 2000, the Board of Directors of the Company authorized the purchase of up to 500,000 shares of its own common stock. Purchases pursuant to that authorization were completed in January 2005. On October 13, 2004, the Company announced that its Board of Directors authorized a stock repurchase program to purchase up to 500,000 shares of its common stock in addition to that previously authorized. There is no expiration date relative to the outstanding authorization.

## Item 4. Submission of Matters to a Vote of Security Holders.

- (a) The annual meeting of stockholders of Lawson Products, Inc. was held on May 10, 2005.
- (b) Pursuant to Instruction 3 to Item 4, no response is required to this item.
- (c) At the Annual Meeting conducted on May 10, 2005, the stockholders voted on the election of directors and on approval of the Lawson Products, Inc. Long-Term Cap Accumulation Plan. A summary of the votes is as follows:

	For	Withheld Authority
Ronald B. Port	8,607,602	163,305
Robert G. Rettig	8,555,537	215,370
Wilma J. Smelcer	8,595,094	175,813

Messers. Brophy, Saranow and Shaffer continue to serve as directors of the Company for terms ending in 2006 and Messers. Hillman, S. Port and Washlow continue to serve as directors of the Company for terms ending in 2007.

	<u>For</u>	<u>Against</u>	<u>Abstain</u>	Broker Non-Vote
Lawson Products, Inc. Long- Term Capital	7,712,462	283,726	11,468	763,251
Accumulation Plan				

#### Item 6. Exhibits.

- 10.17 Credit Agreement dated March 27, 2001 between Lawson Products, Inc. and LaSalle Bank National Association, as amended by the First Amendment to Credit Agreement dated August 12, 2002, as amended by Second Modification to Loan Documents dated July 11, 2003, and as further amended by Third Modification to Credit Agreement dated as of June 15, 2005
- 15 Letter from Ernst & Young LLP Regarding Unaudited Interim Financial Information
- 31.1 Certification of Chief Executive Officer pursuant to 18 U.S.C. Section 1350 as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002
- 31.2 Certification of Chief Financial Officer pursuant to 18 U.S.C. Section 1350 as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002
- 32 Certification of Chief Executive Officer and Chief Financial Officer pursuant to 18 U.S.C. Section 1350 as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

# **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

LAWSON PRODUCTS, INC. (Registrant)

Dated August 9, 2005 /s/ Robert J. Washlow

Robert J. Washlow

Chief Executive Officer and Chairman of the Board

Dated August 9, 2005 /s/ Thomas Neri

Thomas Neri

Executive Vice President, Chief Financial Officer, and Treasurer

#### CREDIT AGREEMENT

This Credit Agreement together with all Exhibits and Schedules attached hereto and hereby made a part hereof ("Agreement") is made as of the 27th day of March, 2001, by and between Lawson Products, Inc., a Delaware Corporation ("Lawson"), with its principal place of business and chief executive office at 1666 E. Touhy Ave., Des Plaines, Illinois, 60018, various Subsidiaries of Lawson listed on Schedule 6.12 hereof (Lawson and the Subsidiaries may be referred to herein collectively as the "Borrower"), and LASALLE BANK NATIONAL ASSOCIATION (the "Lender").

#### PRELIMINARY STATEMENTS:

- 1. The Borrower has requested the Lender to provide it with a multi-currency revolving credit facility in an aggregate amount of up to \$50,000,000.00.
- 2. The Borrower will use the proceeds of the facility to provide for its working capital requirements, general corporate purposes and for planned and future acquisitions including, but not limited, to Lawson's planned acquisition of the assets of IDP North America, a division of Premier Farnell Corporation.
  - 3. The Lender is willing to make such a facility available to Borrower based on the terms and subject to the conditions set forth in this Agreement.
- 4. Each Subsidiary has determined that its joint and several liabilities under this Agreement are in furtherance of its corporate or other organizational purposes and in its best interest and that it will derive substantial benefit, whether directly or indirectly, from entering into such obligations by, among other things, enabling (i) each Subsidiary to receive proceeds from the Loan to be used as working capital, (ii) each Subsidiary to directly receive proceeds for capital expenditures or indirectly receive capital assets from capital expenditures made with the proceeds received by other Subsidiary and (iii) each Subsidiary to obtain additional capital in the future by direct borrowing or from the proceeds of borrowings of a Subsidiary or Affiliate.

#### AGREEMENT

For and in consideration of the foregoing, which is made a part hereof, the mutual promises, covenants and conditions contained herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

#### DEFINITIONS.

1.1 <u>General Terms</u>. When used herein, the following terms shall have the following meanings:

"Account Debtor" shall mean the party who is obligated on or under an Account. "Accounting Systems Letter" shall have the meaning set forth in subsection 7.1(G).

- "Accounts" shall mean all present and future rights of the Borrower to payment for goods sold or leased or for services rendered, which are not evidenced by instruments or chattel paper, and whether or not they have been earned by performance.
- "Acquisition" means the purchase by Lawson of all of the assets from IPD North America, a division of Premier Farnell Corporation pursuant to the terms of the Acquisition Documentation.
  - "Acquisition Documentation" means, collectively, all agreements, documents, instruments and orders executed and delivered in connection with the Acquisition.
- "Affiliate" shall mean any Person (a) that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, the Borrower, (b) that directly or beneficially owns or holds 25% or more of any class of the voting stock of the Borrower, or (c) 25% or more of the voting stock (or in the case of a Person which is not a corporation, 25% or more of the equity interest) of which is owned directly or beneficially or held by the Borrower. As used in this definition, "control" (including with correlative meanings the terms "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise. Unless the context otherwise clearly requires, any reference to an "Affiliate" is a reference to an Affiliate of the Borrower.
- "Authorized Officer" shall mean, at any time, an individual whose signature has been certified to Lender on behalf of Borrower pursuant to a Signature Authorization Certificate actually received by Lender at such time and whose authority has not been revoked prior to such time in the manner prescribed in such Signature Authorization Certificate.
  - "Bankruptcy Code" shall have the meaning set forth in subsection 6.6 hereof.
  - "Business Day" shall mean a day other than Saturday or Sunday on which banks in Chicago are open for the transaction of banking business.
- "Capital Expenditure" shall mean, as to any Person, any and all expenditures of such Person for fixed or capital assets, including, without limitation, the incurrence of capitalized lease obligations, all as determined in accordance with GAAP, except that capital expenditures shall not include expenditures for fixed or capital assets to the extent such expenditures are paid or reimbursed from the proceeds of insurance.
- "Capitalized Lease" shall mean, as to any Person, at any time, any lease which, in accordance with GAAP, is required to be capitalized on the consolidated balance sheet of such Person at such time, and "capitalized lease obligations" of such Person at any time shall mean the aggregate amount which, in accordance with GAAP, is required to be reported as a liability on the consolidated balance sheet of such Person at such time as lessee under Capitalized Leases.
- "Change in Control" shall mean the failure of Sidney L. Port or his Immediate Family to own individually, or through a trust or other entity for their benefit, including but not limited to Port Investments LLP, a Delaware Limited Liability Partnership, twenty percent (20%) or more of the shares of stock of Lawson.

- "Closing Date" shall mean March 27, 2001, the closing of the Loan pursuant to the Agreement.
- "Code" shall have the meaning set forth in subsection 1.3 hereof.
- "Current Assets" and "Current Liabilities" "Current Assets" shall mean the amount of all current assets (exclusive of prepaid expenses) which would be classified as such on a balance sheet prepared in accordance with GAAP. "Current Liabilities" shall mean the amount of all current liabilities which would be classified as such on a balance sheet prepared in accordance with GAAP, and which include, without limitation, trade debt, all accrued liabilities, the current portions of long-term debt and Capitalized Lease obligations.
- "Debt Service Coverage Ratio" shall mean EBITDA (as hereinafter defined) less unfunded Capital Expenditures, divided by current maturities of long term debt plus interest expense plus dividends.
  - "Deposit Inventory System" shall have the meaning set forth in Schedule 3.4 hereof.
  - "Default" shall mean the occurrence or existence of any one or more of the events described in subsection 9.1 hereof.
- "Environmental Lien" shall mean a lien in favor of any governmental entity for (a) any liability under federal or state environmental laws or regulations, or (b) damages arising from costs incurred by such governmental entity in response to a release of a hazardous or toxic waste, substance or constituent, or other substance into the environment.
  - "ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended from time to time, any successor statute, and any regulations promulgated thereunder.
- "ERISA Affiliate" shall mean with respect to Lawson (i) any corporation which is a member of the same controlled group of corporations (within the meaning of Section 414(b) of the Internal Revenue Code) as Lawson; (ii) a trade or business under common control (within the meaning of Section 414(c) of the Internal Revenue Code) with Lawson; or (iii) a member of the same affiliated service group (within the meaning of Section 414(m) of the Internal Revenue Code) as Lawson.
  - "Event of Default" shall mean any event that, if it continues uncured, will, with lapse of time or notice or both, constitute a Default.
- "EBITDA" for any period, shall mean the Borrower's pretax net income (determined on a consolidated basis in accordance with GAAP) before interest, tax distributions, dividends, state replacement tax expense, depreciation, amortization expense (of intangibles, including capitalized fees and goodwill) and other noncash expenses determined in accordance with GAAP.

- "Financials" shall have the meaning set forth in subsection 6.4 hereof.
- "Fiscal Quarter" shall mean a period, three (3) months in duration, beginning on January 1, April 1, July 1, or October 1 of any Fiscal Year.
- "Fiscal Year" shall mean a period, twelve (12) months in duration commencing on January 1 and ending on December 31.
- "GAAP" shall mean generally accepted accounting principles as in effect on the date hereof in accordance with the rules, regulations, pronouncements and opinions of the Financial Accounting Standards Board and the American Institute of Certified Public Accountants (or their successors), and as applied in a manner consistent with preparation of the Financials, subject to the provisions of subsection 1.2.

"Good Faith" shall have the meaning set forth for that term in Section 1-201(19) of the Code, provided that Good Faith shall also mean the absence of malice or capriciousness on the part of Lender.

"Guaranteed Indebtedness" of any Person means all Indebtedness referred to in the definition of "Indebtedness" in this Section guaranteed directly or indirectly in any manner by such Person, or in effect guaranteed directly or indirectly by such Person (or secured by any assets of such Person) regardless of whether the liability of such Person is limited to such assets or otherwise nonrecourse through an agreement (i) to pay or purchase such Indebtedness or to advance or supply funds for the payment or purchase of such Indebtedness; (ii) to purchase, sell or lease (as lessee or lessor) property, or to purchase or sell services, primarily for the purpose of enabling the debtor to make payment of such Indebtedness or to assure the holder of such Indebtedness against loss; (iii) to supply funds to, or in any other manner invest in, the debtor (including any agreement to pay for property or services irrespective of whether such property is received or such services are rendered); or (iv) otherwise to assure a creditor against loss or to grant a security interest in property for the benefit of any such creditor.

"Immediate Family" shall mean the spouse, former spouse, children, grandchildren, parents or grandparents of a Person.

"Indebtedness" of any Person means (without duplication), as of any specified date, the aggregate amount outstanding or owing under (a) all indebtedness of such Person for borrowed money or for the deferred purchase price of property or services (including, without limitation, all obligations in respect of principal, premium, if any, and interest payable on such indebtedness and all other obligations, contingent or otherwise, of such Person and in connection with any agreement to purchase, redeem, exchange, convert or otherwise acquire for value any capital stock of, or other equity interest in, such Person or any other Person), but excluding current liabilities for trade payables and other current liabilities other than for money borrowed, incurred in the ordinary course of business; (b) all obligations of such Person evidenced by bonds, notes, debentures or other similar instruments; (c) all indebtedness created or arising under any conditional sale or other title retention agreement with respect to property acquired by such Person (even though the rights and remedies of the seller or Lender under such agreement in the event of default are limited to repossession or sale of such property); (d) all obligations of such Person under Capitalized Leases; (e) all Indebtedness referred to in clause (a), (b), (c) or (d) above secured by (or for which the holder of such Indebtedness has an existing right, contingent or otherwise, to be secured by) any lien, security interest or other charge or encumbrance upon or in property (including, without limitation, accounts and contract rights) owned by such Person, even though such Person has not assumed or become liable for the payment of such Indebtedness; (f) all Guaranteed Indebtedness of such Person; (g) all liabilities incurred by such Person or any ERISA Affiliates; and (i) all increase in the amount of contributions required to be made by such Person and its ERISA Affiliates in each fiscal year of such Person to Multiemployer Plans, due to the reorganization or termin

"Internal Revenue Code, shall mean the Internal Revenue Code of 1986, as amended from time to time, and any successor statute.

"Inventory" shall mean any and all goods including, without limitation, goods in transit, wheresoever located, whether now owned or hereafter acquired by the Borrower, which are held for sale or lease, furnished under any contract of service or held as raw materials, work-in-process or supplies, and all materials used or consumed in the Borrower's business, and shall include such property the sale or other disposition of which has given rise to Accounts and which has been returned to or repossessed or stopped in transit by the Borrower.

"Legal Requirement" shall mean any requirement imposed upon Lender by any law of the United States of America or the United Kingdom or by any regulation, order, interpretation, ruling of official directive (whether or not having the force of law) of the Bank of England or any other board, central bank or governmental or administrative agency, institution or authority of the United States of America, the United Kingdom or any political subdivision of either thereof.

"Lending Affiliate" means (a) each office and branch of the Lender, and (b) each entity which, directly or indirectly, is controlled by or under common control with the Lender or which controls the Lender and each office and branch thereof.

"Liabilities" shall mean all of the Borrower's liabilities, obligations, and indebtedness to the Lender of any and every kind and nature, whether heretofore, now or hereafter owing, arising, due or payable and howsoever evidenced, created, incurred, acquired, or owing, whether primary, secondary, direct, contingent, fixed or otherwise (including obligations of performance) and whether arising or existing under written agreement, oral agreement or operation of law, and all of the Borrower's other indebtedness and obligations to the Lender under this Agreement and the other Loan Documents.

"LIBOR" shall mean the interest rate determined by the following formula, rounded upward to the nearest 1/100 of one percent (all amounts in the calculations will be determined by Lender as of the first day of the interest period:

LIBOR = London Inter-Bank Offered Rate (1.00 - Reserve Percentage)

#### Where,

- (a) "London Inter-Bank Offered Rate" means the rate per annum equal to the offered rate for deposits in U.S. dollars for the applicable interest period and for amounts comparable to the LIBOR Portion published by Bloomberg's Financial Markets Commodities News at approximately 8:00 a.m. Chicago time three (3) Business Days before the commencement of the interest period (or if not so published, Lender, in its sole discretion, shall designate another daily financial or governmental publication of national circulation to determine such rate); provided, however, that after the first election of an interest period with respect to any LIBOR Portion, the London Inter-Bank Offered Rate shall be determined at approximately 8:00 a.m. Chicago time on the first Business Day of the month for each interest period thereafter with respect to such LIBOR Portion.
- (b) "Reserve Percentage" means the total of the maximum reserve percentages for determining the reserves to be maintained by member banks of the Federal Reserve System for Eurocurrency Liabilities, as defined in Federal Reserve Board Regulation D, rounded upward to the nearest 1/100 of one percent.

The percentage will be expressed as a decimal and will include, but not be limited to, marginal, emergency, supplemental, special, and other reserve percentages.

- "LIBOR Based Rate" shall mean LIBOR plus seventy-five (75) basis points.
- "LIBOR Based Rate Advances" shall mean advances under the Loan bearing interest at rates based upon the LIBOR Based Rate.
- "LIBOR Based Rate Interest Period" shall mean a period of 30, 60, 90 or 180 days, as selected by Lawson, commencing on a Business Day selected by Lawson pursuant to this Agreement. All LIBOR Based Rate Interest Periods shall be subject to the following additional conditions: (i) each selection of an Interest Period shall be irrevocable for the period so selected; (ii) each Interest Period shall be selected in such a way that no Interest Period shall extend beyond the Maturity Date; and (iii) if any Interest Period ends on a day other than a Business Day (a day of the year on which dealings are carried on in the London interbank market and banks are not required or authorized to close in Chicago), such Interest Period shall be extended to the next succeeding day which is a Business Day unless such succeeding day would fall in the next calendar month, in which event such Interest Period shall end on the immediately preceding Business Day.
- "LIBOR Option" shall mean the option granted pursuant to subsection 2.6(B) to have the interest on all or any portion of the principal amount of the Loan based on a LIBOR Based Rate.
- "LIBOR Portion" that portion of the Loan specified in a LIBOR Request which is not less than \$1,000,000 and an integral multiple of \$250,000 which does not exceed the outstanding balance of the Loan not already subject to a LIBOR Option and, which, as of the date of the LIBOR Request specifying such LIBOR Portion, has met the conditions for basing interest on the LIBOR Based Rate in subsection 2.6(B) hereof and the LIBOR Based Rate Interest Period of which was commenced and not terminated.

"LIBOR Request" shall mean a notice in writing (or by telephonic communication confirmed by telex, telecopy or other facsimile transmission on the same day as the telephone request) from Lawson to Lender requesting that interest on a LIBOR Loan be based on the LIBOR Based Rate, specifying: (i) the first day of the LIBOR Based Rate Interest Period, (ii) the length of the LIBOR Based Rate Interest Period consistent with the definition of that term, and (iii) a dollar amount of the LIBOR Portion consistent with the definition of such terms.

- "Liquid Assets" shall mean the aggregate amount of Borrower's cash, marketable securities, Accounts and Inventory.
- "Loan Account" shall mean the loan account established on the books of the Lender established pursuant to Section 2.4 hereof.
- "Loan" shall have the meaning set forth in subsection 2.1 hereof.

"Loan Documents" shall mean, collectively, all agreements, instruments and documents, including, without limitation, this Agreement and any security agreements, loan agreements, notes, guarantees, mortgages, deeds of trust, subordination agreements, pledges, powers of attorney, consents, assignments, intercreditor agreements, mortgagee waivers, reimbursement agreements, contracts, notices, leases, financing statements and all other written matter whether heretofore, now or hereafter executed by or on behalf of the Borrower and delivered to the Lender, in any case in connection with the Loans made hereunder, together with all agreements, documents or instruments referred to therein or contemplated thereby, including, without limitation, the Note.

- "Loan Maturity Date" shall have the meaning set forth in subsection 2.8(B). hereof.
- "Management Letter" shall have the meaning set forth in subsection 7.1(G) hereof.
- "Maximum Facility" shall mean the maximum amount which the Lender has agreed to consider as a ceiling on the outstanding principal balance of loans and other extensions of credit to be made to or for the account of the Borrower under this Agreement. The Maximum Facility shall be Fifty Million Dollars (\$50,000,000.00) United States currency, unless permanently reduced at Lawson's election pursuant to Section 2.3.
- "Multiemployer Plan" shall mean, with respect to any Person, an employee benefit plan defined in Section 4001(a) (3) of ERISA which is, or within the immediately preceding six (6) years was, contributed to by such Person or an ERISA Affiliate of such Person.
  - "Negative Pledge" shall have the meaning set forth in subsection 5.1 hereof.
- "Negative Pledge Assets" shall mean all property and interests in property now owned or hereafter acquired by the Borrower in or upon which a Negative Pledge is granted to the Lender by the Borrower, whether under this Agreement, the other Loan Documents, or under any other documents, instruments or writings executed by the Borrower, and delivered to the Lender.

"Net Income" shall have the meaning provided, and shall be calculated provided in accordance with GAAP.

- "Note" shall have the meaning set forth in subsection 2.1 hereof.
- "PBGC" shall mean the Pension Benefit Guaranty Corporation.
- "Pension Plan" shall mean any employee pension benefit plan as defined in Section 3(2) of ERISA in respect of which Lawson or any ERISA Affiliate is, or at any time during the immediately preceding five (5) years was, an "employer" as defined in Section 3(5) of ERISA.
  - "Permitted Liens" shall mean only those liens, claims and encumbrances set forth upon Schedule 6.5 hereto.
- "Person" shall mean any individual, sole proprietorship, partnership, limited liability company, joint venture, trust, unincorporated organization, association, corporation, institution, entity, party, or government (whether national, federal, state, provincial, county, city, municipal or otherwise, including, without limitation, any instrumentality, division, agency, body or department thereof).
  - "Prime Based Rate" shall mean the Prime Rate minus 150 basis points.
  - "Prime Based Rate Advance" shall mean advances under the Loan bearing interest at rates based upon the Prime Based Rate.
- "Prime Rate" shall mean the rate of interest most recently announced by Lender at Chicago, Illinois as its prime or base rate. A certificate made by an officer of Lender stating the Prime Rate in effect on any given day, for the purposes hereof, shall be conclusive evidence of the Prime Rate in effect on such day. The "Prime Rate" is a base reference rate of interest adopted by Lender as a general benchmark from which Lender determines the floating interest rates chargeable on various loans to borrowers with varying degrees of creditworthiness and Borrower acknowledges and agrees that Lender has made no representations whatsoever that the "Prime Rate" is the interest rate actually offered by Lender to borrowers of any particular creditworthiness.
  - "Prime Portion" shall mean that portion of the Loan not subject to a LIBOR Option.
  - "Property" shall have the meaning set forth in subsection 10.8(B).
- "Real Property" means all of the Borrower's right, title, and interest in all of those plots, pieces or parcels of land now owned, leased or hereafter acquired by the Borrower (the "Land"), including, without limitation, those more particularly described on Exhibit A1, together with the right, title and interest of the Borrower in and to the following: the streets, the land lying in the bed of any streets, roads or avenues, opened or proposed, in front of, adjoining, or abutting the Land to the center line thereof, the air space and development rights pertaining to the Land and right to use such air space and development rights, all rights of way, privileges, liberties, tenements, hereditaments and appurtenances belonging or in any way appertaining thereto, all easements now or hereafter benefitting the Land and all royalties and all rights appertaining to the use and enjoyment of the Land, including, without limitation, all alley, vault, drainage, mineral, water, oil, and gas rights, timber, sewers, pipes, conduits, wires, and other facilities furnishing utility or other services to the Land and other similar rights, together with all of the buildings and other improvements and fixtures now or hereafter erected on the Land.

"Responsible Officer" shall mean (i) any one or more of the officers of Borrower listed on Exhibit F, (ii) an officer elected or appointed by the Board of Directors of a Borrower to replace an officer listed on Exhibit F, or (iii) an officer not listed on Exhibit F but having substantially the same or similar responsibilities as an officer so listed on Exhibit F, including, but not limited to, officers of newly created or acquired subsidiaries.

"Signature Authorization Certificate" shall mean a certificate substantially in the form attached hereto as Exhibit B now or hereafter executed on behalf of Lawson and delivered to Lender.

"Subsidiary" shall mean, with respect to any Borrower, any corporation or other entity of which more than fifty percent (50%) of the outstanding capital stock or other membership interest having ordinary voting power to elect a majority of the board of directors or controlling body of such corporation or entity (irrespective of whether at the time stock of any other class or classes of such corporation shall have or might have voting power by reason of the happening of any contingency) is at the time, directly or indirectly, owned by such Borrower.

"Tax" in relation to any LIBOR Portion and the applicable LIBOR Based Rate, any tax, levy, impost, duty, deduction, withholding or other charges of whatever nature required by any Legal Requirement (i) to be paid by Lender and/or (ii) to be withheld or deducted from any payment otherwise required hereby to be made by Borrower to Lender; provided, that the term "Tax" shall not include any taxes imposed upon the net income of Lender by the United States of America or any political subdivision thereof.

"Term" shall mean the period from the Closing Date to the Loan Maturity Date.

"Termination Event" shall mean (i) with respect to any Pension Plan, the occurrence of a reportable event described in Section 4043 of ERISA and the regulations issued thereunder; or (ii) the withdrawal of the Borrower or any ERISA Affiliate from a Pension Plan during a plan year in which it is a "substantial employer" as defined in Section 4001(a) (2) of ERISA; or (iii) the occurrence of an obligation of the Borrower or any ERISA Affiliate arising under Section 4041 of ERISA to provide participants in a Pension Plan and other affected parties with a written notice of intent to terminate the Pension Plan; or (iv) the institution of proceedings to terminate a Pension Plan by the Pension Benefit Guaranty Corporation; or (v) any other event or condition which might constitute grounds under Section 4041(A) or 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Pension Plan; or (vi) the partial or complete withdrawal of the Borrower or any ERISA Affiliate from a "Multiemployer Plan" (as defined in subsection 6.18 hereof).

"Third Party Goods" shall mean all raw materials, work-in-process and finished goods owned by Persons other than the Borrower and in Borrower's possession.

"UFCA" shall mean the Uniform Fraudulent Conveyances Act.

"UFTA" shall mean the Uniform Fraudulent Transfers Act.

"Withdrawal Liability" shall have the meaning given to such term under Part I of the Subtitle E of Title IV of ERISA.

1.2 Accounting Terms. Calculations and determinations of financial and accounting terms used and not otherwise specifically defined hereunder shall be made and determined, both as to classification of items and as to amount, in accordance with GAAP. If any changes in accounting principles or practices from GAAP are occasioned by the promulgation of rules, regulations, pronouncements and opinions by or required by the Financial Accounting Standards Board or the American Institute of Certified Public Accountants (or any successor thereto or agencies with similar functions), which result in a change in the method of accounting in the calculation of financial covenants, standards or terms contained in this Agreement or any other Financing Agreement, the parties hereto agree to enter into negotiations to amend such provisions so as equitably to reflect such changes to the end that the criteria for evaluating Borrower's financial condition and performance will be the same after such changes as they were before such changes, and if the parties fail to agree on the amendment of such provisions, Borrower shall continue to provide calculations for all financial covenants, perform all financial covenants and otherwise observe all financial standards and terms in accordance with applicable accounting principles and practices in effect immediately prior to such changes.

Calculations with respect to financial covenants required to be stated in accordance with applicable accounting principles and practices in effect immediately prior to such changes shall be reviewed and certified by Borrower's independent certified public accountants.

- 1.3 Other Terms Defined in Illinois Uniform Commercial Code. All other terms contained in this Agreement (and which are not otherwise specifically defined herein) shall have the meanings provided in the Uniform Commercial Code of the State of Illinois (the "Code") to the extent the same are used or defined therein.
- 1.4 Other Definitional Provisions. Whenever the context so requires, the neuter gender includes the masculine and feminine, the singular number includes the plural, and vice versa.

#### CREDIT.

#### 2.1 <u>Loan, Purpose, etc.</u>

(A) Provided there does not then exist a Default or an Event of Default, subject to the provisions of Section 4 below, and subject to the other provisions and conditions of this Agreement, the Lender agrees, following the Borrower's execution of this Agreement and all agreements and documents contemplated hereby, to advance to the Borrower on a revolving credit basis (collectively, the "Loan") the Maximum Facility. Each advance to the Borrower under this subsection 2.1 shall be in multiples of Two Hundred Fifty Thousand Dollars (\$250,000.00) and shall, on the day of such advance, be deposited, in immediately available funds, in the Lawson's demand deposit account with the Lender, or in such other account with the Lender as Lawson may, from time to time, designate. The Loan made by the Lender to the Borrower under this subsection 2.1 shall be evidenced, in part, by a promissory note of even date herewith in the form attached hereto as Exhibit C (the "Note") with the blanks appropriately filled. The Liabilities evidenced by the Note shall become immediately due and payable, (i) as provided in subsection 9.1 hereof; (ii) without notice or demand upon termination of this Agreement pursuant to subsection 2.8 hereof.

- (B) Purpose. The proceeds of the Loan shall be used by Borrower for working capital requirements and general corporate purposes and for the Acquisition or future acquisitions.
  - (C) Limits. The aggregate outstanding principal balance of the Loan shall not at any time exceed the Maximum Facility.
- 2.2 <u>Prepayments</u>. Borrower shall at any time and from time to time have the right to prepay, in whole or in part, the principal amount of the Loan outstanding or interest due without any penalty or premium. Such prepayment shall be applied in the following order of priority: (1) amounts (other than principal and interest) due and payable to Lender under this Agreement, the Note, or the other Loan Documents; (2) accrued but unpaid interest on the outstanding principal balance of the Loan; and (3) the outstanding principal balance of the Loan.
- 2.3 <u>Facility Reduction</u>. Borrower may, at any time during the Term, elect to permanently reduce the Maximum Facility upon the following terms and conditions: (i) Lawson shall provide Lender an irrevocable written notice no less than thirty (30) days prior to the intended effective date of the permanent reduction to the Maximum Facility (the "Facility Reduction Notice"); (ii) the Facility Reduction Notice shall state the total amount of the intended reduction to the Maximum Facility ("Maximum Facility Reduction") and the effective date of the Maximum Facility Reduction; (iii) the aggregate outstanding principal balance of the Loan as of the date of the Facility Reduction Notice shall not exceed the Maximum Facility Reduction (the "Reduced Maximum Facility"); and (iv) the aggregate outstanding principal balance of the Loan subsequent to the Facility Reduction Notice shall not at any time thereafter exceed the Reduced Maximum Facility.

Notwithstanding anything to the contrary contained in this Section 3.2 to the contrary, the Maximum Facility Reduction shall in no event be less than five million dollars (\$5,000,000.00).

2.4 The Borrower's Loan Account. The Lender shall maintain a loan account (the "Loan Account") on its internal data control system in which shall be recorded (i) all loans and advances made by the Lender to the Borrower pursuant to this Agreement, (ii) all payments made by the Borrower on all such loans and advances, and (iii) all other appropriate debits and credits as provided in this Agreement, including, without limitation, all fees, charges, expenses and interest. All entries in the Borrower's Loan Account shall be made in accordance with the Lender's customary accounting practices as in effect from time to time. Subject to adjustments and objections (if any) pursuant to subsection 2.5 below, the Borrower promises to pay the amount reflected as owing by it under its Loan Account and all of its other obligations hereunder and under any other Loan Documents as such amounts become due (whether by scheduled maturity, required prepayment, acceleration, demand or otherwise) pursuant to the terms of this Agreement and the other Loan Documents.

2.5 <u>Statements.</u> All advances and other financial accommodations to the Borrower, and all other debits and credits provided for in this Agreement, shall be evidenced by entries made by the Lender in its internal data control systems showing the date, amount and reason for each such debit or credit. Until such time as the Lender shall have rendered to the Borrower written statements of account as provided herein, the balance in the Borrower's Loan Account, as set forth on the Lender's most recent printout, shall be rebuttably presumptive evidence of the amounts due and owing the Lender by the Borrower. Not more than twenty (20) days after the last day of each Fiscal Quarter, the Lender shall render to the Borrower a statement setting forth the balance of the Borrower's Loan Account, including principal, interest, expenses and fees. Each such statement shall be subject to subsequent adjustment by the Lender but shall, absent manifest errors or omissions, be rebuttably presumed correct and shall constitute an account stated unless, within thirty (30) days after receipt of such statement from the Lender, the Borrower shall deliver to the Lender written objection thereto specifying the error or errors, if any, contained in such statement. Lender shall, upon request, provide supporting detail as to third party charges, such as attorneys' fees.

#### 2.6 <u>Interest and Fees</u>.

Interest Rate. The Borrower shall pay to the Lender interest, as follows:

- (A) Interest.
  - (i) Prime Portion. Interest shall accrue on the Prime Portion outstanding at the end of each day (computed on the basis of a calendar year of 360 da per annum equal to the Prime Based Rate. After the date hereof, the foregoing rate of interest shall be increased or decreased, as the case may b any increase or decrease in the Prime Based Rate, with such adjustments to be effective as of the opening of business on the day that any such c Rate becomes effective. The Prime Based Rate in effect on the date hereof shall be the Prime Based Rate effective on the opening of business or this Agreement is executed on a day that is not a Business Day, the Prime Based Rate in effect on the date hereof shall be the Prime Based Rate opening of business on the last Business Day immediately preceding the date hereof.
  - (ii) LIBOR Portion. Interest shall accrue on each LIBOR Portion outstanding at the end of each day (computed on the basis of a calendar year of 36 the sum of the LIBOR Based Rate applicable to each such LIBOR Portion.

- (B) LIBOR Option.
  - (i) Conditions for Basing Interest for the Loan on the LIBOR Based Rate. Upon the condition that:
    - (a) The Lender shall have received a LIBOR Request from Lawson by 12:00 p.m. on the first Business Day immediately prior to the first Based Rate Interest Period requested:
    - (b) There shall have occurred no change in applicable law which would make it unlawful for the Lender to obtain deposits of U.S. dollar foreign currency deposits market;
    - (c) As of the date of the LIBOR Request and the first day of the LIBOR Based Rate Interest Period, there shall exist no Default or Event either case has not been waived by the Lender; and
    - (d) The Lender shall not have determined in Good Faith that the Lender is unable to determine the LIBOR Based Rate in respect of the r Rate Interest Period or that the Lender is unable to obtain deposits of U.S. dollars in the London interbank foreign currency deposits amounts and for the requested LIBOR Based Rate Interest Period;
  - (ii) Indemnification for Funding and Other Losses. Each LIBOR Request shall be irrevocable and binding on Borrower. Borrower shall indemnify Lender fo Lender to the extent required as a result of Borrower's failure to fulfill, on or before the date specified in any LIBOR Request, the applicable conditions: Agreement, including, without limitation, any loss or expense incurred by reason of the liquidation or redeployment of deposits or other funds acquired to maintain the requested LIBOR Portion, when, as a result of such failure on the part of Borrower, interest on such LIBOR Portion is not based on the application for the requested LIBOR Based Rate Interest Period.
  - (iii) Change in Applicable Laws, Regulations, etc. If any Legal Requirement shall make it unlawful for Lender to fund through the purchase of U.S. dollar de Portion, or otherwise to give effect to its obligations as contemplated under this subsection 2.6(B) or shall impose on Lender any costs based on or meast a specified level of the amount of a category of deposits or other liabilities of Lender which includes deposits by reference to which the LIBOR Based Raprovided herein or a category of extensions of credit or other assets of Lender which includes any LIBOR Portion, or shall impose on Lender any restrict such a category of liabilities or assets of Lender which includes any LIBOR Portion, or shall impose on Lender any restrictions on the amount of such a cassets which Lender may hold, (i) Lender may by notice thereof to Lawson prospectively terminate the LIBOR Option, with respect to Loans made or to any LIBOR Portion of Lender's Loans subject thereto shall immediately bear interest thereafter at the Prime Based Rate payable on the dates provided in and (iii) Borrower shall indemnify Lender against any loss, penalty or expense incurred by Lender by reason of the liquidation or redeployment of depos acquired by Lender to fund or maintain such LIBOR Portion.

(iv) Taxes. It is the understanding of Borrower and Lender that Lender shall receive payments of amounts of principal of and interest on the Note with respect to LIBOR Portions from time to time subject to a LIBOR Option free and clear of, and without deduction for, any Taxes. If (i) Lender shall be subject to any such Tax in respect of any such LIBOR Portion or any part thereof or (ii) Borrower shall be required to withhold or deduct any such Tax from any such amount, the LIBOR Based Rate applicable to such LIBOR Portion shall be adjusted by Lender to reflect all additional costs incurred by Lender in connection with the payments by Lender or the withholding by Borrower of such Tax and Borrower shall provide Lender with a statement detailing the amount of any such Tax actually paid by Borrower. Determination by Lender of the amount of such costs shall, in the absence of manifest error, be conclusive, and at Borrower's request, Lender shall demonstrate the basis of such determination. If after any such adjustment, any part of any Tax paid by Lender is subsequently recovered by Lender, Lender shall reimburse Borrower to the extent of the amount so recovered. A certificate of an officer of Lender setting forth the amount of such recovery and the basis therefor shall, in the absence of manifest error, be conclusive.

- (C) Default Rate of Interest. Upon and after the occurrence of a Default and during the continuation thereof, the principal amount of the Loan then outstanding shall bear interest, calculated daily (computed on the actual days elapsed over a year of 360 days), at a fluctuating rate per annum equal to (i) three percent (3%) above the Prime Based Rate (the "Default Base Rate"). (The rate of interest calculated pursuant to this subsection 2.6(C) shall be referred to herein as the "Post Default Rate").
- (D) Maximum Interest. In no contingency or event whatsoever shall the aggregate of all amounts deemed interest hereunder or under the Note and charged or collected pursuant to the terms of this Agreement or pursuant to the Note exceed the highest rate permissible under any law which a court of competent jurisdiction shall, in a final determination, deem applicable hereto. In the event that such a court determines that Lender has charged or received interest hereunder in excess of the highest applicable rate, the rate in effect hereunder shall automatically be reduced to the maximum rate permitted by applicable law and Lender shall promptly refund to Lawson any interest received by Lender in excess of the maximum lawful rate or, if so requested by Lawson, shall apply such excess to the principal balance of the Liabilities. It is the intent hereof that Borrower not pay or contract to pay, and that Lender not receive or contract to receive, directly or indirectly in any manner whatsoever, interest in excess of that which may be paid by Borrower under applicable law.
- (E) Payments. Except where evidenced by notes or other instruments issued or made by Borrower to Lender specifically containing payment provisions which are in conflict with this subsection 2.6(E) (in which event the conflicting provisions of said notes or other instruments shall govern and control), that portion of the Loan consisting of:
  - (1) Interest accrued on the Prime Portion shall be due on the earliest of (i) the first day of each Fiscal Quarter (for the immediately preceding Fiscal Quarter), computed through the last calendar day of the preceding Fiscal Quarter, (ii) the occurrence of a Default as a consequence of which Lender elects to accelerate the maturity and payment of the Liabilities or (iii) termination of this Agreement pursuant to subsection 2.8 hereof; provided, however, the Borrower hereby irrevocably authorizes Lender, in Lender's sole discretion, to charge Borrower's demand deposit account with Lender, a sum sufficient each Fiscal Quarter to pay all interest accrued on the Prime Portion during the immediately preceding month.
  - (2) Interest on the LIBOR Portion shall be due on the earliest of (i) the last day of each LIBOR Based Rate Interest Period relating to such LIBOR Portion (but in no event less often than on the first day of each Fiscal Quarter), (ii) the occurrence of a Default in consequence of which Lender elects to accelerate the maturity and payment of the Liabilities, or (iii) termination of this Agreement pursuant to subsection 2.8, provided, however, that Borrower irrevocably authorizes Lender, in Lender's sole discretion, to charge Borrower's demand deposit account with Lender at the termination of each LIBOR Based Rate Interest Period, a sum sufficient to pay all interest accrued and payable with respect to each LIBOR Portion.

- (3) Reasonable costs, fees and expenses payable pursuant to this Agreement shall be payable to Lender by Borrower within five (5) Business Days after written notice from Lender, to Lawson or to such other Person designated in writing by Lawson.
- (F) Number of Portions: At no time shall there exist more than six (6) separate LIBOR Portions during the Term.
- (G) Fee Reimbursement. Borrower shall reimburse Lender, whether or not the Loan is funded, an amount not to exceed Seven Thousand Dollars (\$7,000.00) for Lender's reasonable costs and expenses, including, but not limited to attorney's fees, and due diligence search costs and fees.
- (H) Unused Commitment Fee. Beginning on the first anniversary of the date of this Agreement and throughout the Term, the Borrower shall pay to Lender a fee (the "Unused Commitment Fee") on the last day of each Fiscal Quarter beginning with the Fiscal Quarter ending on June 30, 2002, equal to one eighth of one percent (1/8%) per annum of the face amount of the average daily unused Maximum Facility that exceeds Five Million Dollars (\$5,000,000.00) during such Fiscal Quarter. The Unused Commitment Fee shall be computed on the basis of a 360-day year for the actual number of days elapsed.
- 2.7 <u>Method for Making Payments</u>. All payments of principal and interest hereunder shall be paid by automatic debit, wire transfer, check or in coin or currency which, at the time or times of payment, is the legal tender for public and private debts in the United States of America and shall be made at such place as Lender or the legal holder or holders of the Note may from time to time appoint in the payment invoice or otherwise in writing, and in the absence of such appointment, then at the offices of Lender at 135 South LaSalle Street, 11th Floor, Chicago, Illinois 60603. Payment made by check shall be deemed paid on the date Lender receives such check; provided, however, that if such check is subsequently returned to Lender unpaid due to insufficient funds or otherwise, the payment shall not be deemed to have been made and shall continue to bear interest until collected. Notwithstanding the foregoing, the final payment due hereunder must be made by wire transfer or other final funds. If requested by Lawson, interest, principal payments and any fees and expenses owed Lender from time to time will be deducted by Lender automatically on the due date from the designated Borrower's account with Lender, as designated in writing by Lawson. Borrower will maintain sufficient funds in the account on the dates Lender enters debits authorized under the Note. If there are insufficient funds in the account on the date Lender enters any debit authorized by the Note, the debit will be reversed. Lawson may terminate this direct debt arrangement at any time by sending written notice to Lender at the address specified above.

#### 2.8 Maturity, Term.

- (A) Loan Maturity. The Loan, including the full outstanding principal balance thereon and all accrued and then unpaid interest thereon, if not sooner paid, shall be immediately due and payable without notice or demand on March 27, 2006 (the "Loan Maturity Date").
- (B) Termination. This Agreement shall terminate at the end of the Term; provided, however, that the Lender shall retain the right to terminate this Agreement sooner at any time upon the occurrence and only during the continuance of a Default; and further provided, however, that notwithstanding any such termination (i) all of the Lender's rights and remedies under this Agreement and the Borrower's Negative Pledge pursuant to subsection 5.1 herein shall survive such termination until all of the Liabilities have been fully paid and satisfied. Notwithstanding the foregoing, Lawson may by written notice to Lender terminate this Agreement at any time as provided above conditioned upon and subject to the prior payment by Borrower to Lender of all then outstanding principal and accrued interest and payment and performance of all other Liabilities. Upon the effective date of termination of this Agreement, all of the Liabilities (other than contingent and indemnity obligations) shall become immediately due and payable without notice or demand. Notwithstanding any termination, until all of the Liabilities shall have been fully paid and satisfied and all Loan Documents between the Borrower and the Lender shall have been terminated, all of the Lender's rights and remedies under this Agreement and the other Loan Documents shall survive, and the Borrower shall continue to be obligated under the Negative Pledge contained herein in and to all existing and future Negative Pledge Assets, as hereinafter defined.

## 3. <u>REPORTING AND ELIGIBILITY REQUIREMENTS.</u>

Account Warranties. Borrower warrants and represents to the Lender that: (i) to Borrower's knowledge, the Accounts are genuine, are in all respects what they purport to be, and are not evidenced by a judgment; (ii) the Accounts represent undisputed, bona fide transactions the performance of which has been completed by the Borrower; (iii) the amounts shown on the Borrower's books and records and all invoices and statements actually and absolutely owing to the Borrower and are not in any way contingent; (iv) to the best of Borrower's knowledge, there are no setoffs, counterclaims or disputes and the Borrower has not made any agreement with any Account Debtor for any deduction therefrom except returns, discounts or allowances for prompt payment allowed by the Borrower in the ordinary course of its business which are not likely to result in any material adverse change in Borrower's financial condition or business operations; (v) to the best of Borrower's knowledge, there are no facts, events or occurrences known to the Borrower which in any way impair the validity or enforcement thereof or tend to reduce the amount payable thereunder as shown on the Borrower's books and records; (vi) to the Borrower's knowledge, and except to the extent of the allowance for doubtful Accounts shown on the Financials, all Account Debtors have the capacity to contract and are solvent; (vii) the services furnished and/or goods sold giving rise thereto are not subject to any lien, claim, encumbrance or security interest except that of the Lender and except as specifically permitted below; (viii) except to the extent of the allowance for doubtful Accounts shown on the Financials, to the Borrower's knowledge, there are no proceedings or actions which are threatened or pending against any Account Debtor which are likely to result in any material adverse change in such Account Debtor's financial condition. The Borrower agrees to notify the Lender with respect to any Accounts with respect to which the warranties in this subsection 3.1 are no

- 3.2 <u>Verification of Accounts</u>. The Lender shall have the right, at any time or times hereafter, in the Lender's name or in the name of a nominee of the Lender, to verify the validity, amount or any other matter relating to any Accounts, by mail, telephone, telegraph or otherwise.
- 3.3 Account Covenants. The Borrower shall promptly upon the Borrower's learning thereof: (i) inform the Lender in writing of any material delay in the Borrower's performance of any of its material obligations to any Account Debtor or of any assertion of any material claims, setoff or counterclaims by any Account Debtor; (ii) furnish to and inform the Lender of all material adverse information of which the Borrower obtains knowledge relating to the financial condition of any Person who is then an Account Debtor as to open Accounts with a face amount, in the aggregate, in excess of \$1,000,000.00.
- 3.4 <u>Inventory Warranties.</u> With respect to Inventory scheduled, listed or referenced in any report pursuant to subsection 7.1 herein, and to the extent not otherwise stated herein, Borrower warrants in all material respects that (i) it is located at one of the premises listed on Exhibit A1 or Exhibit A2 and is not in transit or is subject to the Deposit Inventory System more fully described on Schedule 3.4; (ii) it is not subject to a lien or security interest whatsoever except for the Negative Pledge granted to Lender hereunder and except as specifically permitted in subsection 8.1 below; and (iii) to the best of Borrower's knowledge, it is of good and merchantable quality, free from any defects which would materially adversely affect the market value of such Inventory. Borrower agrees to notify Lender with respect to any of its Inventory with respect to which the warranties in this subsection 3.4 are not true.
- 3.5 <u>Inventory Records</u>. The Borrower shall at all times hereafter maintain a perpetual inventory, keeping correct and accurate records itemizing and describing the kind, type, quality and quantity of Inventory, and Borrower's cost therefor and daily withdrawals therefrom and additions thereto, all of which records as prepared at the last inventory count in the normal course of business, shall be available at the applicable Borrower's location without prior notice, or at Lawson's location upon five (5) Business Days prior written notice during Borrower's usual business hours at the request of any of Lender's officers, employees or agents. Borrower shall continue to make cycle counts and other physical counts of Inventory substantially as may be required by its independent public accountants and by good accounting practices (and, following the occurrence of a Default, Borrower shall make such additional counts as may be reasonably requested by Lender) and, upon Lender's request, promptly following any such counts of inventory shall supply Lender with a report in a form and with such specificity as may be reasonably satisfactory to Lender concerning such physical count of the Inventory.
- 3.6 <u>Safekeeping of Inventory and Inventory Covenants.</u> The Lender shall not be responsible for: (i) the safekeeping of the Inventory; (ii) any loss of or damage to the Inventory; (iii) any diminution in the value of the Inventory; or (iv) any act or default of any carrier, warehouseman, bailee, forwarding agency or any other Person. All risk of loss, damage, destruction or diminution in value of the Inventory shall be borne by Borrower. Except as expressly set forth in this Agreement or pursuant to the Deposit Inventory System, no Inventory shall, without Lender's prior written consent, be at any time or times hereafter stored with a bailee, warehouseman, consignee or similar third party.

Except for the Deposit Inventory System, Borrower shall not sell any of its Inventory on a bill-and-hold, guaranteed sale, sale-or-return, sale on approval or consignment basis or any other basis subject to a repurchase obligation or return right.

The Borrower shall not sell any of its Inventory on a bill-and-hold basis if it would cause the aggregate outstanding amount of all Accounts arising therefrom to exceed at any time \$50,000.00 with respect to all such Account Debtors. No Inventory shall be at any time or times hereafter stored with a bailee, warehouseman, consignee or similar third party unless the Borrower first (i) obtains the Lender's written consent as to the identity of such third party, and (ii) furnishes to the Lender such agreements, instruments and documents as the Lender shall in its sole discretion specify with respect to such stored Inventory.

3.7 Third Party Goods. Borrower shall not hold or accept any Third Party Goods, whether on a consignment basis or otherwise, except from a Person approved by Lender, provided that, with respect to any other such Person, Lender has (i) been furnished with an accurate description of the arrangements and agreements (if not reduced to writing) with respect to such Third Party Goods, together with true, correct and complete copies of any written instruments, documents or agreements with respect thereto, and (ii) received an executed intercreditor agreement from such Person in form and substance satisfactory to Lender. If any of the arrangements or instruments, documents or agreements referred to in this subsection 3.7 are amended, modified or supplemented, Borrower promptly shall provide Lender with notice thereof and copies of such amendment, modification or supplement. Schedule 3.7 attached hereto sets forth a complete listing of all Persons now having an interest in any Third Party Goods. Borrower shall take such action as Lender may from time to time require to assure Lender that no Person storing any Third Party Goods with Borrower has any interest in the Negative Pledge Assets.

## 4. CONDITIONS OF ADVANCES.

Notwithstanding any other provisions contained in this Agreement the making of any Loan shall be conditioned upon the following:

- 4.1 <u>Borrower's Written Request</u>. (i ) as to any Prime Based Rate Advance, the Lender shall have received by 12:00 Noon (Chicago time) on the first (1st) Business Day prior to the date such advance is to be made, a written request (or telephonic request promptly confirmed in writing) from an Authorized Officer of Lawson for such an advance specifying the principal amount thereof; (ii) as to any LIBOR Based Rate Advance, Lawson shall comply with subsection 2.6(B). In addition, prior to making any advance or loan, Lender shall have received copies of all documents required to have been delivered to the Lender pursuant to this Agreement (including, without limitation, subsections 5.2 and 7.1).
- 4.2 <u>Financial Condition</u>. No material adverse change, as determined by the Lender in its sole discretion, in the financial condition or operations of the Borrower shall have occurred (a) at any time or times subsequent to the most recent annual financial statements provided pursuant to subsection 7.1(B) of this Agreement and (b) prior to the receipt of the first of such statements, at any time subsequent to receipt of the Financials.

- 4.3 No Default. There shall not have occurred any Default or an Event of Default which is then continuing, nor shall any such Default or Event of Default occur after giving effect to the advance or loan.
- 4.4 <u>Representations and Warranties True and Correct</u>. The representations and warranties of Borrower contained in this Agreement shall be true and correct in all material respects on and as of the date of any advance or loan, as though made on and as of such date, except for any waivers thereof expressly granted by an officer of Lender in writing delivered to Lawson.
- 4.5 <u>Other Requirements</u>. The Lender shall have received, in form and substance satisfactory to the Lender, all certificates, orders, authorities, consents, affidavits, schedules, opinions, instruments, security agreements, financing statements, mortgages and other documents which are required hereunder, or which the Lender may at any time reasonably request.
- 4.6 <u>Conditions as to Initial Advance</u>. Prior to the initial loan made hereunder, each of the conditions set forth on Exhibit D hereto shall be fully performed in form and substance satisfactory to Lender and its legal counsel.

#### NEGATIVE PLEDGE.

- 5.1 Negative Pledge. Borrower shall not, and shall not permit any of its Subsidiaries who are not parties hereto to create, incur, permit, or suffer to exist any lien, pledge or negative pledge ("Negative Pledge") except for the Negative Pledge made in favor of Lender herein and any Permitted Liens, upon the following assets: (i) Borrower's Accounts; (ii) Borrower's Inventory; (iii) all insurance proceeds of or relating to any of the foregoing; (iv) all of the Borrower's books and records relating to the foregoing; (v) all accessions and additions to, substitutions for, and replacements, products and proceeds of any of the foregoing, whether now owned or hereafter acquired (collectively the "Negative Pledge Assets").
- 5.2 <u>Preservation of Negative Pledge Assets.</u> The Borrower shall execute and deliver to the Lender at any time or times at the request of the Lender, all financing statements, instruments or other documents (and pay the cost of filing or recording the same in all public offices deemed necessary by the Lender) as the Lender may request, in a form satisfactory to the Lender, to preserve and keep preserved the Negative Pledge granted by the Borrower to the Lender. Should the Borrower fail to do so, the Lender is authorized to sign any such financing statements as the Borrower's agent. The Borrower further agrees that, to the extent permitted by law, a carbon, photographic, photostatic or other reproduction of this Agreement or of a financing statement is sufficient as a financing statement.
- 5.3 Setoff. Borrower agrees that Lender has all rights of setoff and banker's lien provided by applicable law and, in addition thereto, Borrower agrees that at any time any Default exists, Lender may apply to the payment of the Liabilities any and all balances, credits, deposits, accounts or moneys of Borrower then or thereafter with Lender. Without limitation of the foregoing, Borrower agrees that, upon the occurrence and during the continuance of a Default, Lender and each of its branches and offices is hereby authorized, at any time and from time to time, without notice, (i) to setoff against, and to appropriate and apply to the payment of, the Liabilities (whether matured or unmatured, fixed or contingent or liquidated or unliquidated) any and all amounts owing by Borrower to Lender or any such office or branch of Lender (whether matured or unmatured, and, in the case of deposits, whether general or special, time or demand and however evidenced) and (ii) pending any such action, to the extent necessary, to hold such amounts as collateral to secure such Liabilities and to return as unpaid for insufficient funds any and all checks and other items drawn against any deposits so held as Lender may elect in its sole discretion exercised in Good Faith.

#### WARRANTIES ETC.

Each Borrower jointly and severally represents, warrants and agrees, except to the extent not applicable to such Borrower that, as of the date hereof and each day thereafter, continuing so long as the Liabilities remain outstanding, and (even if there shall be no Liabilities outstanding) so long as this Agreement remains in effect:

- Existence. (i) Lawson is a corporation, duly organized and in good standing under the laws of the State of Delaware and in good standing in Illinois and all other states where the nature and extent of the business transacted by it or the ownership of its assets makes such qualification necessary, except for those jurisdictions in which the failure so to qualify would not, in the aggregate, have a material adverse effect on Lawson's financial condition, results of operations or business or the ability of Lawson to perform its obligations hereunder; (ii) each Subsidiary listed on Schedule 6.12 is the business entity type as indicated on Schedule 6.12, duly organized and in good standing under the laws of the state or country of its organization and all other states or countries where the nature and extent of the business transacted by it, or the ownership of its assets makes such qualification necessary, except for those jurisdictions in which the failure so to qualify would not, in the aggregate, have a material adverse effect on such Subsidiary's financial condition, results of operations or business or the ability of such Subsidiary to perform its obligations hereunder.
- 6.2 Entity Authority. The execution and delivery by each Borrower hereunder of this Agreement and all of the other Loan Documents executed by it and the performance of the Borrower's obligations hereunder and thereunder: (i) are within the Borrower's corporate, company or other entity powers; (ii) are duly authorized by the Borrower's Directors, Managers or the equivalent, and, to the extent required, Shareholders, Members, Partners or the equivalent; (iii) are not in contravention of the terms of the Borrower's Articles of Incorporation and by-laws, Articles of Organization and Operating Agreement, or other such similar entity formation and operating agreement, or of any indenture, or other material agreement or undertaking to which the Borrower is a party or by which the Borrower or any of its property is bound or any judgment, decree or order applicable to Borrower; (iv) do not, as of the execution hereof, require the Borrower to obtain any governmental consent, registration or approval; (v) do not contravene any contractual or governmental restriction binding upon the Borrower; and (vi) will not, except as contemplated herein, result in the imposition of any lien, charge, security interest or encumbrance upon any property of the Borrower under any existing indenture, mortgage, deed of trust, loan or credit agreement or other material agreement to which the Borrower is a party or by which it or any of its property may be bound or affected.

- 6.3 <u>Binding Effect</u>. This Agreement and all of the other Loan Documents to which each Borrower is a party are the legal, valid and binding obligations of such Borrower and are enforceable against such Borrower, as applicable, in accordance with their respective terms.
- Financial Data. Lawson has furnished to the Lender consolidated financial statements as of December 31, 1999 and consolidated financial statements as of various dates subsequent to December 31, 1999, including, without limitation, financial statements as of September 30, 2000 (collectively, the "Financials"). The Financials are, and all financial statements to be furnished to the Lender in accordance with subsection 7.1 below will be, in accordance with the books and records of the Borrower and fairly present, and, as to all financial statements to be furnished to the Lender in accordance with subsection 7.1 below, will fairly present, the financial condition of the Borrower at the dates thereof and the results of operations for the periods indicated (subject, in the case of unaudited financial statements, to normal year-end adjustments), and such Financials and financial statements have been and will be prepared in conformity with generally accepted accounting principles consistently applied throughout the periods involved. Since the date of the Financials, there have been no changes in the condition, financial or otherwise, of the Borrower as shown on such Financials, except (a) as expressly contemplated herein, and (b) for changes in the ordinary course of business (none of which individually or in the aggregate has been materially adverse). All information, reports and other papers and data to be furnished to the Lender are or will be, at the time the same are so furnished to the Lender, accurate and correct in all material respects and complete insofar as completeness may be necessary to give the Lender a true and accurate knowledge of the subject matter thereof.
- 6.5 Negative Pledge Assets. Except as expressly permitted by subsections 8.1 and 8.17, all of the Borrower's Negative Pledge Assets are and will continue to be owned by the Borrower (except for Goods and Inventory sold in the ordinary course of Borrower's business), have been or will (in the ordinary course of Borrower's business) be fully paid for and are free and clear of all security interests, liens, pledges, negative pledges, claims and encumbrances, except the Negative Pledge in favor of Lender and Permitted Liens, as set forth on Schedule 6.5 hereto. The Negative Pledge Assets are located at the locations set forth on Exhibit A1 or Exhibit A2 attached hereto, except for Inventory in transit or at processors or packagers in conformity with the terms of the Agreement and the Inventory sold through the Deposit Inventory System which is located at Borrower's Customer's Customer's locations.
- 6.6 Solvency. As of the date hereof after giving effect to the transaction contemplated herein, Lawson (i) is not "insolvent" as that term is defined in Section 101(32) of the Federal Bankruptcy Code (the "Bankruptcy Code") (11 U.S.C. ss. 101(32)), Section 2 of the Uniform Fraudulent Transfer Act ("UFTA") or Section 2 of the Uniform Fraudulent Conveyance Act ("UFCA"); (ii) does not have "unreasonably small capital," as that term is used in Section 548 (a) (2) (B) (ii) of the Bankruptcy Code or Section 5 of the UFCA; (iii) is not engaged or about to engage in a business or a transaction for which its remaining property is "unreasonably small" in relation to the business or transaction as that term is used in Section 4 of the UFTA; (iv) is able to pay its debts as they mature or become due in the ordinary course within the meaning of Section 548(a) (2)(B) (iii) of the Bankruptcy Code, Section 4 of the UFTA and Section 6 of the UFCA; and (v) now owns assets having a value both at "fair valuation" and at "present fair salable value" greater than the amount required to pay Lawson's "debts" in the ordinary course as such terms are used in Section 2 of the UFTA and Section 2 of the UFCA. Lawson shall not be rendered insolvent (as defined above) by the execution and delivery of this Agreement on March 27, 2001, or any of the other Loan Documents or by the transactions contemplated hereunder or thereunder.

- 6.7 <u>Chief Place of Business.</u> As of the execution hereof, the principal place of business and chief executive office of Lawson is located at the address set forth above in the preamble to this Agreement and of each Subsidiary at the address set forth in Schedule 6.12. If any change in such locations occur, Lawson shall promptly notify the Lender thereof in accordance with subsection 10.17 hereof. As of the execution hereof, the books and records of each Borrower and all records of account are located at Lawson's, or the principal place of business and chief executive office of such Borrower, and if any change in such location occurs, Lawson shall promptly notify the Lender thereof in accordance with subsection 10.17 hereof.
- 6.8 Other Corporate Names. Except as disclosed on Schedule 6.8 attached hereto, the Borrower has not used any corporate or fictitious name other than the names shown for Lawson and certain of the Subsidiaries on Schedule 6.12.
- 6.9 Tax Liabilities. The Borrower has filed all federal, state and local tax reports and returns required by any law or regulation to be filed by it except those for which extensions have been duly obtained. The Borrower has either duly paid all taxes, duties and charges indicated due on the basis of such returns and reports, other than those being contested in good faith and except as set forth in Schedule 6.9 or has made adequate provision for the payment thereof, and the assessment of any material amount of additional taxes in excess of those paid and reported is not reasonably expected. No federal income tax returns of Borrower have been audited by the Internal Revenue Service other than audits which did not have a material adverse effect on Borrower. The reserves for taxes, if any, reflected on the Financials constitute, and the consolidated balance sheets of the Borrower submitted to the Lender in accordance with the terms of subsection 7.1 below will constitute, reasonable estimations of the amount necessary for the payment of all liabilities for all federal, state and local taxes (whether or not disputed) of the Borrower accrued through the date of such balance sheets. There are no material unresolved questions or claims concerning any tax liability of the Borrower.
- 6.10 <u>Margin Security</u>. The Borrower does not own any margin securities and none of the loans advanced hereunder will be used for the purpose of purchasing or carrying any margin securities or for the purpose of reducing or retiring any indebtedness which was originally incurred to purchase any margin securities or for any other purpose not permitted by Regulation U of the Board of Governors of the Federal Reserve System.
- 6.11 <u>Survival of Warranties</u>. All representations and warranties contained in this Agreement or any of the other Loan Documents shall survive the execution and delivery of this Agreement.
  - 6.12 Subsidiaries. All of Borrower's Subsidiaries, including the principal place of business and chief executive office thereof, are listed on Schedule 6.12.

- 6.13 <u>Litigation and Proceedings</u>. Except as disclosed on Schedule 6.13 attached hereto, no judgments are outstanding against the Borrower nor is there now pending or, to the best of the Borrower's knowledge after reasonably diligent inquiry, threatened any litigation, contested claim, or governmental proceeding by or against the Borrower except judgments and pending or threatened litigation, contested claims and governmental proceedings set forth in and upon Schedule 6.13 hereto that exceed \$500,000 in the aggregate. To the best of Borrower's knowledge, the amount of liability set forth on Schedule 6.13 as to each suit listed thereon is the maximum amount of Borrower's liability under such suit.
- 6.14 <u>Other Agreements</u>. The Borrower is not in default under any material contract, lease, or commitment to which it is a party or by which it is bound except such defaults which are not likely to result in a materially adverse effect on Borrower's financial condition or business operation. The Borrower knows of no dispute regarding any contract, lease, or commitment which is material to the continued financial success and well-being of the Borrower.
- 6.15 <u>Employee Controversies</u>. There are no controversies pending or, to the best of the Borrower's knowledge after diligent inquiry, threatened or anticipated, between the Borrower and any of its employees, other than employee grievances arising in the ordinary course of business which are not, in the aggregate, material to the continued financial success and well-being of the Borrower. The Borrower has no accrued and unpaid liability to any of its employees arising under the Fair Labor Standards Act, as amended.

#### 6.16 Compliance with Laws and Regulations.

- (A) General Compliance. The execution and delivery by the Borrower of this Agreement and all of the other Loan Documents to which it is a party and the performance of the Borrower's obligations hereunder and thereunder are not in contravention of any law or laws applicable to Borrower. The Borrower is in compliance in all material respects with all laws, orders, regulations and ordinances of all federal, foreign, state and local governmental authorities relating to the business operations and the assets of the Borrower, except for laws, orders, regulations and ordinances the violation of which would not, in the aggregate, have a material adverse effect on the Borrower's financial condition, results of operations or business.
- (B) Environmental Compliance. The operations of the Borrower comply in all material respects with all applicable federal, state or local environmental, health and safety statutes and regulations. The Borrower has not received notice of any judicial or administrative proceeding alleging the violation of any federal, state or local environmental, health or safety statute or regulation by or pertaining to the Real Property, the Borrower or its property or operations or stating that the Borrower is the subject of any federal or state investigation evaluating whether any remedial action is needed to respond to a release of any hazardous or toxic waste, substance, material or constituent, or other substance into the environment which has not been settled or resolved with such governmental agency. Except as expressly set forth on Schedule 6.13 hereto, the Borrower has not filed any notice under any federal or state law indicating past or present treatment, storage or disposal of a hazardous waste or reporting a spill or release of a hazardous or toxic waste, substance, material or constituent, or other substance into the environment. Except as expressly set forth on Schedule 6.13 hereto, the Borrower does not have any contingent liability of which the Borrower has knowledge or reasonably should have knowledge in connection with any release of any hazardous or toxic waste, substance, material or constituent, or other substance into the environment.

- 6.17 <u>Patents, Trademarks, Licenses, Etc</u>. The Borrower possesses adequate assets, licenses, patents, patent applications, copyrights, service marks, trademarks applications, trade styles and trade names, governmental approvals or other authorizations and other rights that are necessary for the Borrower to continue to conduct its business as heretofore conducted by it.
- ERISA. Neither Lawson nor any ERISA Affiliate of Lawson maintains or contributes to any Pension Plan other than a Pension Plan identified on Schedule 6.18 attached hereto. Each Pension Plan which is intended to be a qualified plan under Section 401(a) of the Internal Revenue Code has been determined by the Internal Revenue Service to be so qualified and each trust related to any such Pension Plan has been determined to be exempt from federal income tax under subsection 501(a) of the Internal Revenue Code or will be submitted to the Internal Revenue Service prior to the end of the remedial amendment period. Except as otherwise disclosed on Schedule 6.18 attached hereto, neither Lawson nor any ERISA Affiliate of Lawson maintains or contributes to any employee welfare benefit plan within the meaning of subsection 3(1) of ERISA which provides lifetime medical benefits to retirees. Each Pension Plan has been administered in all material respects in accordance with its terms and the terms of ERISA, the Internal Revenue Code and all other statutes and regulations applicable thereto. Neither Borrower nor any ERISA Affiliate of Borrower has breached in any material respect any of the responsibilities, obligations or duties imposed on it by ERISA or regulations promulgated thereunder with respect to any Pension Plan. No accumulated funding deficiency (as defined in subsection 302(a) (2) of ERISA and Section 412(a) of the Internal Revenue Code) exists in respect to any Pension Plan. Neither Lawson nor any ERISA Affiliate of Lawson nor any fiduciary of any Pension Plan which is not a Multiemployer Plan (i) has engaged in a nonexempt "prohibited transaction" described in Section 406 of ERISA or Section 4975 of the Internal Revenue Code which could result in any liability to Borrower, or (ii) has taken any action which would constitute or result in a Termination Event with respect to any Pension Plan which could result in any liability to Borrower, Schedule B. if any, to the most recent annual report filed with the Internal Revenue Service with respect to each Pension Plan has been furnished to Lender and is complete and accurate; since the date of each such Schedule B, there has been no material adverse change in the funding status or financial condition of the Pension Plan relating to such Schedule B. Neither Lawson nor any ERISA Affiliate of Lawson has incurred any liability to the PBGC which remains outstanding. Neither Lawson nor any ERISA Affiliate of Lawson has (i) failed to make a required contribution or payment to a Multiemployer Plan, or (ii) made or expects to make a complete or partial withdrawal under subsections 4203 or 4205 of ERISA from a Multiemployer Plan for which Lawson or any ERISA Affiliate of Lawson has any liability which has not been satisfied. Neither Lawson nor any ERISA Affiliate of Lawson has failed to make a required installment under subsection (m) of Section 412 of the Internal Revenue Code or any other payment required under Section 412 of the Internal Revenue Code on or before the due date for such installment or other payment. Neither Lawson nor any ERISA Affiliate of Lawson is required to provide security to a Pension Plan under Section 401(a) (29) of the Internal Revenue Code due to a Pension Plan amendment that results in an increase in current liability for the plan year. The present value of the benefits of each Pension Plan of Borrower and each ERISA Affiliate of the Borrower as of the last day of the year for such Plan, as determined by such Pension Plan's independent actuaries, does not exceed the aggregate value, as determined by such actuaries, of all assets under such Pension Plan. Borrower is not required to contribute to any Multiemployer Plan. No matter is pending relating to any Pension Plan before any court or governmental agency. Borrower has given to Lender all of the following: copies, if any, of each Pension Plan and related trust agreement (including all amendments to such Plan and trust) in existence or committed to as of the date hereof and the most recent summary plan description, actuarial report, determination letter issued by the Internal Revenue Service and Form 5500 filed in respect of each such Pension Plan; a listing of all of the Multiemployer Plans with the aggregate amount of the most recent annual contributions required to be made by Lawson and all ERISA Affiliates of Lawson to each such Multiemployer Plan; copies of any information which has been provided to Lawson or any ERISA Affiliate of Lawson regarding withdrawal liability under any Multiemployer Plan and all collective bargaining agreements pursuant to which such contributions are required to be made; and copies of each employee welfare benefit plan within the meaning of subsection 3(1) of ERISA which provides lifetime medical benefits to employees, the most recent summary plan description for such plan and the aggregate amount of the most recent annual payments made to terminated employees under each such plan.

- 6.19 <u>Financial Condition</u>. Since the date of the consolidated financial statements of Lawson, dated December 31, 1999 and the consolidated interim statement dated September 30, 2000, there has been no material adverse change in Borrower's financial condition, results of operations or business or in the value of Borrower's Negative Pledge Assets.
  - 6.20 Subordinated Debt. Borrower has no subordinated debt.
  - 6.21 Officers and Directors. The officers and directors of Lawson and each Subsidiary are set forth on Schedule 6.21.
- 6.22 <u>Certain Proceedings</u>. There are no pending or, to the best of the Borrower's knowledge upon due investigation, threatened eminent domain, condemnation, special assessment or other governmental proceedings pertaining to the Real Property which would have a material adverse effect on Borrower's financial condition or business operations.
- 6.23 No Violations. Except as set forth in Schedule 6.13, The Borrower has not received any written notice of, and has no actual knowledge of, violations of any zoning, building, fire or health code statutes or ordinances of any applicable governmental body existing upon the Real Property which have not been corrected and which would have a material adverse effect on Borrower's financial condition or business operations. Borrower's use of the Real Property is consistent with and allowed by applicable zoning law.
- 6.24 <u>Taxes.</u> All real estate taxes for the Real Property which have become due have been paid in full except any such taxes or charges which are being diligently contested in good faith by appropriate proceedings and do not cause a material adverse effect to Borrower's financial condition or business operations.
- 6.25 <u>Utilities</u>. All water, sewer, gas, electric, telephone, drainage and other utility equipment, facilities and services required or necessary for the operation of Real Property for the business to be conducted by the Borrower are installed and connected. To the best of the Borrower's knowledge, no fact, condition or proceeding exists which would result in the termination or impairment of the furnishing of such utility services to the Borrower and would have a material adverse effect on Borrower's financial condition or business operations.

#### 7. <u>AFFIRMATIVE COVENANTS</u>.

Each Borrower jointly and severally covenants and agrees, except to the extent not applicable to such Borrower that, so long as any of the Liabilities remain outstanding, and (even if there shall be no Liabilities outstanding) so long as this Agreement remains in effect:

- 7.1 <u>Financial Statements.</u> Borrower shall keep proper books of record and account in which full and true entries will be made of all dealings or transactions of or in relation to the business and affairs of Borrower, in accordance with GAAP. Lawson shall cause to be furnished to Lender in accordance with past practice consistently applied:
  - (A) Periodic Reporting. As soon as practicable, and in any event within forty-five (45) days after the end of each Fiscal Quarter:
    - (i) consolidated statements of income, retained earnings and cash flow of Borrower for such calendar month and for the period from the beginning of the then current Fiscal Year to the end of such Fiscal Quarter, and a consolidated balance sheet of Borrower as of the end of such Fiscal Quarter, setting forth in each case, in comparative form, figures (1) in the case of statements, for the corresponding periods in the preceding Fiscal Year and (2) in the case of balance sheets, as of a date one year earlier, all in reasonable detail and certified as accurate by an Authorized Officer pursuant to a certificate in the form of Exhibit E attached hereto, subject to changes resulting from normal year-end adjustments;
    - (ii) statements prepared in the ordinary course, if any, in which the actual cash flow and income for such Fiscal Quarter and for the period from the start of the then current Fiscal Year to the end of such Fiscal Quarter, and the actual balance sheets at the end of such Fiscal Quarter (in each case as required to be delivered pursuant to subsection 7.1(A) (i) hereof are compared with the corresponding projected statements of income and cash flow and balance sheets for such periods and time furnished to Lender pursuant to subsection 7.1 (F) below, in each case in the same format as the audited statements of income and cash flow and the audited balance sheet:

- (iii) (a) as reasonably requested by Lender, copies of consolidated operating statements for such Fiscal Quarter prepared by Borrower for internal use, including, without limitation, statements of cash flow, purchases and sales of inventory and other similar data, and (b) a comparison of actual cash flow and Capital Expenditures with amounts budgeted for such Fiscal Quarter;
- (iv) calculations setting forth the compliance with the financial covenants set forth in subsection 8.14 hereof for the most recently completed Fiscal Ouarter: and
- (v) in the event that any of the foregoing statements indicate that Borrower has varied in any material respect from any financial projections provided by Borrower to Lender, upon Lender's reasonable request a statement of explanation of such deviation from an Authorized Officer;
- (B) Annual. As soon as practicable and in any event within one hundred and ten (110) days after the end of each Fiscal Year of Borrower, consolidated statements of income, retained earnings and cash flow of Borrower for such Fiscal Year, and a consolidated balance sheet of Borrower as of the end of such Fiscal Year, setting forth in each case, in comparative form, corresponding figures for the period covered by the preceding annual statement (in the case of statements) and as of the end of the preceding Fiscal Year (in the case of balance sheets), all in reasonable detail and reasonably satisfactory in scope to Lender and audited by independent certified public accountants selected by Lawson and reasonably satisfactory to Lender, whose opinion shall be in scope and substance reasonably satisfactory to Lender and, if reasonably requested by Lender, Borrower shall cause such opinion to be addressed on its face to Lender or to be the subject of a reliance letter from such accountants permitting Lender to rely on the contents thereof as if prepared specifically for use by Lender;
- (C) Default Notices. As soon as practicable (but in any event not more than five (5) Business Days after any Responsible Officer of Borrower obtains knowledge of the occurrence of an event or the existence of a circumstance giving rise to an Event of Default or a Default), notice of any and all Events of Default or Defaults hereunder;
- (D) List of Account Debtors. (i) Upon and during the continuance of an Event of Default, within five (5) Business Days after the written request of Lender, Borrower shall deliver to Lender its most current Accounts aging report; (ii) upon a Default, within five (5) Business Days after the written request of Lender, Borrower shall deliver to the Lender the names, addresses and phone numbers of Borrower's Account Debtors;
- (E) Other Defaults and Material Information. As soon as practicable (but in any event not more than five (5) Business Days after any officer of Borrower obtains knowledge of the occurrence of an event or the existence of a circumstance giving rise to a Default or Event of Default under or a material violation of any term hereunder, notice of any such Default, Event of Default or violation, and immediately upon the receipt thereof, copies of any notices with respect thereto and any other material correspondence or information relating thereto shall be provided to Lender by Borrower;

- (F) Budget. To the extent prepared by Borrower in the ordinary course of business, or if reasonably requested in writing by Lender, Borrower shall (i) prepare an annual budget for the next Fiscal Year, in reasonable detail (on a fiscal month basis for the immediately succeeding Fiscal Year), and a detailed statement of the methods and assumptions used in the preparation of such budget or (ii) prepare a pro forma budget in the preparation for or investigation regarding an acquisition;
- (G) Letters from Accountants and Consultants. As soon as practicable and in any event within ten (10) days of delivery to Lawson, a copy of (i) to the extent that such letters may from time to time be issued by Borrower's independent certified public accountants, each "Management Letter" prepared by Borrower's independent certified public accountants in connection with the financial statements referred to in subsection 7.1(B) hereof and (ii) to the extent that such letters may from time to time be issued by Borrower's independent certified public accountants or other management consultants (collectively, "Accounting Systems Letters"), any letter issued by Borrower's independent certified public accountants or other management consultants with respect to recommendations relating to Borrower's financial or accounting systems or controls, and Borrower shall use its best efforts to cause each Management Letter and Accounting Systems Letter to be either addressed to Lender or accompanied by a reliance letter from such accountants permitting Lender to rely on the contents of each of the above as if prepared specifically for use by Lender (Borrower further agrees that upon Borrower's receipt of any Accounting Systems Letters wherein such accountants or consultants have made recommendations for improvements to Borrower's financial or accounting systems or controls, Borrower promptly shall commence actions to correct any material defects in such financial or accounting systems or controls unless Lender otherwise consents or Borrower reasonably disagrees with the need for such actions.);
  - (H) Other Information. With reasonable promptness, such other business or financial data as Lender may reasonably request.

All financial statements delivered to Lender pursuant to the requirements of this subsection 7.1 (except where otherwise expressly indicated) shall be prepared in accordance with GAAP (subject in the case of interim financial statements to the lack of footnotes and normal year-end adjustments) consistently applied, except for changes therein with which the independent certified public accountants issuing the opinion on the financial statements delivered pursuant to subsection 7.1(B) hereof have previously concurred in writing. Together with each delivery of financial statements required by subsections 7.1(A) and 7.1(B) hereof, Lawson shall deliver to Lender a certificate of an Authorized Officer of Lawson in the form attached hereto as Exhibit E setting forth in such detail as is reasonably acceptable to Lender calculations with respect to Borrower's compliance with each of the financial covenants contained in this Agreement and stating that to the best knowledge of the Authorized Officer upon due investigation there exists no Default or Event of Default, or, if any Default or Event of Default exists, specifying the nature and the period of existence thereof and what action Borrower proposes to take with respect thereto.

Lender shall exercise reasonable efforts to keep such information, and all information acquired as a result of any inspection conducted in accordance with subsection 7.2 hereof, confidential, provided that Lender may communicate such information (a) to any other Person in accordance with the customary practices of commercial banks relating to routine trade inquiries, (b) to any regulatory authority having jurisdiction over Lender, (c) to any other Person in connection with Lender's sale of any participations in the Liabilities or assignment of any rights and obligations of Lender under this Agreement and the other Loan Documents, (d) to any other Person in connection with the exercise of Lender's rights, prerogatives or performance hereunder or under any of the other Loan Documents, (e) to any Person in any litigation in which Lender is a party, or (f) to any Person if Lender believes in Good Faith that disclosure is necessary or appropriate to comply with any applicable law, rule or regulation or in response to a subpoena, order or other legal process or informal investigative demand, whether issued by a court, judicial or administrative or legislative body or committee or other governmental authority or (g) to any agents, employees, attorneys, accountants or appraisers of the Lender. Notwithstanding the foregoing, information shall not be deemed to be confidential to the extent such information (i) was already lawfully in the possession of Lender prior to such information being provided to Lender by Borrower, (ii) is available in the public domain, (iii) becomes available in the public domain other than as a result of unauthorized disclosure by Lender, or (iv) is acquired from a Person not known by Lender to be in breach of an obligation of secrecy to Borrower. Borrower authorizes Lender to discuss the financial condition of Borrower with Borrower's independent certified public accountants and other professional representatives.

- 1.2 Inspection. The Lender, or any Person designated by the Lender in writing, shall have the right, from time to time hereafter, to call at the Borrower's place or places of business (or any other place where the Negative Pledge Assets or any information relating thereto is kept or located) during reasonable business hours upon three (3) Business Days prior written notice, and, without hindrance or delay, (i) to inspect, audit, check and make copies of and extracts from the Borrower's books, records, journals, orders, receipts and any correspondence and other data relating to the Borrower's business or to any transactions between the parties hereto, (ii) to make such verification concerning the Negative Pledge Assets as the Lender may consider reasonable under the circumstances, and (iii) to discuss the affairs, finances and business of the Borrower with any officers, employees or directors of the Borrower. The Borrower shall pay on demand all reasonable photocopying expenses incurred by the Lender under this subsection 7.2.
- 7.3 <u>Conduct of Business</u>. Lawson shall maintain its legal existence as a Delaware Corporation, each Subsidiary shall maintain its legal existence as the State or County entity type stated on Schedule 6.12, Borrower shall maintain in full force and effect all material licenses, permits, authorizations, bonds, franchises, leases, patents, contracts and other rights necessary for the profitable conduct of its business, shall continue in, and limit its operations to, the same general line of business as that presently conducted by it and shall comply with all applicable laws and regulations of any federal, foreign, state or local governmental authority, except for such laws and regulations the violation of which would not, in the aggregate, have a material adverse effect on the Borrower's financial condition, results of operations or business or the Borrower's ability to perform its obligations.

- Claims and Taxes. The Borrower agrees to indemnify and hold Lender and each of its officers, directors, employees, attorneys and agents harmless from and against any and all claims, demands, liabilities, losses, damages, penalties, costs, and expenses (including without limitation reasonable attorneys' and consultants' fees) relating to or in any way arising out of the possession, use, operation or control of any of the Borrower's assets. The Borrower agrees that it shall pay or cause to be paid all license fees, bonding premiums and related taxes and charges, and shall pay or cause to be paid all of Borrower's real and personal property taxes, assessments and charges and all of Borrower's franchise, income, unemployment, use, excise, old age benefit, withholding, sales and other taxes and other governmental charges assessed against Borrower or payable by Borrower, at such times and in such manner as to prevent any penalty from accruing or any lien or charge from attaching to its property or the Real Property, provided that Borrower shall have the right to contest in good faith, by an appropriate proceeding promptly initiated and diligently conducted, the validity, amount or imposition of any such tax, assessment or charge, and during the pendency of such good faith contest to delay or refuse payment thereof, if (i) Borrower establishes adequate reserves to cover such contested taxes, assessments or charges, and (ii) such contest does not have a material adverse effect on Borrower's financial condition, results of operations or business, the ability of Borrower to pay any of the Liabilities, or the value to Lender of Borrower's Negative Pledge in the Negative Pledge Assets.
- 7.5 The Lender's Costs and Expenses as Additional Liabilities. The Borrower, agrees to reimburse the Lender promptly for all reasonable expenses and fees paid or incurred in connection with the analysis, documentation, negotiation and closing of the loans and other extensions of credit described herein, including, without limitation, lien search, filing and recording fees and the reasonable fees and expenses of the Lender's attorneys and paralegals and consultants (whether such attorneys and paralegals are employees of Lender or are separately engaged by Lender), whether such expenses and fees are incurred prior to or after the date hereof. All costs and expenses incurred by the Lender with respect to the negotiation, documentation, enforcement, collection and protection of the Lender's interests in the collateral shall be additional Liabilities of the Borrower to the Lender, payable on demand, repaid as provided in subsection 2.6 hereof.
- 7.6 The Borrower's Liability Insurance. The Borrower shall maintain, at its expense, such public liability and third party property damage insurance in such amounts and with such deductibles as is acceptable to the Lender in its reasonable discretion exercised in Good Faith.
- 7.7 The Borrower's Property Insurance and Business Interruption Insurance. The Borrower shall, at its expense, maintain business interruption insurance and keep and maintain its assets insured against loss or damage by fire, theft, burglary, pilferage, loss in transit, explosion, spoilage and all other hazards and risks and in such amounts as is ordinarily insured against by other owners or users of such properties in similar businesses. All such policies of insurance shall be in form and substance reasonably satisfactory to the Lender, and the Borrower shall not amend or otherwise change any such policies in any way which may materially adversely affect the Lender without the prior written consent of the Lender. The Borrower shall deliver to the Lender a certificate or other satisfactory evidence of each policy of insurance and evidence of payment of all premiums therefor. If the Borrower, at any time or times hereafter, shall fail to obtain or maintain any of the policies of insurance required above or to pay any premium in whole or in part relating thereto, then the Lender, without waiving or releasing any obligation or default by the Borrower hereunder, may at any time or times thereafter (but shall be under no obligation to do so) obtain and maintain such policies of insurance and pay such premiums and take any other action with respect thereto which the Lender deems advisable.

- 7.8 ERISA. Lawson shall deliver to Lender, at Lawson's expense, the following information as and when provided below:
  - (i) as soon as possible, and in any event within twenty (20) days after Lawson or an ERISA Affiliate of Lawson knows or has reason to know that a Termination Event has occurred, a written statement of an Authorized Officer of Lawson describing such Termination Event and the action, if any, which Lawson or such ERISA Affiliate of Lawson has taken, is taking or proposes to take with respect thereto, and when known, any action taken or threatened by the Internal Revenue Service ("IRS"), the Department of Labor ("DOL") or PBGC with respect thereto;
  - (ii) as soon as possible, and in any event within thirty (30) days, after Lawson or an ERISA Affiliate of Lawson knows or has reason to know that a prohibited transaction (defined in Section 406 of ERISA and Section 4975 of the Internal Revenue code) has occurred, a statement of an Authorized Officer of Lawson describing such transaction;
  - (iii) promptly after the filing thereof with the DOL, IRS or PBGC, copies of each annual report, including Schedule B thereto, filed with respect to each Pension Plan;
  - (iv) promptly after the filing thereof with the IRS, a copy of each funding waiver request filed with respect to any Pension Plan and all communications received by Lawson or any ERISA Affiliate of Lawson with respect to such request;
  - (v) promptly upon the occurrence thereof, notification of any increases in the benefits of any existing Pension Plan or the establishment of any new Pension Plan or the commencement of contributions to any Pension Plan to which Lawson or any ERISA Affiliate of Lawson was not previously contributing;
  - (vi) promptly upon, and in any event within ten (10) Business Days after, receipt by Lawson or an ERISA Affiliate of Lawson of the PBGC's intention to terminate a Pension Plan or to have a trustee appointed to administer a Pension Plan, copies of each such notice;

- (vii) promptly upon, and in any event within ten (10) Business Days after, receipt by Lawson or an ERISA Affiliate of Lawson of an unfavorable determination letter from the IRS regarding the qualification of a Pension Plan under Section 401(a) of the Internal Revenue Code, copies of such letter:
- (viii) promptly upon, and in any event within ten (10) Business Days after receipt by Lawson or an ERISA Affiliate of Lawson of a notice from a Multiemployer Plan regarding the imposition of withdrawal liability, copies of such notice; and (ix) promptly upon, and in any event within twenty (20) Business Days after either Lawson or an ERISA Affiliate of Lawson fails to make a required installment under subsection (m) of Section 412 of the Code or any other payment required under Section 412 on or before the due date for such installment or payment, a notification of such failure.

Lawson shall, and shall cause each of its ERISA Affiliates to, (a) keep in full force and effect any Pension Plans that are presently in existence or may, from time to time, come into existence, (b) make contributions to all Pension Plans in a timely manner and in a sufficient amount to comply with the requirements of the Pension Plans, the Code and ERISA, (c) comply with all requirements of ERISA and the Code which relate to all Pension Plans, and (d) notify Lender immediately upon receipt by Lawson or any of its ERISA Affiliates of any notice of the institution of any proceeding or other action which may result in the termination of any Pension Plan or where there may constitute a Termination Event. Lawson shall, and shall cause each of its ERISA Affiliates to, make any and all payments to any Multiemployer Plan that Lawson or any ERISA Affiliate thereof may be required to make under any agreement relating to any Multiemployer Plan or any law pertaining thereto, except for any such payments being contested in good faith by appropriate proceedings.

7.9 Notice of Suit or Adverse Change in Business. The Borrower shall, as soon as possible, and in any event within five (5) Business Days after any Responsible Officer of the Borrower learns of the following, give written notice to the Lender of: (i) any material proceeding(s) (including, without limitation, litigation, arbitration or governmental proceedings) being instituted or threatened to be instituted by or against the Borrower in any federal, state, local or foreign court or before any commission or other regulatory body (federal, state, local or foreign) which seeks an award against Borrower in an amount in excess of \$500,000 or would result in a materially adverse effect on the financial condition or operation of business of Borrower; (ii) notice that the Borrower's operations are in material noncompliance with requirements of applicable federal, state or local environmental, health and safety statutes and regulations; (iii) notice that the Borrower is subject to federal or state investigation evaluating whether any material remedial action is needed to respond to the release of any hazardous or toxic waste, substance or constituent, or other substance into the environment; (iv) notice that any material portion of the properties or assets of the Borrower or the Real Property is subject to an Environmental Lien; (v) any material adverse change in the business, assets or condition, financial or otherwise, of the Borrower; and (vi) any changes in the locations of any Negative Pledge Assets from the locations listed on Exhibit A1 or Exhibit A2 or otherwise permitted hereunder.

- 5.10 Supervening Illegality. If, at any time or times hereafter, there shall become effective any amendment to, deletion from or revision, modification or other change in any provision of any statute, or any rule, regulation or interpretation thereunder or any similar law or regulation, affecting, in the Lender's reasonable determination, the Lender's extension of credit described in this Agreement or the selling of participations therein, the Borrower shall either (i) pay to the Lender the then outstanding balance of the Liabilities which are in violation thereof, and hold the Lender harmless from and against any and all obligations, fees, liabilities, losses, penalties, costs, expenses and damages of every kind and nature imposed upon or incurred by the Borrower by reason of the Lender's failure or inability to comply with the terms of this Agreement or any of the other Loan Documents as a result of, but only to the extent of such change, or (ii) indemnify and hold the Lender harmless from and against any and all obligations, fees, liabilities, losses, penalties, costs, expenses and damages of every kind and nature imposed upon or incurred by the Lender by reason of such amendment, deletion, revision, modification, or other change as a result of, but only to the extent of such change. The obligations of the Borrower under this subsection 7.10 shall survive payment of the Liabilities and termination of this Agreement.
- Environmental Laws. If the Borrower shall (a) receive notice that any violation of any federal, state or local environmental law or regulation may have been committed or is about to be committed by the Borrower which will have a material adverse effect on Borrower's financial condition or business operations, (b) receive notice that any administrative or judicial complaint or order has been filed or is about to be filed against the Borrower alleging a material violation of any federal, state or local environmental law or regulation or requiring the Borrower to take any action in connection with the material release of toxic or hazardous substances into the environment, or (c) receive any notice from a federal, state, or local governmental agency or private party alleging that the Borrower may be liable or responsible for any material amount of costs associated with a response to or cleanup of a release of a toxic or hazardous substance into the environment or any damages caused thereby, the Borrower shall provide the Lender with a copy of such notice within fifteen (15) days after the Borrower's receipt thereof. Within fifteen (15) days after the Borrower has learned of the enactment or promulgation of any federal, state or local environmental law/or regulation which may result in any material adverse change in the condition, financial or otherwise, of the Borrower, the Borrower shall provide the Lender with notice thereof.
- 7.12 <u>Lender as Depository</u>. Lawson shall continue to utilize Lender as its depository and remittance point for funds received by Lawson in the ordinary course of business to the extent so used as of the date hereof. Lawson shall maintain sufficient balances to cover the costs of account activity.

### 8. <u>NEGATIVE COVENANTS</u>.

Each Borrower jointly and severally covenants and agrees, except to the extent not applicable to it that, so long as any of the Liabilities remain outstanding, and (even if there shall be no Liabilities outstanding) so long as this Agreement remains in effect (unless the Lender shall give its prior written consent thereto):

- 8.1 Encumbrances. Except as set forth on Schedule 8.1 hereto, or contemplated herein, the Borrower will not create, incur, assume or suffer to exist any security interest, mortgage, pledge, negative pledge, lien or other encumbrance of any nature whatsoever on the Negative Pledge Assets, other than: (i) deposits under workmen's compensation, unemployment insurance, social security and other similar laws, or to secure the performance of bids, tenders or contracts (other than for the repayment of borrowed money) or to secure indemnity, performance or other similar bonds for the performance or contracts (other than for the repayment of borrowed money) or to secure statutory obligations or surety or appeal bonds, or to secure indemnity, performance or other similar bonds in the ordinary course of business; (ii) the liens and security interests in favor of Lender; (iii) liens which arise by operation of law, other than Environmental Liens; and (iv) Permitted Liens as set forth upon Schedule 6.5 hereto.
- 8.2 <u>Indebtedness</u>. Except as otherwise provided herein, Borrower shall not create, assume, or become liable in any manner with respect to or permit to exist any obligations or indebtedness in excess of Five Million Dollars (\$5,000,000.00) without prior written notice to Lender.
- 8.3 <u>Consolidations, Mergers or Acquisitions</u>. Borrower shall not recapitalize, consolidate with, merge with, or except for the Acquisition, otherwise acquire all or substantially all of the assets or properties of any other Person without prior written notice to Lender.
- 8.4 Investments or Loans. Borrower shall not make or permit to exist investments or loans in or to any other Person, except (i) investments in short-term direct obligations of the United States Government, (ii) investments in negotiable certificates of deposit or other investment accounts (i.e. money market accounts) issued by Lender, an affiliate of Lender or by any other bank or financial institution satisfactory to Lender, in its reasonable discretion, and payable to the order of Borrower or to bearer, (iii) investments in commercial paper rated Al or Pl, (iv) advances to employees of Borrower in the ordinary course of business to the extent that such advances do not materially adversely affect Borrower's financial condition or operation of business, (vi) intercompany transfers between Lawson and its Subsidiaries and between Subsidiaries in the ordinary course of business to the extent that such intercompany transfers do not materially adversely affect Borrower's financial condition, operation of business or the financial covenants contained in Section 8.13 herein, (vii) the Acquisition and future acquisitions to the extent permitted hereunder, (viii) Lawson's purchase of treasury stock, (ix) loans to current employees or officers of Borrower as permitted pursuant to Section 8.6.
- 8.5 <u>Guarantees.</u> Except for obligations of other Persons not to exceed \$25,000 in the aggregate or as related to a relocation of an employee or independent sales agent of Borrower, Borrower shall not guarantee, endorse or otherwise in any way become or be responsible for obligations of any other Person, whether by agreement to purchase the indebtedness of any other Person or through the purchase of goods, supplies or services, or maintenance of working capital or other balance sheet covenants or conditions, or by way of stock purchase, capital contribution, advance or loan for the purpose of paying or discharging any indebtedness or obligation of such other Person or otherwise, except endorsements of negotiable instruments for collection in the ordinary course of business.

- 8.6 <u>Compensation to Officers and Others.</u> Except for (a) advances and reimbursements for travel and expenses to Borrower's officers, directors or employees in the ordinary course of business, (b) reasonable salaries and bonuses for all salaried personnel, officers, and directors, (c) payments pursuant to existing agreements entered into by Borrower in the ordinary course, (d) existing loans to officers and employees of Borrower, and (e) advances permitted pursuant to Section 8.4, the Borrower shall not make any loans to or pay any bonuses, fees or other amounts to any officers, directors, employees or stockholders of the Borrower.
- 8.7 <u>Issuance of Stock</u>. Any Borrower except Lawson shall not issue or distribute any capital stock or other securities for consideration or otherwise without the prior written consent of Lender.
- Amendment of Articles of Incorporation, By-Laws; Company Name; Places of Business. Borrower shall not amend its Articles of Incorporation, Articles of Organization, or similar organizational documents or By-Laws, operating agreements, or similar governing documents, without prior written notice to Lender, and except that Borrower may amend its Articles of Incorporation, Articles of Organization or similar organizational document to effect a change in its company name or adopt assumed names, provided that Borrower furnishes to Lender such financing statements executed by Borrower which Lender may reasonably request prior to the filing of such amendment and furnishes to Lender a copy of such amendment, certified by all appropriate Secretaries of State or other state officials within ten (10) Business Days of the date such amendment is filed with such Secretary of State. Borrower shall not make any change to the location of its principal place of business or chief executive office unless prior to the effective date of such change in location, Borrower delivers to Lender such financing statements executed by Borrower which Lender may request to reflect such change in location. Borrower shall deliver such other documents and instruments as Lender may request in connection with such change in name or location within ten (10) Business Days of the effectiveness of such change or Lender's request therefor.
- 8.9 <u>Transactions with Subsidiaries and Affiliates.</u> Borrower will not enter into any transaction including, without limitation: (a) the making of any loans to, or the payment of any bonuses, fees or other money to, any Affiliate, and/or (b) the purchase, sale or exchange of property or the rendering of any service to any Subsidiary or Affiliate except for (i) transactions permitted under Section 8.4 or (ii) transactions in the ordinary course of and pursuant to the reasonable requirements of Borrower's business and upon fair and reasonable terms no less favorable to Borrower than Borrower would obtain in a comparable arm's-length transaction with an unaffiliated person or corporation.
  - 8.10 ERISA Violations. Borrower shall not:
- (A) engage, or permit an ERISA Affiliate of Borrower to engage, in any prohibited transaction described in Section 406 of ERISA or Section 4975 of the Internal Revenue Code for which a class exemption is not available or a private exemption has not been previously obtained from the DOL;

- (B) permit to exist any accumulated funding deficiency for any Pension Plan (as defined in subsection 302 of ERISA and Section 412 of the Internal Revenue Code), whether or not waived;
- (C) fail, or permit an ERISA Affiliate of Borrower to fail, to pay timely required contributions or annual installments due with respect to any Plan including without limitation any installments due with respect to any waived funding deficiency to any Pension Plan;
- (D) terminate, or permit an ERISA Affiliate of Borrower to terminate, any Pension Plan which would result in any liability of Borrower or an ERISA Affiliate of Borrower under Title IV of ERISA;
- (E) fail, or permit an ERISA Affiliate of Borrower to fail, to pay to any Pension Plan any required installment under section (m) of Section 412 of the Internal Revenue Code or any other payment required under Section 412 of the Internal Revenue Code on or before the due date for such installment or other payment;
- (F) amend, or permit an ERISA Affiliate of Borrower to amend, a Pension Plan resulting in an increase in current liability for the plan year such that either Borrower or an ERISA Affiliate of Borrower is required to provide security to such Plan under Section 401(a) (29) of the Internal Revenue Code;
  - (G) permit any reportable event (as defined in Section 4043 of ERISA) to occur;
- (H) withdraw, or permit any ERISA Affiliate to withdraw, from any Pension Plan during a plan year for which Borrower or any ERISA Affiliate is a substantial employer with respect to such plan if Borrower or such ERISA Affiliate would incur liability to the PBGC with respect to such plan under Sections 4063 or 4064 of ERISA; or
- (I) withdraw, or permit any ERISA Affiliate to withdraw, from any Multiemployer Plan if a withdrawal liability would result to Borrower or any ERISA Affiliate pursuant to Section 4201 of ERISA.
- 8.11 <u>Fiscal Year</u>. Borrower's Fiscal Year commences January 1 and ends December 31 of each year. Borrower may change its Fiscal Year only with the prior written consent of Lender.
- 8.12 <u>Subsidiaries</u>. Borrower shall not form or acquire any additional Subsidiaries other than those listed on Schedule 6.12 without prior written notice to Lender or to the extent that the formation or acquisitions of additional Subsidiaries materially reduce the value of the Negative Pledge, or Borrower's ability to repay the Loans.

#### 8.13 Financial Covenants.

- (A) Minimum Liquid Assets. Borrower collectively shall not permit the sum of Borrower's Liquid assets at any time to be less than Seventy-Five Million Dollars (\$75,000,000).
  - (B) Minimum Shareholder Equity. Lawson shall not permit its total shareholder equity any time to be less than One Hundred Million Dollars (\$100,000,000).
- (C) Minimum Debt Service Coverage Ratio. The Debt Service Coverage Ratio shall be calculated as of the end of each Fiscal Quarter during the Term beginning with the Fiscal Quarter ending March 31, 2001 with respect to the one (1) year period encompassing such fiscal quarter and the previous three fiscal quarters then elapsed (i.e. for the Fiscal Quarter ending March 31, 2001, the period subject to calculation shall include the Fiscal Quarter ending December 31, 2000, September 30, 2000, and June 30, 2000). The Borrower's Debt Service Coverage Ratio shall not be less than 1.20 to 1.0.
- (D) Maximum Indebtedness. Subject to the terms of this Agreement, the Borrower shall not permit the sum of Borrower's Indebtedness at any time to exceed One Hundred Million Dollars (\$100,000,000).
- 8.14 <u>Inventory Covenants</u>. Except as permitted by subsection 3.6 or 3.7, Borrower shall not sell any of the Inventory on a bill-and-hold, guaranteed sale, sale-or-return, sale on approval or consignment basis or any other basis subject to a repurchase obligation or return right (other than Borrower's customary practice of accepting returns of Inventory within stated time periods as such practice is now in effect as heretofore described to Lender). Except as permitted by subsection 3.6 or 3.7, Borrower shall not hold any inventory of others, whether on a consignment basis or otherwise.
- 8.15 Environmental. Borrower shall not fail to comply in any material respect with all federal, state or local environmental laws and regulations, including, without limitation, environmental, land use, occupational safety or health laws, rules, regulations, requirements or permits in all jurisdictions in which it, is or may at any time be doing business, including without limitation the federal Resource Conservation and Recovery Act, the Federal Comprehensive Environmental Response, Compensation and Liability Act, the Federal Clear Air Act, the Federal Clean Water Act, and the Federal Occupational Safety and Health Act, as the same may be amended from time to time; provided, however that nothing contained in this subsection 8.16 shall prevent the Borrower from contesting, in good faith by appropriate legal proceedings, any such law, regulation, interpretation thereof or application thereof, provided, further, that Borrower shall not fail to comply with the order of any court or other governmental body of applicable jurisdiction relating to such laws unless Borrower shall currently be prosecuting an appeal or proceedings for review and shall have secured a stay of enforcement or execution or other arrangement postponing enforcement or execution pending such appeal or proceedings for review.
- 8.16 <u>Disposal of Property.</u> Borrower shall not sell, lease, transfer or otherwise dispose of any of the Negative Pledge Assets to any Person, except for sales of Inventory to customers in the ordinary course of business or the sale of uncollectible Accounts to collection companies in the ordinary course of business and for fair value.

#### DEFAULT, RIGHTS AND REMEDIES OF THE LENDER.

- 9.1 <u>Defaults</u>. Each of the following Events of Default which occur while any Liabilities remain outstanding and continues uncured for the applicable cure period contained herein shall constitute a Default under this Agreement:
- (A) Failure to pay interest in accordance with the terms of this Agreement or the other Loan Documents upon the date that such payment is due and such default shall continue for five (5) days after written notice to Lawson;
- (B) Failure to pay principal in accordance with the terms of this Agreement or the other Loan Documents upon the date that such payment is due and such default shall continue for five (5) days after written notice to Lawson;
- Default by Borrower in the due observance or performance of any non-monetary covenants, terms, provisions, agreements or conditions hereinbefore or hereinafter contained in this Agreement or any other Loan Document, required to be kept or performed or observed by Borrower, which default continues for thirty (30) days after service of written notice thereof, provided that if in Lender's reasonable judgment such breach cannot reasonably be cured within such 30-day period, the Borrower shall commence such cure and proceed to so cure in a diligent manner and to complete such cure within sixty (60) days after service of written notice thereof, provided further that such grace period shall not apply, and a Default shall be deemed to have occurred promptly upon such breach, if such breach may not, in Lender's reasonable determination, be cured by Borrower during such thirty (30) day grace period;
  - (D) an Event of Default under any of the Loan Documents;
- (E) any warranty or representation now or hereafter made by Borrower is untrue or incorrect in any material respect when made, or any schedule, certificate, statement, report, financial data, notice, or writing furnished at any time by Borrower to Lender is untrue or incorrect in any material respect on the date as of which the facts set forth therein are stated or certified or any of the foregoing omits to state a fact necessary to make the statements therein contained not misleading in any material respect;
- (F) a judgment or order requiring payment in excess of \$500,000.00 (except for judgments constituting liens and which are not a lien on the Negative Pledge Assets, which are being contested by Borrower in good faith) shall be rendered against Borrower and such judgment or order shall remain unsatisfied or undischarged and in effect for ten (10) consecutive days without a stay of enforcement or execution, provided that this subsection 9.1(F) shall not apply to any judgment for which Borrower is fully insured (except for normal deductibles in connection therewith) and with respect to which the insurer has assumed the defense or is not defending under reservation of right and with respect to which Lender reasonably believes the insurer will pay the full amount thereof (except for normal deductibles in connection therewith);
- (G) a notice of lien, levy or assessment is filed or recorded with respect to the Negative Pledge Assets or all or a substantial part of the assets of Borrower by the United States, or any department, agency or instrumentality thereof, or by any state, county, municipality or other governmental agency, or any taxes or debts owing at any time or times hereafter to any one or more of them become a lien upon any part of Borrower's Negative Pledge Assets; and (i) such lien, levy or assessment is not discharged or released or the enforcement thereof is not stayed within thirty (30) days of the notice or attachment thereof, or (ii) if the enforcement thereof is stayed, such stay shall cease to be in effect, provided that this subsection 9.1(G) shall not apply to any liens, levies or assessments which relate to current taxes not yet due and payable;

<ul><li>(H) there shall occur any loss, theft,</li></ul>	substantial damage or destruction of any item or items of Borrower's Negative Pledge Assets owned by Borrower for which
Borrower is not fully insured as required by this Agreement	or the other Loan Documents (a "Loss"), if the amount of such Loss not fully covered by insurance (excluding any reasonable
deductible amount in connection therewith), together with the	he amount of all other Losses not fully covered by insurance (excluding any deductibles in connection therewith) occurring in
the same Fiscal Year, exceeds \$500,000.00;	

- (I) all or any part of Borrower's Negative Pledge Assets are attached, seized, subjected to a writ or distress warrant, or is levied upon, or comes within the possession of any receiver, trustee, custodian or assignee for the benefit of creditors and on or before the thirtieth (30th) day thereafter such assets are not returned to Borrower, and/or such writ, distress warrant or levy is not dismissed, stayed or lifted;
- (J) (1) a proceeding under any bankruptcy, reorganization, arrangement of debt, insolvency, readjustment of debt or receivership law or statute is filed (a) against Borrower and an adjudication or appointment is made or order for relief is entered, or such proceeding remains undismissed for a period in excess of sixty (60) days; or (b) by Borrower; or (2) Borrower (x) makes an assignment for the benefit of creditors; or (y) or Borrower takes any corporate action to authorize any of the foregoing;
- (K) Lawson voluntarily or involuntarily dissolves or is dissolved, terminates or is terminated or any other Borrower voluntarily or involuntarily dissolves or is dissolved, terminates or is terminated without prior written notice to Lender;
  - (L) Borrower, on a consolidated basis, becomes insolvent or fails generally to pay its debts as they become due;
- (M) Borrower is enjoined, restrained, or in any way prevented by the order of any court or any administrative or regulatory agency from conducting all or any material part of its business affairs on a consolidated basis, for a period in excess of twenty-one (21) Business Days;
- (N) a breach by Borrower shall occur under any material agreement, document or instrument (other than an agreement, document or instrument evidencing the lending of money), whether heretofore, now or hereafter existing between Borrower and any other Person, and such breach, continues unwaived for more than thirty (30) days after such breach first occurs, provided that if in Lender's reasonable judgment such breach cannot reasonably be cured within such 30-day period, the Borrower shall commence such cure and proceed to so cure in a diligent manner and to complete such cure within sixty (60) days after service of written notice thereof, provided further that such grace period shall not apply, and a Default shall be deemed to have occurred promptly upon such breach, if such breach may not, in Lender's reasonable determination, be cured by Borrower during such thirty (30) day grace period;

(O) as to more than \$500,000.00 in indebtedness in the aggregate at any time (i) Borrower shall fail to make any payment due (whether by scheduled maturity
required prepayment, acceleration, demand or otherwise) on any other obligation for borrowed money and such failure shall continue after the applicable grace period, if any, specified i
the agreement or instrument relating to such indebtedness; (ii) any other default under any agreement or instrument relating to any such indebtedness, or any other event, shall occur an
shall continue after the applicable grace period, if any, specified in such agreement or instrument if the effect of such default or event is to accelerate, or to permit the acceleration of, th
maturity of such indebtedness; or (iii) any such indebtedness shall be declared to be due and payable or required to be prepaid (other than by a regularly scheduled required prepayment
prior to the stated maturity thereof;

- (P) a material and adverse change shall occur (i) in the present or reasonably foreseeable prospective operations or financial condition of Borrower or in the value of any material portion of the Negative Pledge Assets, or (ii) which materially impairs the ability of Borrower to perform Borrower's obligations under this Agreement and the other Loan Documents, in each case as determined by Lender in its sole Good Faith discretion;
- (Q) the plan administrator of any Pension Plan applies under Section 412(d) of the Internal Revenue Code for a waiver of the minimum funding standards of Section 412(a) of the Internal Revenue Code and Lender in good faith believes that the approval of such waiver could subject Borrower or an ERISA Affiliate of Borrower to liability in excess of Five Hundred Thousand Dollars (\$500,000.00).
- (R) an accumulated funding deficiency (as defined in Section 203 of ERISA and Section 412 of the Code) exists with respect to any Pension Plan as of the last day of any plan year;
- (S) as of the last day of any plan year, the present value of the benefits under any Pension Plan, as determined by such Plan's independent actuaries, exceeds the value as of such date, as determined by such actuaries, of all assets of such Plan by Five Hundred Thousand Dollars (\$500,000.00);
- (T) the aggregate present value of the benefits under all Pension Plans that do not satisfy clause (S) above, as of the end of each Plan's plan year, as determined by such Plans' independent actuaries, exceeds the aggregate value as of such date, as determined by such actuaries of all assets of all such Pension Plans by Five Hundred Thousand Dollars (\$500,000.00);
- (U) a Termination Event occurs which Lender in good faith believes could individually, or together with any other Termination Event subject either Borrower or an ERISA Affiliate of Borrower to liability in excess of Five Hundred Thousand Dollars (\$500,000.00); or

- (V) except as otherwise specifically permitted herein, any sale, conveyance, assignment or other transfer of, or grant of a security interest, pledge or negative pledge in, all or any part of the title to the Negative Pledge Assets other than to Lender;
- (W) a Change in Control without the written consent of Lender provided, however, that a Change in Control shall be permitted if written notice of the intended sale, conveyance, assignment or other transfer of or grant of security interest in one or more shares of the capital stock of Lawson that causes the Change in Control is given to Lender and if such sale, conveyance, assignment or transfer is to (i) a member of the Immediate Family of the assigning Shareholder, or (ii) a trust, partnership or other entity for the benefit of the assigning Shareholder or his Immediate Family, including but not limited to Port Investment LLP, a Delaware Limited Liability Partnership. (In the event of a permitted transfer of capital stock in Lawson hereunder Lender shall be provided written notice thereof at least five (5) Business Days prior to such transfer.)

Upon the occurrence of any of the foregoing Defaults, Lender may, without notice to Borrower (i) terminate Lender's obligation to make advances to Borrower and/or (ii) deem all of the Liabilities immediately due and payable, except that if an Event of Default described in subsection 9.1(I) hereof shall exist or occur, all of the Liabilities shall automatically, without notice of any kind, be immediately due and payable.

- 9.2 <u>Rights and Remedies Generally.</u> In the event of a Default, the Lender shall have, in addition to any other rights and remedies contained in this Agreement or in any of the other Loan Documents, all of the rights and remedies of an unsecured party under the Code or other applicable laws, all of which rights and remedies shall be cumulative, and non-exclusive, to the extent permitted by law.
- 9.3 <u>Waiver of Demand</u>. Demand, presentment, protest and notice of nonpayment are hereby waived by the Borrower. The Borrower also waives the benefit of all valuation, appraisal and exemption laws.

#### 10. MISCELLANEOUS

10.1 <u>Waiver</u>. Lender's failure, at any time or times hereafter, to require strict performance by Borrower of any provision of this Agreement shall not waive, affect or diminish any right of Lender thereafter to demand strict compliance and performance therewith. Any suspension or waiver by Lender of a Default by Borrower under this Agreement or any of the other Loan Documents shall not suspend, waive or affect any other Default by Borrower under this Agreement or any of the other Loan Documents, whether the same is prior or subsequent thereto and whether of the same or of a different kind or character. None of the undertakings, agreements, warranties, covenants and representations of Borrower contained in this Agreement or any of the other Loan Documents and no Default by Borrower under this Agreement or any of the other Loan Documents shall be deemed to have been suspended or waived by Lender unless such suspension or waiver is in writing signed by an officer of Lender, and directed to Borrower specifying such suspension or waiver. All Defaults shall continue until the same are waived by Lender in accordance with the preceding sentence.

- 10.2 Costs and Attorneys' Fees. If at any time or times hereafter Lender employs counsel in connection with protecting Lender's Negative Pledge in the Negative Pledge Assets or in connection with any of Lender's other rights contemplated by or arising out of this Agreement or any of the other Loan Documents, whether (a) to prepare, negotiate or execute (i) any amendment to or modification or extension of this Agreement, any other Loan Documents or any instrument, document or agreement executed by any Person in connection with the transactions contemplated by this Agreement, (ii) any new or supplemental Loan Documents, or any instrument, document or agreement to be executed by any Person in connection with the transactions contemplated by this Agreement, or (iii) any instrument, document or agreement in connection with any sale or attempted sale of any interest herein to any participant provided, however that prior to Lender employing counsel for such purpose, Lender shall give Borrower sixty (60) days advance written notice of such intent and Borrower may issue a Facility Reduction Notice pursuant to Section 2.3, (b) to commence, defend, or intervene in any litigation with respect to the Negative Pledge, the Negative Pledge Assets or the solvency of Borrower, or to file a petition, complaint, answer, motion or other pleadings with respect to the Negative Pledge, the Negative Pledge Assets or the solvency of Borrower, (c) to take any other action in or with respect to any suit or proceeding (bankruptcy or otherwise), (d) to consult with officers of Lender to advise Lender, (e) to the extent Lender has such rights to protect, collect, lease, sell, take possession of, release or liquidate any of the Negative Pledge Assets, or (f) to attempt to enforce or to enforce any interest of Lender in any of the Negative Pledge Assets, or to enforce any rights of Lender, including, without limitation, Lender's rights to collect any of the Liabilities, then in any of such events, all of the reasonable attorneys' fees arising
- 10.3 <u>Expenditures by the Lender</u>. In the event Borrower shall fail to pay taxes, insurance, assessments, costs or expenses which Borrower is, under any of the terms hereof, required to pay, or fails to keep the Negative Pledge Assets free from other security interests, liens or encumbrances, except as permitted herein, Lender may, in its sole discretion, make expenditures for any or all of such purposes, and the amount so expended, together with interest thereon at the rate prescribed in subsection 2.6(C) hereof, shall be part of the Liabilities, payable on demand.
- 10.4 <u>Custody and Preservation of Negative Pledge Assets.</u> Lender shall be deemed to have exercised reasonable care in the custody and preservation of any of the Negative Pledge Assets in its possession if it takes such action for that purpose as Borrower shall request in writing, but failure by Lender to comply with any such request shall not of itself be deemed a failure to exercise reasonable care, and no failure by Lender to preserve or protect any right with respect to such Negative Pledge Assets against prior parties, or to do any act with respect to the preservation of such Negative Pledge Assets not so requested by Borrower shall of itself be deemed a failure to exercise reasonable care in the custody or preservation of such Negative Pledge Assets.

- 10.5 Reliance by the Lender. All covenants, agreements, representations, obligations and warranties made herein by Borrower shall be joint and several, notwithstanding whether the proceeds of the Loan are paid to Borrower and shall, notwithstanding any investigation by Lender, be deemed to be material to and to have been relied upon by Lender.
- 10.6 Parties. Whenever in this Agreement there is reference made to any of the parties hereto, such reference shall be deemed to include, wherever applicable, a reference to the successors and assigns of Borrower and the successors and assigns of Lender, and the provisions of this Agreement shall be binding upon and shall inure to the benefit of said successors and assigns. Notwithstanding anything herein to the contrary, the Borrower may not assign or otherwise transfer its rights or obligations under this Agreement without the prior written consent of Lender. Without in any way limiting Lender's rights, Lender may sell participations in the Liabilities or sell or assign its rights hereunder and under the other Loan Documents, in whole or in part, on such terms as Lender may determine. In connection with any such proposed participations or assignments, Lender may disclose information otherwise required to be kept confidential hereunder provided such disclosure shall not be made unless the party to whom it is disclosed shall have agreed to keep such information confidential as set forth herein.
- 10.7 CHOICE OF LAW. THIS AGREEMENT SHALL BE DEEMED TO BE EXECUTED AND HAS BEEN DELIVERED AND ACCEPTED IN CHICAGO, ILLINOIS BY SIGNING AND DELIVERING IT THERE. ANY DISPUTE BETWEEN THE PARTIES HERETO ARISING OUT OF, CONNECTED WITH, RELATED TO, OR INCIDENTAL TO THE RELATIONSHIP ESTABLISHED BETWEEN THEM IN CONNECTION WITH THIS AGREEMENT, AND WHETHER ARISING IN CONTRACT, TORT, EQUITY, OR OTHERWISE, SHALL BE RESOLVED IN ACCORDANCE WITH THE INTERNAL LAWS AND NOT THE CONFLICTS OF LAW PROVISIONS OF THE STATE OF ILLINOIS.

#### 10.8 CONSENT TO JURISDICTION.

- (A) EXCLUSIVE JURISDICTION. EXCEPT AS PROVIDED IN SUBSECTION 10.8(B) HEREOF, LENDER AND BORROWER AGREE THAT ALL DISPUTES BETWEEN THEM ARISING OUT OF, CONNECTED WITH, RELATED TO OR INCIDENTAL TO THE RELATIONSHIP ESTABLISHED BETWEEN THEM IN CONNECTION WITH THIS AGREEMENT, AND WHETHER ARISING IN CONTRACT, TORT, EQUITY OR OTHERWISE, SHALL BE RESOLVED ONLY BY STATE OR FEDERAL COURTS LOCATED IN COOK COUNTY, ILLINOIS, AND THE BORROWER AND THE LENDER WAIVE ANY OBJECTION BASED ON VENUE OR FORUM NON CONVENIENCE WITH RESPECT TO ANY ACTION INSTITUTED THEREIN, BUT THE LENDER AND THE BORROWERS ACKNOWLEDGE THAT CERTAIN APPEALS FROM THOSE COURTS MAY HAVE TO BE HEARD BY A COURT LOCATED OUTSIDE OF COOK COUNTY, ILLINOIS. THE BORROWER WAIVES IN ALL DISPUTES ANY OBJECTION THAT IT MAY HAVE TO THE LOCATION OF THE COURT CONSIDERING THE DISPUTE.
- (B) OTHER JURISDICTIONS. THE BORROWER AGREES THAT LENDER SHALL HAVE THE RIGHT TO PROCEED AGAINST BORROWER OR THE BORROWER'S PROPERTY ("PROPERTY") IN A COURT IN ANY LOCATION TO ENABLE LENDER TO ENFORCE A JUDGMENT OR OTHER COURT ORDER ENTERED IN FAVOR OF LENDER. BORROWER AGREES THAT IT WILL NOT ASSERT ANY PERMISSIVE COUNTERCLAIM IN ANY PROCEEDING BROUGHT BY LENDER TO REALIZE ON PROPERTY, NEGATIVE PLEDGE ASSETS OR ANY OTHER SECURITY FOR THE LIABILITIES, OR TO ENFORCE A JUDGMENT OR OTHER COURT ORDER IN FAVOR OF LENDER. BORROWER THEREBY WAIVES ANY OBJECTION THAT IT MAY HAVE TO THE LOCATION OF THE COURT IN WHICH LENDER HAS COMMENCED A PROCEEDING DESCRIBED IN THIS SUBSECTION 10.8(B).

10.9 <u>SERVICE OF PROCESS.</u> THE BORROWER HEREBY WAIVES PERSONAL SERVICE OF ANY AND ALL PROCESS UPON IT AND IRREVOCABLY APPOINTS PEARL A. ZAGER, y VEDDER, PRICE, KAUFMAN & KAMMHOLZ, 222 N. LASALLE, CHICAGO, ILLINOIS, 60601, AS BORROWER'S AGENT FOR THE PURPOSE OF ACCEPTING SERVICE OF PROCESS WITHIN THE STATE OF ILLINOIS. LENDER AGREES TO PROMPTLY FORWARD BY REGISTERED MAIL (NO RETURN RECEIPT REQUIRED) A COPY OF ANY PROCESS SO SERVED UPON SAID AGENT TO BORROWER AT ITS ADDRESS SET FORTH IN SUBSECTION 10.17 HEREOF. BORROWER HEREBY CONSENTS TO SERVICE OF PROCESS AS AFORESAID.

BORROWER FURTHER IRREVOCABLY CONSENTS TO THE SERVICE OF PROCESS OUT OF THE COURTS REFERRED TO IN SUBSECTION 10.8 HEREOF IN ANY SUCH ACTION OR PROCEEDING BY MAILING COPIES OF SUCH SERVICE BY REGISTERED MAIL, POSTAGE PREPAID TO BORROWER AT SAID ADDRESS. NOTHING IN THIS AGREEMENT SHALL AFFECT THE RIGHT OF LENDER TO SERVE PROCESS IN ANY OTHER MANNER PERMITTED BY LAW BUT ANY FAILURE TO RECEIVE SUCH COPY SHALL NOT AFFECT IN ANY WAY THE SERVICE OF SUCH PROCESS.

#### 10.10 WAIVER OF JURY TRIAL AND BOND.

- (A) WAIVER OF JURY TRIAL. THE BORROWER AND THE LENDER EACH WAIVE ANY RIGHT TO HAVE A JURY PARTICIPATE IN RESOLVING ANY DISPUTE, WHETHER SOUNDING IN CONTRACT, TORT, OR OTHERWISE, BETWEEN THE LENDER AND THE BORROWER ARISING OUT OF, CONNECTED WITH, RELATED TO OR INCIDENTAL TO THE RELATIONSHIP ESTABLISHED BETWEEN THEM IN CONNECTION WITH THIS AGREEMENT OR ANY OTHER INSTRUMENT, DOCUMENT OR AGREEMENT EXECUTED OR DELIVERED IN CONNECTION THEREWITH OR THE TRANSACTIONS OR IN ANY PROCEEDING REFERRED TO IN THE SECTIONS 10.8, 10.9 AND 10.10.
- (B) WAIVER OF BOND. THE BORROWER WAIVES THE POSTING OF ANY BOND OTHERWISE REQUIRED OF LENDER IN CONNECTION WITH ANY JUDICIAL PROCESS OR PROCEEDING TO OBTAIN POSSESSION OF, REPLEVY, ATTACH OR LEVY UPON NEGATIVE PLEDGE ASSETS OR ANY OTHER SECURITY FOR THE LIABILITIES, TO ENFORCE ANY JUDGMENT OR OTHER COURT ORDER ENTERED IN FAVOR OF LENDER, OR TO ENFORCE BY SPECIFIC PERFORMANCE, TEMPORARY RESTRAINING ORDER, PRELIMINARY OR PERMANENT INJUNCTION, THIS AGREEMENT, OR ANY OTHER AGREEMENT OR DOCUMENT BETWEEN LENDER AND THE BORROWER.

- 10.11 ADVICE OF COUNSEL. THE BORROWER ACKNOWLEDGES AND REPRESENTS TO THE LENDER THAT IT DISCUSSED THIS AGREEMENT WITH ITS LAWYERS
- 10.12 <u>SEVERABILITY</u>. WHEREVER POSSIBLE, EACH PROVISION OF THIS AGREEMENT SHALL BE INTERPRETED IN SUCH MANNER AS TO BE EFFECTIVE AND VALID UNDER APPLICABLE LAW, BUT IF ANY PROVISION OF THIS AGREEMENT SHALL BE PROHIBITED BY OR INVALID UNDER APPLICABLE LAW, SUCH PROVISION SHALL BE INEFFECTIVE ONLY TO THE EXTENT OF SUCH PROVISION OR INVALIDITY, WITHOUT INVALIDATING THE REMAINDER OF SUCH PROVISION OR THE REMAINING PROVISIONS OF THIS AGREEMENT.
- 10.13 <u>Application of Payments</u>. Notwithstanding any contrary provision contained in this Agreement or in any of the other Loan Documents, Borrower irrevocably waives the right to direct the application of any and all payments at any time or times hereafter received by Lender from Borrower or with respect to any of the Negative Pledge Assets, and Borrower does hereby irrevocably agree that Lender shall have the continuing exclusive right to apply and reapply any and all payments received at any time or times hereafter, whether with respect to the Negative Pledge Assets or otherwise, against the Liabilities in such manner as Lender may deem advisable, notwithstanding any entry by Lender upon any of its books and records.
- Marshalling; Payments Set Aside. The Lender shall be under no obligation to marshall any assets in favor of the Borrower or any other party or against or in payment of any or all of the Liabilities. To the extent that the Borrower makes a payment or payments to Lender or Lender enforces its security interests or exercises its rights of set off, and such payment or payments or the proceeds of such enforcement or set off or any part thereof are subsequently invalidated, declared to be fraudulent or preferential set aside and/or required to be repaid to a trustee, receiver or any other party under any bankruptcy law, state or federal law, common law or equitable cause, then, to the extent of such recovery, the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such enforcement or set off had not occurred.
- 10.15 Section Titles. The section titles contained in this Agreement shall be without substantive meaning or content of any kind whatsoever and are not a part of the agreement between the parties.
- 10.16 <u>Continuing Effect</u>. This Agreement, Lender's security interests in the Negative Pledge Assets, and all of the other Loan Documents shall continue in full force and effect so long as any Liabilities shall be owed to Lender, and (even if there shall be no Liabilities outstanding) so long as this Agreement has not been terminated as provided in subsection 2.8 hereof.
- 10.17 Notices. Except as otherwise expressly provided herein, any notice required or desired to be served, given or delivered hereunder shall be in writing, and shall be deemed to have been validly served, given or delivered (i) three (3) days after deposit in the United States mails, with proper postage prepaid, (ii) if sent before 5:00 p.m. CST or CSDT on a Business Day, then on the date when sent after receipt of confirmation or answer back if sent by telecopy, or other similar facsimile transmission and if sent after 5:00 p.m. CST or CSDT on a Business Day or on a non-Business Day, then on the next Business Day, (iii) one (1) Business Day after deposited with a reputable overnight courier with all charges prepaid, or (iv) when delivered, if hand-delivered by messenger, all of which shall be properly addressed to the party to be notified and sent to the address or number indicated as follows:

(i) If to the Lender at: LASALLE BANK NATIONAL ASSOCIATION

135 South LaSalle Street Chicago, Illinois 60602

Attn: Michael Vrchota, Vice President

Telephone: 312/904-1149 Facsimile: 312/904-5483

With a copy to: Barry R. Katz, Esq.

Deutsch, Levy & Engel, Chartered 225 West Washington Street - Ste. 1700

Chicago, IL 60606 Telephone: 312/346-1460 Fax: 312/346-1859

(ii) If to Borrower at: LAWSON PRODUCTS, INC.

1666 East Touhy Avenue Des Plaines, Illinois 60018 Attn: Robert J. Washlow, CEO Telephone: 847/827-9666 Facsimile: 847/795-9030

with a copy of such notice to: VEDDER, PRICE, KAUFMAN & KAMMHOLZ

222 North LaSalle Suite 2600 Chicago, Illinois 60601 Attn: Pearl Zager Telephone: 312/609-7548 Facsimile: 312/609-5005

or to such other address or number as each party designates to the other in the manner herein prescribed.

10.18 <u>Equitable Relief.</u> The Borrower recognizes that, in the event Borrower fails to perform, observe or discharge any obligations or liabilities under this Agreement, any remedy at law may prove to be inadequate relief to Lender; therefore, the Borrower agrees that Lender, if Lender so requests, shall be entitled to temporary and permanent injunctive relief in any such case without the necessity of proving actual damages and the granting of any such relief shall not preclude the Lender from pursuing any other relief or remedies for such breach.

- 10.19 <u>Indemnification</u>. Borrower agrees to defend, protect, indemnify and hold harmless Lender and each of its officers, directors, employees, attorneys, consultants and agents (collectively, the "Indemnitees") from and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, claims, costs, expenses and disbursements of any kind or nature whatsoever (including, without limitation, the reasonable fees and disbursements of counsel for and consultants of such Indemnitees in connection with any investigative, administrative or judicial proceeding, whether or not such Indemnitees shall be designated as parties thereto), which may be imposed on, incurred by, or asserted against such Indemnitees (whether direct, or consequential and whether based on any federal or state laws or other statutory regulations, including, without limitation, securities, environmental and commercial laws and regulations, under common law or at equitable cause or on contract or otherwise) in any manner other Loan Documents, or any act, event or transaction related or attendant thereto, the agreements of Lender contained herein, the making of the Loans, the management of such Loans or the Negative Pledge Assets (including any liability under federal, state or local environmental laws or regulations) or the use or intended use of the proceeds of such (collectively, the "Indemnified Matters"); provided that the Borrower shall not have an obligation to any Indemnitee hereunder with respect to Indemnified Matters caused by or resulting from the fraud, bad faith, willful misconduct or gross negligence of any Indemnitee. To the extent that the undertaking to indemnify, pay and hold harmless set forth in this subsection 10.19 may be unenforceable because it is violative of any law or public policy, the Borrower shall contribute the maximum portion which it is permitted to pay and satisfy under applicable law, to the payment and satisfaction of all Indemnified Matters incurred by the Indemnitees.
- 10.20 Non-Discharge. The Borrower waives any defenses based on suretyship or impairment of Negative Pledge Assets, waives any right of discharge under ss.3-605 of the Code and consents in advance to (i) any extension which the Lender may in its sole discretion hereafter grant in the due date for the payment or performance of any obligation of Borrower hereunder, (ii) any modification for the benefit of Borrower, including a material modification of any obligation of the Borrower hereunder which the Lender may in its sole discretion hereafter grant, (iii) any action taken hereunder which may result in any impairment of the value of any Negative Pledge Assets whether provided by Borrower, including, without limitation, any impairment of collateral as described in subsection 3-605(g) of the Code; (iv) any diligence in collection, the discharge or release of any party hereto or party to any subordination agreement, (v) the discharge or release of any collateral or, all presentment for payment, demand, protest or notice of protest, dishonor or non-payment, or (vi) the granting of any additional credit hereunder without notice.
  - 10.21 Effective Date. This Agreement shall be effective from and after the Closing Date.
- 10.22 <u>Lender's Right to Assign</u>. Lender shall have the right to assign, transfer, sell, negotiate, pledge or otherwise hypothecate this Agreement and any of its rights and security hereunder, including the Note and any other Loan Documents. Borrower hereby agrees that all of the rights and remedies of Lender in connection with the interest so assigned shall be enforceable against Borrower by such assignee with the same force and effect and to the same extent as the same would have been enforceable by Lender but for such assignment.

Lender shall have the right to sell participations in the Loan at any time without the consent of Borrower and at no cost to Borrower.

10.23 <u>Construction</u>. In interpreting or constructing the terms of this Agreement, the Loan Documents and the other Loan Documents, there shall be no presumption of construction or interpretation against the drafter thereof.

 $IN\ WITNESS\ WHEREOF, this\ Agreement\ has\ been\ duly\ executed\ as\ of\ the\ day\ and\ year\ first\ above\ written.$ 

(Signature Page Attached)

IN WITNESS WHEREOF, this Agreement has been duly executed as of the day and year first above written.

# LASALLE BANK NATIONAL ASSOCIATION (as Lender)

By: <u>/s/ Michael J. Vrchota</u>
Its: <u>Vice President</u>

LAWSON PRODUCTS, INC., a Delaware Corporation (as Borrower)

By: <u>/s/ Robert J. Washlow</u> Print Name: <u>Robert J. Washlow</u> Its: <u>Chief Executive Officer</u>

LAWSON PRODUCTS, INC. (Georgia), a Georgia Corporation (as Borrower)

By: <u>/s/ Robert J. Washlow</u> Print Name: <u>Robert J. Washlow</u> Its: <u>Chief Executive Officer</u>

LAWSON PRODUCTS, INC. (New Jersey), a New Jersey Corporation (as Borrower)

By: <u>/s/ Robert J. Washlow</u> Print Name: <u>Robert J. Washlow</u> Its: <u>Chief Executive Officer</u>

LAWSON PRODUCTS, INC. (Nevada), a Nevada Corporation (as Borrower)

By: <u>/s/ Robert J. Washlow</u> Print Name: <u>Robert J. Washlow</u> Its: <u>Chief Executive Officer</u>

## LAWSON PRODUCTS, INC. (Ontario) a Canadian Corporation (as Borrower)

By: /s/ Robert J. Washlow Print Name: Robert J. Washlow

Its: President

LP SERVICE CORP., an Illinois Corporation (as Borrower)

By: <u>/s/ Robert J. Washlow</u> Print Name: <u>Robert J. Washlow</u> Its: <u>Chief Executive Officer</u>

LPI HOLDINGS, INC., an Illinois Corporation (as Borrower)

By: <u>/s/ Robert J. Washlow</u> Print Name: <u>Robert J. Washlow</u> Its: <u>Chief Executive Officer</u>

CRONATRON WELDING SYSTEMS, INC., a North Carolina Corporation (as Borrower)

By: <u>/s/ Robert J. Washlow</u> Print Name: <u>Robert J. Washlow</u> Its: <u>Chief Executive Officer</u>

DRUMMOND AMERICAN Corporation (as Borrower), an Illinois Corporation (as Borrower)

By: /s/ Robert J. Washlow Print Name: Robert J. Washlow Its: Chief Executive Officer LAWSON PRODUCTS LIMITED, a United Kingdom Corporation (as Borrower)

By: <u>/s/ Robert J. Washlow</u> Print Name: <u>Robert J. Washlow</u>

Its: Chairman

ACS/SIMCO, INC., an Illinois Corporation (as Borrower)

By: <u>/s/ Robert J. Washlow</u> Print Name: <u>Robert J. Washlow</u> Its: <u>Chairman</u>

ASSEMBLY COMPONENT SYSTEMS, INC., an Illinois Corporation (as Borrower)

By: <u>/s/ Robert J. Washlow</u> Print Name: <u>Robert J. Washlow</u> Its: <u>Chairman</u>

AUTOMATIC SCREW MACHINE PRODUCTS, COMPANY, INC., an Alabama Corporation (as Borrower)

By: <u>/s/ Robert J. Washlow</u> Print Name: <u>Robert J. Washlow</u> Its: <u>Chairman</u>

C.B. LYNN COMPANY, an Illinois Corporation (as Borrower)

By: <u>/s/ Robert J. Washlow</u> Print Name: <u>Robert J. Washlow</u>

Its: Chairman

LP INDUSTRIAL PRODUCTS COMPANY, an Illinois Corporation (as Borrower)

By: <u>/s/ Robert J. Washlow</u> Print Name: <u>Robert J. Washlow</u>

Its: Chairman

LAWSON PRODUCTS DE MEXICO, S.A. DE C.V. (as Borrower)

By: <u>/s/ Robert J. Washlow</u> Print Name: <u>Robert J. Washlow</u> Its: <u>Chairman</u>

LAWSON PRODUCTS, INC., a Texas Corporation (as Borrower)

By: <u>/s/ Robert J. Washlow</u> Print Name: <u>Robert J. Washlow</u>

Its: Chairman of the Board and Chief Executive Officer

## Exhibit A1

Real Property

## Exhibit A2

Locations of Negative Pledge Assets

## Exhibit B

Form of Signature Authorization

Exhibit C

Form of Note

# Exhibit D

Conditions Precedent to Initial Loan, or Advance

# Exhibit E

Form of Certificate to Accompany Quarterly and Annual Financial Statements

## Exhibit F

Responsible Officers

Deposit Inventory System

Third Party Goods

Permitted Liens

Other Company Names or Fictitious Names

Tax Liabilities

Subsidiaries

Litigation

ERISA Plans

Officers and Directors of Borrower

Liens, Security Interests and Encumbrances

Indebtedness

#### FIRST AMENDMENT TO CREDIT AGREEMENT

This FIRST AMENDMENT TO CREDIT AGREEMENT dated as of August 12, 2002 (the "First Amendment"), is entered into by and among LAWSON PRODUCTS, INC., a Delaware corporation ("Lawson"). whose address is 1666 East Touhy Avenue, Des Plaines, Illinois 60018, LAWSON PRODUCTS, INC. (GEORGIA), a Georgia corporation ("Lawson"). Georgia"), whose address is 1197 Satellite Boulevard, Suwanee, Georgia 30024, LAWSON PRODUCTS, INC. (NEW JERSEY), a New Jersey corporation ("Lawson New Jersey"), whose address is 28 Industrial Road, Fairfield, New Jersey 07004, LAWSON PRODUCTS, INC. (NEVADA), a Nevada corporation ("Lawson Nevada"), whose address is 1381 Capital Boulevard, Reno, Nevada 89502, LAWSON PRODUCTS, INC. (ONTARIO), a Canadian corporation ("Lawson Canada"), whose address is 7315 Rapistan Court, Mississauga, Ontario. Canada L5N5Z4, LP SERVICE CORP., an Illinois corporation ("LP Service"), whose address is c/o Lawson Products, Inc., 1666 East Touhy Avenue, Des Plaines, Illinois 60018, LPI HOLDINGS, INC., an Illinois corporation ("LPI Holdings"), whose address is c/o Lawson Products. Inc., 1666 East Touhy Avenue, Des Plaines, Illinois 60018, CRONATRON WELDING SYSTEMS, INC., a North Carolina corporation ("Cronatron Welding"), whose address is 6510 Northpark Boulevard, Charlotte, North Carolina 28216-2367, DRUMMOND AMERICAN CORPORATION, an Illinois corporation ("Drummand American"), whose address is 600 Corporate Woods Parkway, Vernon Hills, Illinois 60061, LAWSON PRODUCTS LIMITED, a United Kingdom corporation ("Lawson Limited"), whose address is 300 The Quadrant, Ash Ridge Road, Bradley Stoke, Bristol, England BS12 4QU, ACS/SIMCO, INC., an Illinois corporation ("ACS/Simco"), whose address is 15850 West 108th Street, Lenexa, Kansas 66219, ASSEMBLY COMPONENT SYSTEMS, INC., an Illinois corporation ("Assembly Component"), whose address is 240 West 83rd Street, Bun Ridge, Illinois 60521, AUTOMATIC SCREW MACHINE PRODUCTS COMPANY, INC., an Alabama corporation ("Automatic Screw"), whose address is 709 2nd Avenue, SE Decatur, Alabama 35601, C.B. LYNN COMPANY, an Illinois corporation ("C.B. Lynn"), whose address is c/o Lawson Products, Inc., 1666 East Touhy Avenue, Des Plaines, Illinois 60018, LP INDUSTRIAL PRODUCTS COMPANY, an Illinois corporation ("LP Industrial"), whose address is c/c Lawson Products, Inc., 1666 East Touly Avenue, Des Plaines, Illinois 60018, and LAWSON PRODUCTS DE MEXICO, S.A. DE C.V., a Mexican corporation ("Lawson Mexico" Lawson Georgia, Lawson New Jersey, Lawson Nevada, Lawson Canada, LP Service, LPI Holdings, Cronatron Welding, Drummand American, Lawson Limited, ACS/Simco, Assembly Component, Automatic Screw, C.B. Lynn, LP Industrial and Lawson Mexico each being referenced to herein as the "Subsidiary", and collectively referred to herein as the "Subsidiaries" Lawson and the Subsidiaries each being referred to herein as a "Borrower", and collectively referred to herein as the "Borrowers"), whose address is Av. Washington No. 1103, 44190 Guadelajara, Jal., and LASALLE BANK NATIONAL ASSOCIATION, a national banking association (the "Bank"), whose address is 135 South La Salle Street, Chicago, Illinois 60603.

#### RECITALS:

A. The Borrowers and the Bank entered into that certain Credit Agreement dated as of March 27, 2001 (the "Credit Agreement"), pursuant to which Credit Agreement the Bank has made a revolving Loan to the Borrowers evidenced by that certain Promissory Note dated as of March 27, 2001 in the maximum principal amount of Fifty Million and 00/100 Dollars

(\$50,000,000.00), jointly and severally executed by the Borrowers and made payable to the order of the Bank (the "Revolving Note").

B. At the present time the Borrowers request, and the Bank is agreeable to a modification of the Credit Agreement to provide for the issuance of Letters of Credit tinder the Loan, pursuant to the terms and conditions hereinafter set forth.

NOW THEREFORE, in consideration of the premises and other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the Borrowers and the Bank hereby agree as follows:

#### AGREEMENTS:

- 1. <u>RECITALS</u>. The foregoing Recitals ate hereby made a part of this First Amendment,
- 2. <u>DEFINITIONS</u>. Capitalized words and phrases used herein without definition shall have the respective meanings ascribed to such words and phrases in the Credit Agreement.

## 3. <u>AMENDMENTS TO THE CREDIT AGREEMENT</u>.

- 3.1 <u>Letter of Credit Definitions</u>. The following definitions are hereby to Section 1.1 of the Credit Agreement in the appropriate alphabetical order:
- "<u>Letter of Credit</u>" and "<u>Letters of Credit</u>" shall mean, respectively, a letter of credit and all such letters of credit issued by the Bank, in its sole discretion, upon the execution and delivery by the Borrower and the acceptance by the Bank of a Master Letter of Credit Agreement and an application for Letter of Credit, as set forth in Section 2.1(B) of this Agreement."
- "Letter of Credit Obligations" shall mean, at any time, an amount equal to the aggregate of the original face amounts of all Letters of Credit minus the sum of (i) the amount of any reductions in the original face amount of any Letter of Credit which did not result from a draw thereunder, (ii) the amount of any payments made by the Bank with respect to any draws made tinder a Letter of Credit for which the Borrower has reimbursed the Bank, (iii) the amount of any payments made by the Bank with respect to any draws made under a Letter of Credit which have been converted to a Revolving Loan as set forth in Section 2.1(B), and (iv) the portion of any issued but expired Letter of Credit which has not been drawn by the beneficiary thereunder. For purposes of determining the outstanding Letter of Credit Obligations at any time, the Bank's acceptance of a draft drawn on the Bank pursuant to a Letter of Credit shall constitute a draw on the applicable Letter of Credit at the time of such acceptance."
- " "Maximum Letter of Credit Obligation" shall mean Five Hundred Thousand and 00/100 Dollars (\$500,000.00)."

- "Revolving Loan Availability" shall mean at any time, the Maximum Facility less the Letter of Credit Obligations."
- 3.2 <u>Commitment of the Bank</u>. Section 2.1 of the Credit Agreement is hereby amended in its entirety to read as follows:
- 2.1 Loan, Purpose, etc.
- (A) Loan. Provided there does not then exist a Default or an Event of Default, subject to the provisions of Section 4 below, and subject to the other provisions and conditions of this Agreement, the Lender agrees, following the Borrower's execution of this Agreement and all agreements and documents contemplated hereby, to advance Loans to the Borrower on a revolving credit basis (collectively, the "Loan") at tech times as the Borrowers may from time to time request until, hut not including, the Loan Maturity Date, and in such amounts as the Borrowers may from time to time request, provided, however, that the aggregate principal balance of all Loans outstanding at any time shall not exceed the Revolving Loan Availability. Each advance to the Borrower under this Section 2.1 shall he in multiples of Two Hundred Fifty Thousand Dollars (\$250,000.00) and shall, on the day of such advance, he deposited, in immediately available funds, in the Lawson's demand deposit account with the Lender, or in such other account with the Lender as Lawson may, from time to time, designate. The Loan made by the Lender to the Borrower under this Section 2.1 shall be evidenced, in part, by a promissory note of even date herewith in the form attached hereto as Exhibit C (the "Note") with the blanks appropriately filled. The Liabilities evidenced by the Note shall become immediately due and payable, (i) at provided in Section 9.1 hereof; (ii) without notice or demand upon termination of this Agreement pursuant to Section 2.8 hereof.
- (B) Letters of Credit. Subject to the terms and conditions of this Agreement and upon the execution by any of the Borrowers and the Bank of a Master Letter of Credit Agreement and, the upon the execution and delivery by any of the Borrowers, and the acceptance by the Bank, in its sole and absolute discretion, of an application for letter of credit, the Bank agrees to issue for the account of the Borrowers out of the Maximum Facility, such Letters of Credit in the standard form of the Bank and otherwise in form and substance acceptable to the Bank, from time to time during the term of this Agreement, provided that the Letter of Credit Obligations may not at any time exceed the Maximum Letter of Credit Obligation and provided, further, that no Letter of Credit shall have an expiration date later than the Loan Maturity Date. The Letter of Credit Obligations shall also be evidenced by the Note. The amount of any payments made by the Bank with respect to draws wade by a beneficiary under a Letter of Credit for which the Borrowers have failed to reimburse the Bank upon the earlier of(i) the Bank's demand for repayment, or (ii) five (5) days from the date of such payment to such beneficiary by the Bank, shall be deemed to have been converted to a Loan as of the date such payment was made by the Bank to such beneficiary. Upon the occurrence of an Event of a Default and at the option of the Bank, all Letter of Credit Obligations shall be converted to Loans,

all without demand, presentment, protest or notice of any kind, all of which are hereby waived by the Borrowers.

- (C) <u>Purpose</u>. The proceeds of the Loan and the Letters of Credit shall be used by Borrower for working capital requirements and general corporate purposes and for the Acquisition or future acquisitions.
- (D) <u>Limits</u>. The aggregate outstanding principal balance of the Loan and the Letter of Credit Obligations shall not at any time exceed the Maximum Facility. In the event the aggregate outstanding principal balance of all Loans and Letter of Credit Obligations hereunder exceed the Maximum Facility, the Borrower shall, without notice or demand of any kind, immediately make such repayments of the Loans or take such other actions as shall be necessary to eliminate such excess."
  - 3.3 <u>Issuance of Letters of Credit</u>. The following provision is hereby added to the Credit Agreement as Section 4.7:
- " 4.7Each Letter of Credit shall be issued by the Bank upon the execution of the Bank's standard Master Letter of Credit Agreement by any of the Borrowers and the Bank, and the execution and delivery by such Borrower and the acceptance by the Bank, in its sole discretion, of the Bank's standard application for Letter of Credit and the payment by the Borrowers of the Bank's fees charged in connection therewith. In addition to all other applicable fees, charges and/or interest payable by the Borrower pursuant to the Master Letter of Credit Agreement or otherwise payable in accordance with the Bank's standard letter of credit fee schedule, all standby Letters of Credit issued under and pursuant to this Agreement shall bear an annual fee equal to three-quarters of one percent (0.75%) of the face amount of such standby Letter of Credit, payable by the Borrower on or before the issuance of such Letter of Credit by the Bank and quarterly in advance thereafter unless and until (i) such Letter of Credit has expired or has been returned to the Bank, or (ii) the Bank has paid the beneficiary thereunder the hill face amount of such Letter of Credit other than standby Letters of Credit shall bear such fees, costs and interest as charged by the Bank and shall contain such other terms at set forth in the Master Letter of Credit Agreement and the Bank's standard letter of credit fee schedule."
- 4. <u>REPRESENTATIONS AND WARRANTIES</u>. To introduce the Bank to enter into First Amendment, each of the Borrowers hereby certifies, represents and warrants to the this Bank that:
- 4.1 <u>Organization</u>. Each of the Borrower are corporations duly organized, existing and in good standing under the laws of the respective states and/or countries set forth in the preamble of this First Amendment, with full and adequate corporate power to carry on arid conduct its business as presently conducted. Each of the Borrowers is duly licensed or qualified in all foreign jurisdictions wherein the nature of its activities require such qualification tar licensing. The Articles of Incorporation and Bylaws, Borrowing Resolutions and Incumbency Certificate of each of the Borrower have not been changed or amended since the most recent date

that certified copies thereof were delivered to the Bank and are in full force and effect. The exact legal name of the Borrowers is as set forth in the preamble of this First Amendment, and the Borrowers will not change any of their name, organizational identification numbers, if they have one, their type of organization, their jurisdictions of organization or other legal structure.

- 4.2 <u>Authorization</u>. Each of the Borrowers is duly authorized to execute and deliver this First Amendment and is and will continue to be duly authorized to borrow monies under the Credit Agreement, as amended hereby, and to perform its obligations under the Credit Agreement, as amended hereby.
- 4.3 No Conflicts. The execution and delivery of this First Amendment and the performance by the Borrower of its obligations under the Credit Agreement, as amended hereby, do not and will not conflict with any provision of law or of the Articles of Incorporation or Charter, as the case may be, or Bylaws of each of list Borrower or of any agreement binding upon any of the Borrowers,
- 4.4 <u>Validity and Binding Effect</u>. The Credit Agreement, as amended hereby, is a legal, valid and binding obligation of each of the Borrowers, enforceable against the Borrowers in accordance with its terms, except as enforceability maybe limited by bankruptcy, insolvency or other similar laws of general application affecting the enforcement of creditors' rights or by general principles of equity limiting the availability of equitable remedies.
- 4.5 <u>Compliance with Credit Agreement</u>. The representation and warranties set forth in Section 6 of the Credit Agreement, as amended hereby, are true and correct with the same effect as if such representations and warranties had been made on the date hereof, with the exception that all references to the financial statements shall meats the financial statements most recently delivered to the Bank and except for such changes as are specifically permitted under the Credit Agreement. In addition, each of the Borrower has complied with and is in compliance with all of the covenants set forth in the Credit Agreement, as amended hereby, including, but not limited to, those set forth in Section 7 and Section 8 thereof
- 4.6 No Event of Default. As of the date hereof, no Event of Default under Section 9 of the Credit Agreement, as amended hereby, or event or condition which, with the giving of notice or the passage of time, or both, would constitute an Event of Default, has occurred or is continuing.
  - 5. <u>CONDITIONS PRECEDENT</u>. This First Amendment shall become effective as of the date above first written after receipt by the Bank of the following documents:
    - 5.1 <u>First Amendment</u>. This First Amendment executed by the Borrowers and the Bank.
    - 5.2 Other Documents. Such other documents, certificates, resolutions and/or opinions of counsel as the Bank may request.

#### 6. GENERAL.

- 6.1 <u>Governing Law: Severability.</u> This First Amendment shall be construed in accordance with and governed by the laws of Illinois, Wherever possible each provision of the Credit Agreement and this First Amendment shall he interpreted in such manner as to be effective and valid under applicable law, but if any provision of the Credit Agreement and this First Amendment shall be prohibited by or invalid under such law, such provision shall be ineffective to the extent of such prohibition or invalidity, without invalidating the remainder of such provision or the remaining provisions of the Credit Agreement and this First Amendment.
- 6.2 <u>Successors and Assigns</u>. This First Amendment shall be binding upon the Borrowers and the Bank arid their respective successors and assigns, and shall inure to the benefit of the Borrowers and the Bank and the successors and assigns of the Bank.
- 6.3 <u>Continuing Force and Effect of Loan Documents</u>. Except as specifically modified or amended by the terms of this First Amendment, all other terms and provisions of the Credit Agreement and the other Loan Documents are incorporated by reference herein, and in all respects, shall continue its full force and effect. Each of the Borrowers, by execution of this First Amendment, hereby reaffirms, assumes and binds itself to all of the obligations, duties, rights, covenants, terms and conditions that arc contained in the Credit Agreement nod the other Loan Documents.
- 6.4 <u>References to Credit Agreement.</u> Each reference in the Credit Agreement to "this Agreement", "hereunder", "hereof", or words of like import, and each reference to the Credit Agreement in any and all instruments or documents delivered in connection therewith, shall be deemed to refer to the Credit Agreement, as amended hereby.
- 6.5 Expenses. The Borrowers shall pay all costs and expenses in connection with the preparation of this First Amendment and other related loan documents, including, without limitation, reasonable attorneys' fees and time charges of attorneys who may be employees of the Bank or any affiliate or parent of the Bank, The Borrowers shall pay may and all stamp and other taxes, UCC search fees, filing fees and other costs and expenses in connection with the execution and delivery of this First Amendment and the other instruments and documents to tie delivered hereunder, and agrees to save the Bank harmless from and against any and all liabilities with respect to or resulting from any delay in paying or emission to pay such costs and expenses.
  - 6.6 Counterparts. This First Amendment may be executed in any number of counterparts, all of which shall constitute one and the same agreement.

IN WITNESS WHEREOF, the parties hereto have executed this First Amendment to Credit Agreement as of the date first above written,

LASALLE BANK NATIONAL ASSOCIATION, a national banking association

By: <u>/s/ Michael J. Vrchota</u> Name: Michael J. Vrchota Title: Vice President

LAWSON PRODUCTS, INC., a

Delaware corporation

By: <u>/s/ Robert J. Washlow</u>
Name: Robert J. Washlow
Title: Chief Executive Officer

LAWSON PRODUCTS, INC. (GEORGIA), a Georgia corporation

By: <u>/s/ Robert J. Washlow</u>
Name: Robert J. Washlow
Title: Chief Executive Officer

LAWSON PRODUCTS. INC. (NEVADA), a Nevada corporation

By: <u>/s/ Robert J. Washlow</u>
Name: Robert J. Washlow
Title: Chief Executive Officer

LAWSON PRODUCTS, INC. (ONTARIO), a Canadian corporation

By: <u>/s/ Robert J. Washlow</u>
Name: Robert J. Washlow
Title: Chief Executive Officer

# LP SERVICE CORP., and Illinois corporation

By: <u>/s/ Robert J. Washlow</u>
Name: Robert J. Washlow
Title: Chief Executive Officer

LPI HOLDINGS, INC., an Illinois corporation

By: <u>/s/ Robert J. Washlow</u>
Name: Robert J. Washlow
Title: Chief Executive Officer

CRONATRON WELDING SYSTEMS, INC., a North Carolina corporation

By: <u>/s/ Robert J. Washlow</u>
Name: Robert J. Washlow
Title: Chief Executive Officer

DRUMMOND AMERICAN CORPORATION an Illinois corporation

By: <u>/s/ Robert J. Washlow</u>
Name: Robert J. Washlow
Title: Chief Executive Officer

LAWSON PRODUCTS LIMITED, a United Kingdom corporation

By: <u>/s/ Robert J. Washlow</u> Name: Robert J. Washlow Title: Chairman

# ACS/SIMCO, INC., an Illinois corporation

By: <u>/s/ Robert J. Washlow</u> Name: Robert J. Washlow Title: Chairman

# ASSEMBLY COMPONENT SYSTEMS, INC.

an Illinois corporation

By: <u>/s/ Robert J. Washlow</u> Name: Robert J. Washlow Title: Chairman

AUTOMATIC SCREW MACHINE PRODUCTS COMPANY, INC , an Alabama corporation

By: <u>/s/ Robert J. Washlow</u> Name: Robert J. Washlow Title: Chairman

C.B. LYNN COMPANY, an Illinois corporation

By: <u>/s/ Robert J. Washlow</u> Name: Robert J. Washlow Title: Chairman

LP INDUSTRIAL PRODUCTS COMPANY, an Illinois corporation

By: <u>/s/ Robert J. Washlow</u> Name: Robert J. Washlow Title: Chairman

# LAWSON PRODUCTS DE MEXICO, S.A. BE CV., a Mexican corporation

By: <u>/s/ Robert J. Washlow</u> Name: Robert J. Washlow Title: Chairman

#### SECOND MODIFICATION OF LOAN DOCUMENTS

THIS SECOND MODIFICATION OF LOAN DOCUMENTS (this <u>Modification</u>") is made as of the 11<sup>th</sup> day of July, 2003, by and among Lawson Products, Inc., a Delaware Corporation ("<u>Lawson</u>"), with its principal place of business and chief executive office at 1666 E. Touhy Ave., Des Plaines, Illinois, 60018, various Subsidiaries of Lawson listed on <u>Schedule 6.12</u> to the Credit Agreement (Lawson and the Subsidiaries may be referred to herein collectively as the "<u>Borrower</u>") and **LASALLE BANK NATIONAL ASSOCIATION**, a national banking association, its successors and assigns ("<u>Lender</u>").

#### RECITALS:

- A. Lender has heretofore made a loan ("<u>Loan</u>") to Borrower in the principal amount of Fifty Million and no/100 Dollars (\$50,000,000) pursuant to the terms and conditions of a Credit Agreement dated as of March 27, 2001 between Borrower and Lender, (the "<u>Credit Agreement</u>", all terms not otherwise defined herein shall have the meanings set forth in the Credit Agreement, as amended), and as evidenced by a Promissory Note dated March 27, 2001, in the principal amount of the Loan made payable by Borrower to the order of Lender ("<u>Note</u>").
  - B. The Credit Agreement was amended as of August 12, 2002 to, among other things, add a letter of credit subfacility.
  - C. Borrower desires to amend the Loan Documents in order to increase the Letter of Credit subfacility.

#### AGREEMENT:

**NOW, THEREFORE**, in consideration of (i) the facts set forth hereinabove (which are hereby incorporated into and made a part of this Modification with the intent that Lender may rely upon the matters therein recited as representations and warranties of Borrower), (ii) the agreements by Lender to modify the Loan Documents, as provided herein, (iii) the covenants and agreements contained herein, and (iv) for other good and valuable consideration, the receipt, adequacy and sufficiency of which are hereby acknowledged, the parties hereby agree as follows:

- 1. <u>Amendments to Credit Agreement</u>. Subject to and upon the terms and conditions hereof, Borrower and Lender hereby agree that the Credit Agreement shall be amended as follows:
  - 1.1 The definition of "Maximum Letter of Credit Obligation" shall be amended and restated in its entirety to read as follows:
    - "Maximum Letter of Credit Obligation" shall mean One Million Five Hundred and no/100 Dollars (\$1,500,000)."
  - 1.2 Any reference in the Credit Agreement or any other Loan Document to the Maximum Letter of Credit Obligation shall mean the Maximum Letter of Credit Obligation as amended herein.

- 2. **Representations and Warranties of Borrower**. Borrower hereby represents, covenants and warrants to Lender as follows:
  - 2.1 The representations and warranties in the Credit Agreement, and the other Loan Documents are true and correct as of the date hereof.
- 2.2 There is currently no Default under the Note, the Credit Agreement or the other Loan Documents and Borrower does not know of any event or circumstance which would constitute an Event of Default under the Note, the Credit Agreement or the other Loan Documents.
- 2.3 The Loan Documents are in full force and effect and, following the execution and delivery of this Modification, they continue to be the legal, valid and binding obligations of Borrower enforceable in accordance with their respective terms, subject to limitations imposed by general principles of equity.
- 2.4 There has been no material adverse change in the financial condition of Borrower or any other party whose financial statement has been delivered to Lender in connection with the Loan from the date of the most recent financial statement received by Lender.
  - 2.5 As of the date hereof, Borrower has no claims, counterclaims, defenses, or set-offs with respect to the Loan or the Loan Documents as modified herein.
- Borrower is validly existing under the laws of the State of its formation or organization and has the requisite power and authority to execute and deliver this Modification and to perform the Loan Documents as modified herein. The execution and delivery of this Modification and the performance of the Loan Documents as modified herein have been duly authorized by all requisite action by or on behalf of Borrower. This Modification has been duly executed and delivered on behalf of Borrower.
- 3. **Expenses.** As a condition precedent to the agreements contained herein, Borrower shall pay all out-of-pocket costs and expenses incurred by Lender in connection with this Modification, including, without limitation, title charges, recording fees, appraisal fees and attorneys' fees and expenses.

#### 4. Miscellaneous

- 4.1 This Modification shall be governed by and construed in accordance with the laws of the State of Illinois.
- 4.2 This Modification shall not be construed more strictly against Lender than against Borrower merely by virtue of the fact that the same has been prepared by counsel for Lender, it being recognized that Borrower and Lender have contributed substantially and materially to the preparation of this Modification, and Borrower and Lender each acknowledges and waives any claim contesting the existence and the adequacy of the consideration given by the other in entering into this Modification. Each of the parties to this Modification represents that it has been advised by its respective counnsel of the legal and practical effect of this Modification, and recognizes that it is executing and delivering this Modification, intending thereby to be legally bound by the terms and provisions thereof, of its own free will, without promises or threats or the exertion of duress up on it. The signatories hereto state that they have read and understand this Modification, that they intend to be legally bound by it and that they expressly warrant and represent that they are duly authorized and empowered to execute it.

- 4.3 Notwithstanding the execution of this Modification by Lender, the same shall not be deemed to constitute Lender a venturer or partner of or in any way associated with Borrower nor shall privity of contract be presumed to have been established with any third party.
- 4.4 Borrower and Lender each acknowledges that there are no other understandings, agreements or representations, either oral or written, express or implied, that are not embodied in the Loan Documents and this Modification, which collectively represent a complete integration of all prior and contemporaneous agreements and understandings of Borrower and Lender; and that all such prior understandings, agreements and representations are hereby modified as set forth in this Modification. Except as expressly modified hereby, the terms of the Loan Documents are and remain unmodified and in full force and effect,
  - 4.5 This Modification shall bind and inure to the benefit of the parties hereto and their respective heirs, executors, administrators, successors and assigns.
- 4.6 Any references to the "Note", the Credit Agreement or the "Loan Documents" contained in any of the Loan Documents shall be deemed to refer to the Note, the Mortgage and the other Loan Documents as amended hereby. The paragraph and section headings used herein are for convenience only and shall not limit the substantive provisions hereof. All words herein which are expressed in the neuter gender shall be deemed to include the masculine, feminine and neuter genders. Any word herein which is expressed in the singular or plural shall be deemed, whenever appropriate in the context, to include the plural and the singular.
  - 4.7 This Modification may be executed in one or more counterparts, all of which, when taken together, shall constitute one original Agreement.
  - 4.8 Time is of the essence of each of Borrower's obligations under this Modification.

(Signature Page Follows)

IN WITNESS WHEREOF, the parties hereto have executed this Modification dated as of the day and year first above written.

LASALLE BANK NATIONAL ASSOCIATION (as Lender)

By:/s/ Michael Vrchota

Michael Vrchota ts: Vice President

LAWSON PRODUCTS, INC., a Delaware

Corporation (as Borrower)

By:/s/ Robert J. Washlow

Its:

Robert J. Washlow Chief Executive Officer

LAWSON PRODUCTS, INC., a Georgia

Corporation (as Borrower)

By:/s/ Robert J. Washlow Robert J. Washlow

TODELL J. Washiow

Its: Chief Executive Officer

LAWSON PRODUCTS, INC., a New Jersey Corporation (as Borrower)

By:/s/ Robert J. Washlow

Robert J. Washlow Its: Chief Executive Officer

LAWSON PRODUCTS, INC., a Nevada

Corporation (as Borrower)

By:/s/ Robert J. Washlow

Robert J. Washlow Chief Executive Officer

LAWSON PRODUCTS, INC., a Canadian

Corporation (as Borrower)

By:/s/ Robert J. Washlow

Robert J. Washlow

Its: President

Its:

LP SERVICES CORP., an Illinois Corporation (as Borrower)

By:/s/ Robert J. Washlow

Robert J. Washlow Chief Executive Officer

LPI HOLDINGS, INC., an Illinois Corporation (as Borrower)

By:/s/ Robert J. Washlow

Robert J. Washlow Chief Executive Officer

CRONATRON WELDING SYSTEMS, INC. a North Carolina Corporation (as Borrower)

By:/s/ Robert J. Washlow

Its:

Robert J. Washlow Chief Executive Officer

DRUMMOND AMERICAN Corporation

(as Borrower), an Illinois Corporation (as Borrower)

By:/s/ Robert J. Washlow

Robert J. Washlow Chief Executive Officer

LAWSON PRODUCTS LIMITED, a United Kingdom Corporation (as Borrower)

By:/s/ Robert J. Washlow

Robert J. Washlow

Its: Chairman

ACS/SIMCO, INC., an Illinois Corporation (as Borrower)

By:/s/ Robert J. Washlow

Robert J. Washlow

Its: Chairman

LAWSON PRODUCTS, INC., a Texas Corporation (as Borrower)

By:/s/ Robert J. Washlow

Its:

Robert J. Washlow Chief Executive Officer

ASSEMBLY COMPONENT SYSTEMS, INC., an Illinois Corporation (as Borrower)

By:/s/ Robert J. Washlow

Robert J. Washlow Chairman

AUTOMATIC SCREW MACHINE PRODUCTS COMPANY, INC., an Alabama

Corporation (as Borrower)

By:/s/ Robert J. Washlow

Robert J. Washlow

Its: Chairman

C.B. LYNN COMPANY, an Illinois Corporation (as Borrower)

By:/s/ Robert J. Washlow

Robert J. Washlow

Chairman

Its:

LP INDUSTRIAL PRODUCTS COMPANY, an Illinois Corporation (as Borrower)

By:/s/ Robert J. Washlow

Robert J. Washlow

Its: Chairman

LAWSON PRODUCTS DE MEXICO, S.A. DE C.V. (as Borrower)

By:/s/ Robert J. Washlow

Robert J. Washlow

Its: Chairman

## THIRD MODIFICATION OF LOAN DOCUMENTS

THIS THIRD MODIFICATION OF LOAN DOCUMENTS (this "Modification") is made as of the 15th day of June, 2005, by and among Lawson Products, Inc., a Delaware Corporation ("Lawson"), with its principal place of business and chief executive office at 1666 E. Touhy Ave., Des Plaines, Illinois, 60018, various Subsidiaries of Lawson listed on Schedule 6.12 to the Credit Agreement (Lawson and the Subsidiaries may be referred to herein collectively as the "Borrower") and LASALLE BANK NATIONAL ASSOCIATION, a national banking association, its successors and assigns ("Lender").

## RECITALS:

- A. Lender has heretofore made a loan ("Loan") to Borrower in the principal amount of Fifty Million and no/100 Dollars (\$50,000,000) pursuant to the terms and conditions of a Credit Agreement dated as of March 27, 2001 between Borrower and Lender, (the "<u>Credit Agreement</u>", all terms not otherwise defined herein shall have the meanings set forth in the Credit Agreement, as amended), and as evidenced by a Promissory Note dated March 27, 2001, in the principal amount of the Loan made payable by Borrower to the order of Lender ("<u>Note</u>").
- B. The Credit Agreement was amended as of August 12, 2002 to, among other things, add a letter of credit subfacility. The Credit agreement was further amended as of July 11, 2003 to, among other things, increase the availability under the letter of credit subfacility.
- C. Borrower desires to amend the Loan Documents and increase the available principal amount of the Loan, consolidate certain subfacilities under the Credit Agreement, and for other reasons hereinafter set forth.

## AGREEMENTS:

**NOW, THEREFORE**, in consideration of (i) the facts set forth hereinabove (which are hereby incorporated into and made a part of this Modification with the intent that Lender may rely upon the matters therein recited as representations and warranties of Borrower), (ii) the agreements by Lender to modify the Loan Documents, as provided herein, (iii) the covenants and agreements contained herein, and (iv) for other good and valuable consideration, the receipt, adequacy and sufficiency of which are hereby acknowledged, the parties hereby agree as follows:

## 1. Amendments to the Note.

(a) <u>Maturity Date</u>. The Maturity Date of the Note is extended to March 27, 2009. Any reference in the Note, the Credit Agreement or any other Loan Document to the Maturity Date shall mean March 27, 2009.

(b) <u>Principal</u>. Section 1 of the Note is hereby amended by (i) deleting the numerical reference "FIFTY MILLION DOLLARS (\$50,000,000.00)" and (ii) inserting therefor "SEVENTY-FIVE MILLION DOLLARS (\$75,000,000)" in its place.

## 2. <u>Amendments to the Credit Agreement.</u>

- (a) Section 1.1 of the Credit Agreement is hereby amended by deleting the definition of Debt Service Coverage Ratio.
- (b) Section 1.1 of the Credit Agreement is hereby amended by amending and restating the definitions of "Maximum Facility," "Maximum Letter of Credit Obligation," and "Revolving Loan Availability"

"<u>Maximum Facility</u>" shall mean the maximum amount which the Lender has agreed to consider as a ceiling on the outstanding principal balance of loans and other extensions of credit to be made to or for the account of the Borrower under this Agreement. The Maximum Facility shall be Seventy-Five Million and no/ 100 Dollars (\$75,000,000) United States currency, unless permanently reduced at Lawson's election pursuant to <u>Section 2.3</u>.

"Maximum Letter of Credit Obligation" shall mean Two Million and no/100 Dollars (\$2,000,000)

"Revolving Loan Availability" shall mean at any time, the Maximum Facility less the aggregate of (a) the Letter of Credit Obligations, (b) Maximum Corporate Commercial Card Obligation, and (c) the Maximum Hedging Obligation.

(c) Section 1.1 of the Credit agreement is hereby amended by adding the following new definitions:

"Bank Products" shall mean any service or facility extended to the Borrower or any Subsidiary by the Lender or any Affiliate of the Lender, including: (a) credit cards, (b) credit card processing services, (c) debit cards, (d) purchase cards, (e) ACH Transactions, (f) cash management, including controlled disbursement, accounts or services, or (g) Hedging Agreements.

"Bank Product Agreements" shall mean those certain agreements entered into from time to time by the Borrower with the Lender or any Affiliate of the Lender concerning Bank Products.

"Bank Product Obligations" shall mean all obligations, liabilities, contingent reimbursement obligations, fees, and expenses owing by the Borrower to the Lender or any Affiliate of the Lender pursuant to or evidenced by the Bank Product Agreements and irrespective of whether for the payment of money, whether direct or indirect, absolute or contingent, due or to become due, now existing or hereafter arising.

"Commercial Card Obligations" shall mean Bank Product Obligations owing by the Borrower to Lender or any Affiliate of the Lender for purchase cards pursuant to the Master Commercial Card Agreement (as hereinafter defined).

"Contingent Liability" and "Contingent Liabilities" shall mean, respectively, each obligation and liability of the Borrower and all such obligations and liabilities of the Borrower incurred pursuant to any agreement, undertaking or arrangement by which the Borrower: (a) guarantees, endorses or otherwise becomes or is contingently liable upon (by direct or indirect agreement, contingent or otherwise, to provide funds for payment, to supply funds to, or otherwise to invest in, a debtor, or otherwise to assure a creditor against loss) the indebtedness, dividend, obligation or other liability of any other Person in any manner (other than by endorsement of instruments in the course of collection), including without limitation, any indebtedness, dividend or other obligation which may be issued or incurred at some future time; (b) guarantees the payment of dividends or other distributions upon the shares or ownership interest of any other Person; (c) undertakes or agrees (whether contingently or otherwise): (i) to purchase, repurchase, or otherwise acquire any indebtedness, obligation or liability of any other Person or any property or assets constituting security therefor, (ii) to advance or provide funds for the payment or discharge of any indebtedness, obligation or liability of any other Person (whether in the form of loans, advances, stock purchases, capital contributions or otherwise), or to maintain solvency, assets, level of income, working capital or other financial condition of any other Person, or (iii) to make payment to any other Person other than for value received; (d) agrees to lease property or to purchase securities, property or services from such other Person with the purpose or intent of assuring the owner of such indebtedness or obligation of the ability of such other Person to make payment of the indebtedness or obligation; (e) to induce the issuance of, or in connection with the issuance of, any letter of credit for the benefit of such other Person; or (f) undertakes or agrees otherwise to assur

"Debt" shall mean, as to any Person, without duplication, (a) all indebtedness of such Person; (b) all borrowed money of such Person (including principal, interest, fees and charges), whether or not evidenced by bonds, debentures, notes or similar instruments; (c) all obligations to pay the deferred purchase price of property or services; (d) all obligations, contingent or otherwise, with respect to the maximum face amount of all letters of credit (whether or not drawn), bankers' acceptances and similar obligations issued for the account of such Person (including the Letters of Credit), and all unpaid drawings in respect of such letters of credit, bankers' acceptances and similar obligations; (e) all indebtedness secured by any Lien on any property owned by such Person, whether or not such indebtedness has been assumed by such Person (provided, however, if such Person has not assumed or otherwise become liable in respect of such indebtedness, such indebtedness shall be deemed to be in an amount equal to the fair market value of the property subject to such Lien at the time of determination); (f) the aggregate amount of all Capitalized Lease Obligations of such Person; (g) all Contingent Liabilities of such Person, whether or not reflected on its balance sheet; (h) all Hedging Obligations of such Person; (i) all Debt of any partnership of which such Person is a general partner; and (j) all monetary obligations of such Person under (i) a so-called synthetic, off-balance sheet or tax retention lease, or (ii) an agreement for the use or possession of property creating obligations that do not appear on the balance sheet of such Person but which, upon the insolvency or bankruptcy of such Person, would be characterized as the indebtedness of such Person (without regard to accounting treatment). Notwithstanding the foregoing, Debt shall not include trade payables and accrued expenses incurred by such Person in accordance with customary practices and in the ordinary course of business of such Person.

"Funded Debt" shall mean, as to any Person, all Debt of such Person that matures more than one year from the date of its creation (or is renewable or extendible, at the option of such Person, to a date more than one year from such date).

"Hedging Agreement" shall mean any interest rate, currency or commodity swap agreement, cap agreement or collar agreement, and any other agreement or arrangement designed to protect a Person against fluctuations in interest rates, currency exchange rates or commodity prices.

"Hedging Obligation" shall mean, with respect to any Person, any liability of such Person under any Hedging Agreement.

"Interest Charges" shall mean, for any period, the sum of: (a) all interest, charges and related expenses payable with respect to that fiscal period to a lender in connection with borrowed money or the deferred purchase price of assets that are treated as interest in accordance with GAAP, plus (b) the portion of Capitalized Lease obligations with respect to that fiscal period that should be treated as interest in accordance with GAAP, plus (c) all charges paid or payable (without duplication) during that period with respect to any Hedging Agreements.

"<u>Maximum Commercial Card Obligation</u>" shall mean Bank Product Obligations incurred with respect to purchase cards pursuant to Master Commercial Card Agreement (as hereinafter defined), not to exceed at any, time Five Hundred Twenty-Five thousand and no/100 Dollars (\$525,000).

"Maximum Hedging Obligation" shall mean any liability of a Person under any Hedging Agreement not to exceed at any time One Million and no/100 Dollars (\$1,000,000).

"Sarbanes-Oxley" means the Sarbanes-Oxley Act of 2002.

- "Securities Laws" means the Securities Act of 1933, the Securities Exchange Act of 1934, Sarbanes-Oxley and the applicable accounting and auditing principles, rules, standards and practices promulgated, approved or incorporated by the Securities Exchange Commission or the Public Company Accounting Oversight Board, as each of the foregoing may be amended and in effect on any applicable date hereunder.
- "Total Debt" shall mean all Debt of the Borrower, determined on a consolidated basis, excluding (i) Contingent Liabilities (except to the extent constituting Contingent Liabilities in respect of the Debt of a Person other than the Borrower or any Subsidiaries), (ii) Hedging Obligations, and (iii) Debt of the Borrower to Subsidiaries and Debt of Subsidiaries to the Borrower or to other Subsidiaries.
- (d) Section 2.1 of the Credit Agreement is hereby amended (i) by renumbering subparts (C) and (D) as subparts (E) and (F) respectively, and (ii) adding the following replacement subparts (C) and (D) as follows:
  - "(C) <u>Corporate Commercial Card Facility.</u> Subject to the terms and conditions of this Agreement and upon the execution by the Borrower of a Commercial Card Agreement in form and substance acceptable to the Lender (together with all amendments, modifications and restatements thereof, the "<u>Master Commercial Card Agreement</u>"), the Lender agrees to issue for the account of the Borrower out of the Maximum Facility, such purchase cards in the standard form of the Lender and otherwise in form and substance acceptable to the Lender, from time to time during the term of this Agreement, provided that the Bank Product Obligations represented by purchase cards may not at any time exceed the Maximum Commercial Card Obligation. To the extent the provisions of the Master Commercial Card Agreement differ from, or are inconsistent with, the terms of this Agreement, the provisions of this Agreement shall govern.
  - (D) <u>Interest Rate Protection</u>. The Borrower may, but is not obligated to enter into one or more Hedging Agreements on an ISDA standard form with a qualified counter party to hedge the interest rate with respect to not more than Eleven Million and no/100 Dollars (\$11,000,000) of the principal amount of the Loan, in form and substance reasonably satisfactory to the Lender. Such Hedging Agreement obligation shall be issued for the account of the Borrower out of the Maximum Facility. The Hedging Agreement Obligations may not at any time exceed the Maximum Hedging Agreement Obligation."

- (e) Section 2.1 of the Credit Agreement is hereby amended by amending and restating new subpart (F) (former subpart (D)) to read as follows:
- "(F) <u>Limits</u>. The aggregate outstanding principal balance of the Loan, Letter of Credit Obligations, Commercial Card Obligations and Hedging Agreement Obligations shall not at any time exceed the Maximum Facility. In the event the aggregate outstanding principal balance of all Loans, Letter of Credit Obligations, Commercial Card Obligations and Hedging Agreement Obligations hereunder exceed the Maximum Facility, the Borrower shall, without notice or demand of any kind, immediately make such repayments of the Loan or take such other actions as shall be necessary to eliminate such excess."
- (f) Section 2.6(H) of the Credit Agreement is hereby amended and restated in its entirety to read as follows:
- "(H) <u>Unused Commitment Fee</u>. The Borrower shall pay to Lender a fee (the "<u>Unused Commitment Fee</u>") on the last day of each Fiscal Quarter, equal to one eighth of one percent (1/8%) per annum of the face amount of the average daily unused amount of the Maximum Facility in excess of Seven Million Five Hundred Thousand and no/100 Dollars (\$7,500,000) during such Fiscal Quarter. The Unused Commitment Fee shall be computed on the basis of a 360-day year for the actual number of days elapsed."
- (g) Section 2.8(A) of the Credit Agreement is hereby amended and restated in its entirety to read as follows:
- "(A) Loan Maturity. The Loan, including the full outstanding principal balance thereon and all accrued and then unpaid interest thereon, if not sooner paid, shall be immediately due and payable without notice or demand on March 27, 2009 (the "Loan Maturity Date")."
- (h) Section 6.16 of the Credit Agreement is hereby amended by adding the following subpart (C):
- "(C) Borrower is not an "investment company," or an "affiliated person" of, or "promoter" or "principal underwriter" for, an "investment company," as such terms are defined in the Investment Company Act of 1940, as amended. Neither the making of the Loan, nor the issuance of any Letters of Credit, nor the application of the proceeds or repayment thereof by the Borrower, will violate any provision of such Act or any rule, regulation or order of the Securities and Exchange Commission thereunder or any takeover, disclosure or other federal, state or foreign securities law or Regulations U or X of the Federal Reserve Board. The Borrower is not subject to regulation under any federal, state or foreign statute or regulation which limits its ability to incur Debt."

- (i) Section 7.1(B) of the Credit Agreement is hereby amended by (i) deleting the reference to "one hundred and ten (110) days;" and (ii) inserting "one hundred twenty (120) days" in its place.
  - (j) Section 7.1 of the Credit Agreement is hereby amended by adding the following new subpart (I) as follows:
  - (I) <u>Covenant Compliance Certificate</u>. The Borrower shall, contemporaneously with the furnishing of the financial statements pursuant to this Section 7.1, but in no event later than forty-five (45) days after the end of any Fiscal Quarter, deliver to the Bank a duly completed compliance certificate in form and substance acceptable to Lender, dated as of the date of such financial statements and certified as true and correct by an Authorized Officer, containing a computation of each of the financial covenants set forth in Section 7.1 and stating that the Borrower has not become aware of any Event of Default or Default that has occurred and is continuing or, if there is any such Event of Default or Default describing it and the steps, if any, being taken to cure it."

For purposes hereof and hereafter, the compliance certificate required by 7.1(I) shall be in the form set forth as Exhibit "A" attached hereto and made a part hereof.

- (k) Section 8.13(C) of the Credit Agreement is hereby amended and restated in its entirety to read as follows:
- "(C) Fixed Charge Coverage. As of the end of each of its Fiscal Quarters calculated on a rolling four (4) Fiscal Quarter basis (i.e. the one (1) year period encompassing such Fiscal Quarter and the previous three (3) Fiscal Quarters then elapsed), the Borrower shall maintain a ratio of (a) the total for such Fiscal Quarter of EBITDA minus all Capital Expenditures which are not financed with Funded Debt, to (b) the sum for such Fiscal Quarter of (i) Interest Charges, (ii) required payments of principal of the Liabilities (iii) dividends, and (iv) the sum of all income taxes paid in cash by the Borrower, of not less than 1.10 to 1.00;"
- (l) Section 8.13(E) is hereby amended and restated in its entirety to read as follows:
- "(E) <u>Maximum Indebtedness</u>. Subject to the terms of this Agreement, the Borrower shall not permit the sum of Borrower's Indebtedness at any time to exceed One Hundred Fifty Million and no/100 Dollars (\$150,000,000)."

- (m) Section 8.13 of the Credit Agreement is hereby amended by adding the following additional subparts (F) and (G) as follows:
- "(F) <u>Minimum Liquidity Value</u>. Borrower shall not permit the sum of Borrower's unencumbered cash, Accounts, and Inventory to be less than Seventy-Five Million and no/100 Dollars (\$75,000,000).
- (G) Total Debt to EBITDA. As of the end of each of its fiscal quarters, the Borrower shall maintain a ratio of consolidated Total Debt to consolidated EBITDA for such fiscal quarter, of not greater than 3.00 to 1.00."
- (n) Exhibit "A1" to the Credit Agreement is hereby amended and restated in its entirety to read as set forth on Exhibit "A1" attached hereto and made a part hereof.
- (o) Exhibit "A2" to the Credit Agreement is hereby amended and restated in its entirety to read as set forth on Exhibit "A2" attached hereto and made a part hereof.
- (p) Exhibit "E" to the Credit Agreement is hereby amended and restated in its entirety to read as set forth on Exhibit "E" attached hereto and made a part hereof.
- (q) Schedule 6.8 to the Credit Agreement is hereby amended and restated in its entirety to read as set forth on Schedule 6.8 attached hereto and made a part hereof.
- (r) Schedule 6.12 to the Credit Agreement is hereby amended and restated in its entirety to read as set forth on Schedule 6.12 attached hereto and made a part hereof.
- (s) Schedule 6.21 to the Credit Agreement is hereby amended and restated in its entirety to read as set forth on Schedule 6.21 attached hereto and made a part hereof.
- 3. Representations and Warranties of Borrower. Borrower hereby represents, covenants and warrants to Lender as follows:
  - (a) The representations and warranties in the Credit Agreement, and the other Loan Documents are true and correct as of the date hereof.
- (b) There is currently no Default under the Note, the Credit Agreement or the other Loan Documents and Borrower does not know of any event or circumstance which would constitute an Event of Default under the Note, the Credit Agreement or the other Loan Documents.
- (c) The Loan Documents are in full force and effect and, following the execution and delivery of this Modification, they continue to be the legal, valid and binding obligations of Borrower enforceable in accordance with their respective terms, subject to limitations imposed by general principles of equity.

- (d) There has been no material adverse change in the financial condition of Borrower or any other party whose financial statement has been delivered to Lender in connection with the Loan from the date of the most recent financial statement received by Lender.
  - (e) As of the date hereof, Borrower has no claims, counterclaims, defenses, or set-offs with respect to the Loan or the Loan Documents as modified herein.
- (f) Borrower is validly existing under the laws of the State of its formation or organization and has the requisite power and authority to execute and deliver this Modification and to perform the Loan Documents as modified herein. The execution and delivery of this Modification and the performance of the Loan Documents as modified herein have been duly authorized by all requisite action by or on behalf of Borrower. This Modification has been duly executed and delivered on behalf of Borrower.
- 4. **Expenses**. As a condition precedent to the agreements contained herein, Borrower shall pay all out-of-pocket costs and expenses incurred by Lender in connection with this Modification, including, without limitation, title charges, recording fees, appraisal fees and attorneys' fees and expenses.

#### Miscellaneous.

- (a) This Modification shall be governed by and construed in accordance with the laws of the State of Illinois.
- (b) This Modification shall not be construed more strictly against Lender than against Borrower merely by virtue of the fact that the same has been prepared by counsel for Lender, it being recognized that Borrower and Lender have contributed substantially and materially to the preparation of this Modification, and Borrower and Lender each acknowledges and waives any claim contesting the existence and the adequacy of the consideration given by the other in entering into this Modification. Each of the parties to this Modification represents that it has been advised by its respective counsel of the legal and practical effect of this Modification, and recognizes that it is executing and delivering this Modification, intending thereby to be legally bound by the terms and provisions thereof, of its own free will, without promises or threats or the exertion of duress upon it. The signatories hereto state that they have read and understand this Modification, that they intend to be legally bound by it and that they expressly warrant and represent that they are duly authorized and empowered to execute it.
- (c) Notwithstanding the execution of this Modification by Lender, the same shall not be deemed to constitute Lender a venturer or partner of or in any way associated with Borrower nor shall privity of contract be presumed to have been established with any third party.

- (d) Borrower and Lender each acknowledges that there are no other understandings, agreements or representations, either oral or written, express or implied, that are not embodied in the Loan Documents and this Modification, which collectively represent a complete integration of all prior and contemporaneous agreements and understandings of Borrower and Lender; and that all such prior understandings, agreements and representations are hereby modified as set forth in this Modification. Except as expressly modified hereby, the terms of the Loan Documents are and remain unmodified and in full force and effect.
  - (e) This Modification shall bind and inure to the benefit of the parties hereto and their respective heirs, executors, administrators, successors and assigns.
- (f) Any references to the "Note", the Credit Agreement or the "Loan Documents" contained in any of the Loan Documents shall be deemed to refer to the Note, the Mortgage and the other Loan Documents as amended hereby. The paragraph and section headings used herein are for convenience only and shall not limit the substantive provisions hereof. All words herein which are expressed in the neuter gender shall be deemed to include the masculine, feminine and neuter genders. Any word herein which is expressed in the singular or plural shall be deemed, whenever appropriate in the context, to include the plural and the singular.
  - (g) This Modification may be executed in one or more counterparts, all of which, when taken together, shall constitute one original Agreement.
  - (h) Time is of the essence of each of Borrower's obligations under this Modification.
- (i) <u>Customer Identification USA Patriot Act Notice</u>. The Lender hereby notifies the Borrower that pursuant to the requirements of the USA Patriot Act (Title III of Pub. L. 107-56, signed into law October 26, 2001) (the "Act"), and the Lender's policies and practices, the Lender is required to obtain, verify and record certain information and documentation that identifies the Borrower, which information includes the name and address of the Borrower and such other information that will allow the Lender to identify the Borrower in accordance with the Act.

(Signature Page Follows)

IN WITNESS WHEREOF, the parties hereto have executed this Modification dated as of the day and year first above written.

LASALLE BANK NATIONAL ASSOCIATION (as Lender)

LAWSON PRODUCTS, INC., a Texas corporation

By: /s/ Michael Vrchota Michael Vrchota

First Vice President

LAWSON PRODUCTS, INC., a Delaware Corporation (as Borrower)

By: /s/ Robert J. Washlow

Robert J. Washlow Chief Executive Officer

LAWSON PRODUCTS, INC., a Georgia Corporation (as Borrower)

By: /s/ Robert J. Washlow

Its:

Robert J. Washlow Chief Executive Officer

LAWSON PRODUCTS, INC., a New Jersey Corporation (as Borrower)

By:/s/ Robert J. Washlow

Robert J. Washlow Its: Chief Executive Officer

LAWSON PRODUCTS, INC., a Nevada Corporation (as Borrower)

By: /s/ Robert J. Washlow

Robert J. Washlow Its: Chief Executive Officer By: /s/ Robert J. Washlow

Robert J. Washlow s: Chief Executive Officer

LAWSON PRODUCTS, INC., a Canadian Corporation (as Borrower)

By: /s/ Robert J. Washlow

Robert J. Washlow

President

Its:

Its:

Its:

LP SERVICE CO., an Illinois Corporation (as Borrower)

By: /s/ Robert J. Washlow

Robert J. Washlow Chief Executive Officer

LPI HOLDINGS, INC., an Illinois Corporation (as Borrower)

By:/s/ Robert J. Washlow

Robert J. Washlow Chief Executive Officer

CRONATRON WELDING SYSTEMS, INC., a North Carolina Corporation (as Borrower)

By: /s/ Robert J. Washlow

Robert J. Washlow Its: Chief Executive Officer DRUMMOND AMERICAN Corporation, an Illinois Corporation (as Borrower)

By: /s/ Robert J. Washlow

Robert J. Washlow Chief Executive Officer

ASSEMBLY COMPONENT SYSTEMS , LIMITED, a United Kingdom Corporation (as

Borrower)

Its:

By: /s/ Robert J. Washlow Robert J. Washlow

ts: Chairman of the Board of Directors

ASSEMBLY COMPONENT SYSTEMS, INC., an Illinois Corporation (as Borrower)

By: /s/ Robert J. Washlow

Robert J. Washlow

Its: Chairman of the Board of Directors

AUTOMATIC SCREW MACHINE PRODUCTS COMPANY, INC., an Alabama

Corporation (as Borrower)

By: /s/ Robert J. Washlow

Robert J. Washlow

Its: Chairman of the Board of Directors

C.B. LYNN COMPANY, an Illinois Corporation (as Borrower)

By: /s/ Robert J. Washlow

Robert J. Washlow

Its: Chairman of the Board of Directors

LP INDUSTRIAL PRODUCTS COMPANY, an Illinois Corporation (as Borrower)

By: /s/ Robert J. Washlow

Robert J. Washlow

Its: Chairman of the Board of Directors

LAWSON PRODUCTS DE MEXICO, S.A. DE C.V. (as Borrower)

By: /s/ Robert J. Washlow

Robert J. Washlow

Its: Chairman of the Board of Directors

July 18, 2005

Board of Directors Lawson Products, Inc.

We are aware of the incorporation by reference in the Registration Statement (Form S-8 No. 33-17912 dated November 4, 1987) of Lawson Products, Inc. and subsidiaries of our report dated July 18, 2005 relating to the unaudited condensed consolidated interim financial statements of Lawson Products, Inc. which are included in its Form 10-Q for the quarter ended June 30, 2005.

Pursuant to Rule 436(c) of the Securities Act of 1933 our report is not part of the registration statement prepared or certified by accountants within the meaning of Section 7 or 11 of the Securities Act of 1933.

/s/ ERNST & YOUNG LLP

Chicago, Illinois

#### **CERTIFICATIONS**

#### I, Robert J. Washlow, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Lawson Products, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
- (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
- (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
- (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
- (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
- (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 9, 2005

<u>/s/ Robert J. Washlow</u> Robert J. Washlow Chief Executive Officer

## CERTIFICATIONS

- I, Thomas Neri, certify that:
- 1. I have reviewed this Quarterly Report on Form 10-Q of Lawson Products, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
- (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
- (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
- (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
- (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
- (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 9, 2005

/s/ Thomas Neri

Thomas Neri Executive Vice President, Finance, Planning and Corporate Development; Chief Financial Officer; and Treasurer

## CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350 AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Lawson Products, Inc. (the "Company") on Form 10-Q for the period ending June 30, 2005 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), the undersigned Chief Executive Officer and Chief Financial Officer of the Company hereby certify, pursuant to 18 U.S.C. §1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002 that based on their knowledge: 1) the Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, and 2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company as of and for the periods covered in the Report.

# /s/ Robert J. Washlow

Robert J. Washlow, Chief Executive Officer

## /s/ Thomas Neri

Thomas Neri Executive Vice President, Finance, Planning and Corporate Development; Chief Financial Officer; and Treasurer

August 9, 2005